Promoting Pre-Service Teachers' TPACK Development in Social Virtual Reality

Practice- and Theory-Oriented Development and Evaluation of a Pedagogical Concept for Initial Teacher Education

> Inaugural-Dissertation zur Erlangung der Doktorwürde der Fakultät für Humanwissenschaften der

Julius-Maximilians-Universität Würzburg

Vorgelegt von Gabriela Ripka aus Würzburg

•

Würzburg

2022

Diese kumulative Inaugural-Dissertation zur Erlangung des Doktorgrades im Fach Erziehungswissenschaft wurde von der Fakultät für Humanwissenschaften der Julius-Maximilians-Universität Würzburg angenommen. Erstgutachterin: Prof. 'in Dr. Silke Grafe Zweitgutachterin: Prof. 'in Dr. Regina Egetenmeyer-Neher Tag der Disputation: 20. Oktober 2022

Acknowledgements

During the time of writing my dissertation and throughout the entire promotion journey, I was accompanied by people who supported me with their expertise, their passion for research and teaching, their warmth, and their way of being. I would like to express my special gratitude to some of them here.

First, I would like to thank my dissertation committee, especially Prof.in Dr. Regina Egetenmeyer-Neher for the competent supervision and reviewing. My deepest gratitude goes to my supervisor and doctoral mother, Prof.in Dr. Silke Grafe, for her invaluable guidance, support, and encouragement during the research and writing of this dissertation. I am so grateful for the trust she placed in me and for the many opportunities, she provided for me to grow and develop as a researcher. Her expert knowledge and dedication, as well as the freedom she gave me, were of crucial importance in helping me through the challenges of this process.

A huge thank you goes to my current and former colleagues at the Chair of School Pedagogy and the Chair of Human-Computer Interaction at the University of Würzburg. I would like to express my gratitude for their cooperation, patience, and understanding during this challenging period. I would like to give a special thank you to Dr. Jennifer Tiede who has supported me since the beginning of my academic journey and has been a constant source of encouragement, friendship, and expertise. She serves as an inspiration, demonstrating what women and mothers can achieve in this field. I also extend my gratitude to Dr. Sebastian Oberdörfer, whose insightful advice, criticism, and encouragement came at the right times.

Furthermore, I would like to express my appreciation to the student assistants who provided support, particularly in data collection and analysis for ENA. Thank you to Benedikt Krogoll, Paul Cebulla, Laura Koch, Veronika Banach, Kevin Werner, and Samuel Ruckdeschel for their excellent work.

Additionally, I must thank my online co-working partners Marta Maroni and Cristiana Levinthal. They stuck with me through the good and bad hour-long writing phases, which were sometimes more and sometimes less disciplined. Our conversations and mutual understanding of the ups and downs carried me through the entire process, and they were the reason I showed up every day. Grazie mille, Marta! Muito obrigada, Cristiana!

Tief verbunden und dankbar bin ich meinen Freund:innen, die mit Ausdauer, viel Kuchen, Hopfentee und guter Laune stets an meiner Seite standen und mich immer wieder angefeuert und aufgemuntert haben. Viele Umarmungen, Spaziergänge, Sportroutinen sowie konstruktive und aufbauende Gespräche halfen mir durch diese Zeit. Vielen Dank an Daniel Ukidve, Constanze Theinl, Tim Ulrich, Alexander und Sofia Rausch, Christian Seufert, Andrea Kleinhenz, Susanne Holzheimer, Niklas und Vicky Darkow.

También quiero agradecer a Elena y Heather Bentley, quienes son más como familia que amigas. No solo durante mi periodo de promoción me apoyaron con mucho cariño, comida, música latina y un lugar de retiro, sino que también me dieron mucha energía.

Un agradecimiento infinito es debido a mis padres, Albert y Noemi Greger, y a mi hermana, Theresia Greger. Les estoy especialmente agradecida por su paciencia infinita, consideración y apoyo incansable en todos los aspectos. Esto incluye muchos paquetes de comida (preparados con mucho cariño) y numerosos servicios de bricolaje. Han sido una roca sólida para mí, brindándome aliento y consuelo en los momentos más difíciles. También estoy agradecida por las reuniones familiares llenas de risas y alegría, que me han ofrecido un respiro bienvenido durante mi tiempo de obstáculos profesionales y personales. ¡Estoy feliz y orgullosa de tener una familia tan estupenda, los adoro!

Abstract (English)

The everyday use of digital media by children and young people offers new opportunities for participation, communication, and collaboration. However, to fully exploit this potential and prepare youths for the risks and challenges of media usage, the promotion of the digital competencies of students and teachers is an indispensable goal for educational institutions.

To meet this requirement, teacher education must be opened to innovative pedagogical concepts for initial teacher education that considers new technologies in a reflective, action-oriented way to promote competencies. Therefore, this work aims to promote the technological pedagogical content knowledge (TPACK) of prospective teachers that enables the purposeful integration of social virtual reality (Social VR) into the classroom. Consequently, a pedagogical concept is developed and evaluated in an iterative research and development process following the design-based research approach (DBR) through four consecutive studies.

The beginning of this paper introduces the theoretical approaches and concepts that form the foundation of the studies. First, TPACK is presented as a meta-conceptual awareness and target perspective for the successful integration of social VR into the classroom. To meet these requirements, the potential affordances of social VR for teaching and learning processes are outlined through a theoretical perspective. For the design and development of a pedagogical concept considering theoretical foundations and practical experiences, DBR is described as framing the work concerning research methodology.

Following DBR, Study 1 analyses and explores the existing expectations and requirements of lecturers and students regarding the goal-oriented use of social VR in the classroom. Hence, qualitative interviews were conducted with both target groups, lecturers, and students. The results show important contextual factors, such as an increased need for the supervision and support of both groups in using social VR hardware and software. Furthermore, lecturers and students want disruption-free communication and interaction processes in social VR. To address both aspects, support services, such as IT support, video tutorials, and abundant recovery breaks between teaching and learning processes are integrated as preliminary design features.

Study 2 analyzes how prospective teachers perceive teaching-learning activities in fully immersive social VR. Consequently, two teaching-learning scenarios based on the theorydriven pedagogical concept will be conducted. Qualitative interviews derived consequences for developing the pedagogical concept to promote TPACK among prospective teachers. For example, the concept design considers principles of action-oriented teaching, flipped classroom, and students' cognitive relief through several breaks.

To develop the pedagogical concept and to concretize the design features, the third study investigates how students perceive teaching and learning processes in social VR compared to video-based communication. For the promotion of TPACK as metaconceptual awareness in social VR, the implementation of peer group supervision enables metacognitive teaching and learning processes, such as reflection and problem solving. To approach a robust intervention in an authentic setting, the seminar takes place remotely. As a central learning activity, learners create instructional designs in an iterative design process with which social VR can be used in an action-oriented manner. Qualitative data collection was conducted through guided student reflection videos and semi-structured group interviews. The results again demonstrate the necessity of breaks and runs of peer group supervision. Generally, balancing the presentation of pedagogical and technical stimuli and avoiding cognitive overload proved to be a criterion that supported learning processes.

Study 4 evaluates pre-service teachers' development of TPACK as metaconceptual awareness in social VR and Zoom. As in previous studies, students plan and design lesson plans that are iteratively adjusted after three peer-group supervisions. For data collection, students used a graphic assessment of TPACK instrument (GATI) tool to estimate their current TPACK levels before and after the seminar. Self-assessments and portfolios were coded and qualitatively analyzed using epistemic network analysis (ENA). The results reveal quite homogeneous TPACK development in both groups. However, based on participants' ENA networks in social VR, more complex connections between knowledge domains become apparent. Consequently, it can be concluded that students who regularly work with social VR in an authentic context, based on their practical experience, can purposefully derive actions for planning and designing lessons with social VR.

As the results from Study 4 indicate, the pedagogical concept successfully promoted TPACK as metaconceptual awareness. In the concluding chapter, appropriate implications for teacher education research and practice are derived from findings. For example, further in-depth investigation of the nature of TPACK as metacognitive awareness could reveal learning-enhancing aspects of teacher education for media integration in class. Furthermore, seminar contents dealing with using new technologies should account for digital literacy aspects, leading to VR use considering moral values and practiced sustainability.

Abstract (Deutsch)

Die tägliche und mittlerweile selbstverständliche Nutzung digitaler Medien durch Kinder und Jugendliche bietet neue Möglichkeiten für Partizipation, Kommunikation und Kollaboration. Um dieses Potenzial jedoch voll ausschöpfen zu können und junge Menschen auf die Risiken und Herausforderungen der Mediennutzung vorzubereiten, ist es eine wichtige Aufgabe, die digitalisierungsbezogenen Kompetenzen von Schülern und Lehrern gezielt zu fördern.

Um dieser Aufgabe nachkommen zu können, besteht die Notwendigkeit der Öffnung der Lehrerbildung für innovative pädagogische Konzepte für die Hochschullehre, die neue Technologien reflektieren und handlungsorientiert für eine Kompetenzförderung berücksichtigen. Ziel dieser Arbeit ist es daher, das Professionswissen TPACK zukünftiger Lehrkräfte zu fördern, damit diese social Virtual Reality (social VR) handlungsorientiert in den Schulunterricht integrieren können. Dazu werden im Rahmen von vier konsekutiven Studien, einem Design-Based Research Ansatz folgend, ein Lehrkonzept für die Lehrerbildung entwickelt und in einem iterativen Forschungs- und Entwicklungsprozess evaluiert.

Zu Beginn der Arbeit werden die theoretischen Ansätze und Konzepte vorgestellt, die die Ausgangsbasis für die Studien bilden. Zunächst wird TPACK, verstanden als metakonzeptionelles Bewusstsein, als Zielperspektive für die gelingende Integration von social VR im Unterricht eingeführt. Weiterhin werden die Potenziale von social VR für Lehr- und Lernprozesse aus einer theoretischen Perspektive skizziert. Für die Gestaltung, Entwicklung und Evaluation eines pädagogischen Konzepts, das sowohl theoretische Grundlagen als auch praktische Erfahrungen berücksichtigt, wird der Design-Based Research Ansatz (DBR) als forschungsmethodische Rahmung der Arbeit beschrieben.

Dem Design-Based Research Ansatz folgend, widmet sich Studie 1 der Analyse und Exploration bestehender Erwartungen und Anforderungen von Dozierenden und Studierenden in Bezug auf den zielführenden Einsatz von social VR im Unterricht. Die Auswertung der qualitativen Interviews beider Zielgruppen zeigen wichtige Kontextfaktoren auf, wie zum Beispiel einen erhöhten Betreuungs- und Unterstützungsbedarf beider Gruppen bei der Nutzung von social VR Hard- und Software. Darüber hinaus wünschen sich Dozierende und Studierende reibungslose Kommunikations- und Interaktionsprozesse in social VR. Um beide Aspekte zu berücksichtigen, werden Unterstützungsangebote, wie zum Beispiel IT-Support, Video-Tutorials und viele Erholungspausen zwischen den Lehr- und Lernprozessen als vorläufige Gestaltungsmerkmale in das pädagogische Konzept integriert.

Studie 2 hat das Ziel zu analysieren, wie angehende Lehrkräfte die Lehr- und Lernaktivitäten in vollständig immersiver social VR wahrnehmen. Dazu werden basierend auf dem Lehrkonzept zwei unterschiedliche Lehrszenarien durchgeführt. Mit Hilfe qualitativer Interviews werden Erkenntnisse zur Weiterentwicklung des pädagogischen Konzepts zur Förderung von TPACK bei angehenden Lehrkräften gewonnen. So werden beispielsweise die Prinzipien des handlungsorientierten Unterrichts und Flipped Classrooms sowie die kognitive Entlastung der Studierenden durch mehrere Pausen bei der Konzeption berücksichtigt.

Zur Weiterentwicklung des pädagogischen Konzepts und zur Konkretisierung der Gestaltungsmerkmale wird in der dritten Studie untersucht, wie Studierende Lehr- und Lernprozesse in social VR im Vergleich zur videobasierten Kommunikation wahrnehmen. Mit TPACK als metakonzeptionelles Bewusstsein als Zielperspektive, ermöglicht die Implementierung von kollegialen Fallberatungen metakognitive Lehr- und Lernprozesse, wie zum Beispiel Reflexions- und Problemlösungsprozesse. Als zentrale Lernaktivität, erstellen Lernende handlungsorientierte Unterrichtspläne für den Einsatz von social VR im Schulunterricht in einem iterativen Designprozess. Die Auswertung der angeleiteten Reflexionsvideos und der halbstrukturierten Gruppeninterviews zeigen erneut, dass weitere Pausen und Durchläufe der kollegialen Fallberatung für eine kognitive Entlastung der Lernenden berücksichtigt werden müssen. Allgemein bewies sich die Einhaltung einer Balance zwischen der Präsentation von pädagogischen und technischen Stimuli und der Vermeidung von kognitiver Überforderung als ein Kriterium, das die Lernprozesse positiv zu unterstützen scheint.

Studie 4 evaluiert die Entwicklung von TPACK als metakonzeptuelles Bewusstsein von Lehramtsstudierenden in social VR und Zoom. Wie in den vorhergehenden Studien gestalten Studierenden in drei iterativen Durchläufen der kollegialen Fallberatung Unterrichtsentwürfe. Zur Datenerhebung nutzen die Studierenden das GATI-Diagramm, um ihren aktuellen TPACK-Entwicklungsstand vor und nach dem Seminar einzuschätzen. Die Kodierung und Auswertung der GATI Diagramme und Portfolios erfolgt mittels der epistemischen Netzwerkanalyse (ENA). Die Ergebnisse zeigen, dass die Entwicklung von TPACK in beiden Gruppen ziemlich konsistent ist. Jedoch weisen die ENA-Netzwerke der social VR Teilnehmer:innen komplexere Verbindungen zwischen den TPACK Wissensdomänen auf. Aus den Ergebnissen lässt sich schließen, dass Studierende, die regelmäßig mit social VR in einem authentischen Kontext arbeiten, basierend auf ihrer Praxiserfahrung, zielführende Handlungen für die Planung und Gestaltung von Unterricht mit social VR ableiten können.

Wie die Ergebnisse aus Studie 4 zeigen, konnte das pädagogische Konzept TPACK als metakonzeptuelles Bewusstsein in social VR erfolgreich fördern. Im abschließenden Kapitel werden aus den Erkenntnissen relevante Implikationen für Forschung und Praxis in der Lehrerbildung abgeleitet. Beispielsweise könnten weitere vertiefende Untersuchungen zum Wesen von TPACK als metakognitives Professionswissen, lernförderliche Aspekte für die Vorbereitung von Lehrkräften auf den Medieneinsatz aufzeigen. Weiterhin, in Anbetracht eines Bewusstseins für Ethik und Nachhaltigkeit beim Einsatz von neuen Technologien in der Lehrerbildung, sollten Curricula vermehrt auch medienerzieherische Aspekte in Betracht ziehen.

Papers included in the article-based dissertation

Paper 1: Ripka, G., Tiede, J., Grafe, S. & Latoschik, M. (2020). Teaching and Learning Processes in Immersive VR – Comparing Expectations of Preservice Teachers and Teacher Educators. In D. Schmidt-Crawford (Ed.), *Proceedings of Society for Information Technology & Teacher Education International Conference* (pp. 1863-1871). Online: Association for the Advancement of Computing in Education (AACE)

Abstract:

The usage of VR in higher education is not uncommon anymore. However, concepts are mainly still focusing on technical rather than pedagogical aspects of VR in the classroom. The exploration of the expectations of teacher educators as well as of preservice teachers appears indispensable (1) to achieve a sound understanding of requirements, (2) to identify potential design spaces, and finally (3) to create and to derive suitable pedagogical approaches for VR in initial teacher education. This paper presents results of guideline-based qualitative interviews comparing the expectations of teacher educators and of preservice teachers regarding teaching and learning in immersive virtual learning environments. The results showed that preservice teachers and teacher educators expect VR to enrich classes through interactive engagement in situations that would otherwise be too costly or dangerous. Regarding the design, teacher educators put the emphasis on functionality. Student teachers emphasized that they do not want to miss social interactions with their peers. Furthermore, both groups stated preferred modes of collaboration and interaction taking into account the characteristics of a virtual learning surrounding such as being able to use diverse learning spaces for group work. Interviewees agreed on two vital factors for effective learning and teaching processes: flexibility and the possibility of customization considering technical properties that are to deal with. Apart from this, preservice teachers emphasized strongly their worries about data usage and the ethics regarding using avatars and agents for representation.

Paper 2: Ripka, G., Grafe, S. & Latoschik, M.E. (2020). Preservice Teachers' encounter with Social VR – Exploring Virtual Teaching and Learning Processes in Initial Teacher Education. In E. Langran (Ed.), *Proceedings of SITE Interactive 2020 Online Conference* (pp. 549-562). Online: Association for the Advancement of Computing in Education (AACE)

Abstract

With 21st century challenges ahead, higher education teaching and learning need new pedagogical concepts. Technologies like social VR enable student-centered, action-oriented, and situated learning. This paper presents findings of the pedagogical implementation of a distributed social VR prototype, a fully immersive VR learning environment, into an Initial Teacher Education program in Germany.

The exploratory study addressed the following research questions: 1) How do preservice teachers perceive teaching and learning activities in fully immersive VR and 2) how should teaching and learning processes using social VR in Teacher Education be designed? It followed a design-based research approach. The pedagogical concept for teaching and learning insocial VR was based on principles of action-orientation. A convenience sample of three groups of five students each took part in a 90-minute teaching and learning scenario using a fully immersive VR learning environment. During these seminar units, students engaged in qualitative group interviews and shared their perception of the action-oriented teaching and learning activities in VR. The results showed that preservice teachers had the feeling of being less distracted in social VR. Additionally, during group activities, missing social and behavioral cues made communication procedures more challenging for participants. However, some participants noticed a stronger sense of community while collaborating with others.

Paper 3: Ripka, G., Grafe, S. & Latoschik, M.E. (2021). Peer group supervision in Zoom and social VR- Preparing preservice teachers for planning and designing digital media integrated classes. In T. Bastiaens (Ed.), *Proceedings of EdMedia + Innovate Learning* (pp. 602-616). United States: Association for the Advancement of Computing in Education (AACE)

Abstract

21-century challenges demand a change towards collaborative and constructive seminar designs in initial teacher education regarding preservice teachers acquiring meta-conceptual awareness (TPACK) about how to implement emerging technologies in their future profession. Against this background the paper addresses the following research questions: 1) How should a pedagogical concept for remote initial teacher education be designed to promote metacognitive learning processes of preservice teachers? 2) How do preservice teachers perceive these learning processes in video-based communication and social VR? Regarding the pedagogical concept, peer group supervision and an action- and development-oriented approach using Zoom and social VR were identified as relevant for an instructional design that provides collaborative and constructive learning processes for students. In this exploratory study, 17 students participated in two iterative cycles of peer group supervision performing design tasks in groups. A content analysis of reflective video statements and qualitative group interviews was carried out using a qualitative research design. Results indicate the successful implementation of peer group supervision. Regarding media's implementation, Zoom's screensharing option and breakout session benefitted the consultation process as well as social VR's "realistic" experience of creating a "sense of community".

Paper 4: Ripka, G., Grafe, S. & Latoschik, M.E. (2021). Mapping pre-service teachers' TPACK development using a social virtual reality and a video-conferencing system. In T. Bastiaens (Ed.), *Proceedings of Innovate Learning Summit 2021* (pp. 145-159). Online, United States: Association for the Advancement of Computing in Education (AACE)

Abstract

Social VR's characteristics, by offering authentic learning environments that enable interaction remotely and synchronously and permit learning experiences that affect learners in a multi-sensory way, offer great potential for teaching and learning processes. However, concerning its use to promote pre-service teachers' TPACK in initial teacher education, there remains a research desideratum. In this context, this exploratory study addressed the following research question: How did pre-service teachers' TPACK develop using a social VR learning environment prototype in comparison to a video-conferencing platform throughout a semester? Following a design-based research approach, an action-oriented pedagogical concept for teaching and learning in social VR was designed and implemented for initial teacher education at a German university with a convenience sample of 14 participants. The lesson plans were collected and analyzed with the help of

Epistemic Network Analysis (Shaffer, 2017) at three points of time during the semester and the GATI reflection process (Krauskopf et al., 2018). Further, 14 GATI diagrams gave insights into pre-service teachers' self-estimated TPACK. As the results indicate, pre-service students constructed more complex mental models of TPACK in social VR compared to the video-conferencing platform, indicating that more interrelations between knowledge domains could be constructed by planning and designing VR-integrated lesson plans.

Table of Contents

1. Introduction	13
2. Context	20
2.1 Technological Pedagogical Content Knowledge	20
2.2 Social Virtual Reality's Affordances for Teaching and Learning Processes	25
2.3 Design-Based Research	28
3. Teaching and Learning Processes in Immersive VR – Comparing Expectations of Pre-ser Teachers and Teacher Educators	vice
3.1 Aims of Study 1	32
3.2 Study 1	33
3.3 Conclusion	42
4. Preservice Teachers' Encounter with Social VR – Exploring Virtual Teaching and Learn Processes in Initial Teacher Education	ing 43
4.1 Aims of Study 2	43
4.2 Study 2	44
4.3 Conclusion	58
5. Peer Group Supervision in Zoom and Social VR – Preparing Pre-service Teachers for Planning and Designing Digital media -integrated Classes	60
5.1 Aims of Study 3	60
5.2 Study 3	61
5.3 Conclusion	76
6. Mapping Pre-service Teachers' TPACK Development Using a Social Virtual Reality and Video Conferencing System	77
6.1 Aims of Study 4	77
6.2 Study 4	79
6.3 Conclusion	94
7. Conclusion and Outlook	95
7.1 Summary of the Development Cycles and Their Research Results	96
7.2. Limitations of the Studies	. 100
7.3 Implications and Outlook	. 101
7.3.1 Modeling Technological Pedagogical Content Knowledge as Metaconceptual Awareness	102
7.3.2 Measuring Technological Pedagogical Content Knowledge as Metaconceptual Awareness	103
7.3.3 Teaching and Research with Social Virtual Reality: Implications for Ethics and Sustainability	103
7.3.4 Social Virtual Reality as an Authentic Learning Environment in Initial Teacher Education	104
7.3.5 Creating Collaborative Cultures in Social Virtual Reality and Beyond	105

.References

1. Introduction

In recent years, there have been gradual changes in how the educational sector is implementing digitization. These changes manifest themselves in the educational policy documents of different countries, for example the resolution of the Conference of the Ministers of Education and Cultural Affairs of the federal states of Germany that defines two main goals for digitization in schools (Ständige Konferenz der Kultusminister der Länder in der Bundesrepublik Deutschland, 2016). As media play an integral part in the socialization of pupils (see Feierabend et al., 2020), one aim is to foster their media competencies and thus sensitize them to the use of the media, with all its diverse aspects, within the framework of school subjects. Preparing students for active and responsible participation in cultural, social, political, economic, and professional life is of great importance because of the gap between students' increasing use of digital media and their own perceived digital competencies (Cress et al., 2018). The use of digital media gives young people new opportunities for participation, communication, and collaboration, but also exposes them to new forms of self-endangerment, political extremism, hatred, and violence on the internet (jugendschutz.net, 2021). Although 14- to 24-year-olds see the advantages of accessing the internet and using social media, they feel unprepared to face the risks involved. Forty percent of the participants of the Milieu Study on Trust and Security on the Internet in 2018 even admitted being scared of the threat posed by digital media and feeling unprepared to face this threat (Deutsches Institut für Vertrauen und Sicherheit im Internet¹ [DIVSI], 2018). Asked whom they would turn to for expert knowledge about mediarelated questions, the respondents indicated that they would rely first on own and friends' experiences instead of asking teachers or parents for advice (DIVSI, 2018).

Since teachers play an important role in fostering the digital and media competencies of children and adolescents, it is important that universities prepare pre-service teachers for the upcoming challenges of digitization. To support the meaningfulness of the didactical and pedagogical realization of lessons with a focus on media competences, teachers and pre-service teachers need to have and develop pedagogical approaches in which such competences are centralized (Tulodziecki, 2012; Tulodziecki et al., 2021).

Different national and international media pedagogical and digital competency models have evolved over the last decades. Depending on origin and background, they differ in terms of

¹German Institute for Trust and Security.

concepts, foci, and competency areas (see Tiede, 2020, for an overview). One prominent and widely cited framework in the discourse of media pedagogical competencies in teacher education is the technological pedagogical content knowledge (TPACK) framework (Mishra & Koehler, 2006), which can be adapted to different subjects. The popularity of this theoretical framework led to the publication of 2,202 TPACK-related works between 2005 and 2021 (Harris & Gallagher, 2021). Extending Lee Shulman's (1986) concept of pedagogical content knowledge (PCK), the TPACK model adds a further dimension to teachers' knowledge dimensions for effective teaching-namely, technology knowledge (TK). Regarding teachers' professionalization, TPACK relates to seven knowledge dimensions needed to plan and design media-integrated classes effectively (Mishra & Koehler, 2006). Along with pedagogical knowledge (PK) and content knowledge (CK) comes the necessity for teachers "to understand information technology broadly enough to apply it productively at work and in their everyday lives, to recognize when information technology can assist or impede the achievement of a goal, and to adapt continually to changes in information technology" (Koehler & Mishra, 2009, p. 64). However, given the rapid development of new technologies, continuous adaptation to change is challenging. Hence, understanding the processes underlying the use of technology in class goes beyond acquiring a dormant state of basic information technology (IT)-related knowledge. Rather, it includes providing teachers with the necessary strategies and supporting them to use new technology and to be self-determined in their planning and design of learning settings. Accordingly, Krauskopf et al. (2012) saw TPACK as a metaconceptual awareness comprising metacognitive learning processes involving higher order thinking.

Bertelsmann Stiftung's Digital Education Monitor indicated that pre-service teachers are not yet sufficiently prepared for the use of media in educational contexts (Schmid et al., 2017). Universities are better equipped with digital media than primary and secondary schools; however, technology implementation mainly occurs in teacher-centered teaching practices (Schmid et al., 2017). As a result, innovative learning formats for social and collaborative learning in higher education are lacking (Schmid et al., 2017). In addition, student teachers make less use of digital media than do their peers in other disciplines (Schmid et al., 2017). Although pre-service teachers use media in a private capacity to communicate with family and friends, they rarely use them for educational purposes (Schmid et al., 2017). In addition, Schmid et al.'s (2017) study showed that pre-service teachers are not media enthusiasts. Furthermore, fostering media pedagogical competencies is not yet an obligatory part of teacher education (Tiede & Grafe, 2016). Universities have to find new ways to engage pre-service teachers in

using digital media for educational purposes so they will be prepared to use media in their future classrooms.

In higher education, institutions increasingly use web conferencing tools, such as Zoom or Microsoft Teams, for remote learning processes (Massner, 2021). Video-conferencing platforms support the communication and collaboration processes of teams with the help of digital features, such as breakout rooms, collaboration boards, and digital polls. Although transmitted via a two-dimensional (2D) desktop screen, these online communication and collaboration processes can create immediacy and social presence (Hacker et al., 2019, 2020). The perception of social presence, which is seen as one affordance of video-conferencing tools, can positively influence the co-construction of knowledge and thus benefit social learning processes (Cleveland-Innes & Garrison, 2020). However, 2D video-conferencing tools also impose certain constraints on social interactions in teams. In face-to-face conversations, speakers interpret and anticipate each other's verbal and nonverbal cues. Speakers tend to pay attention to their own verbal cues but focus rather on the nonverbal cues of their conversation partners (Bucy, 2017). In an in-person conversation, speakers process nonverbal information, such as facial expressions, body language, gaze, or eye contact, in under 100 milliseconds (Bucy, 2017). The unconscious processing of information creates an impression of a person's character and contributes to the formation of social judgment and trust or distrust (e.g., Dorairaj et al., 2012; Hambley et al., 2007; Hacker et al., 2019). This whole process happens spontaneously in face-to-face communication but takes more effort when video-conferencing systems are used (Hacker et al., 2019). Although video-conferencing systems are considered to have high synchronicity, Bucy (2017) found that there are severe delays of 135-487 milliseconds in speakers taking their turn. As a result, participants must handle the cognitive load of producing and interpreting the nonverbal cues of others.

Studies have theorized that alongside the increase in cognitive load, several nonverbal mechanisms may lead to the so-called Zoom fatigue that users have experienced after spending long hours in online Zoom meetings (Bailenson, 2021; Fauville et al., 2021). For example, self-focused attention is quite common because of a concentration on one's speaker window and representation in Zoom, leading to such possible adverse effects as anxiety or depression (Bailenson, 2021). Moreover, unlike in in-person meetings, people using Zoom try to stay in the field of view and unconsciously feel obliged to move within a certain radius (Bailenson, 2021). Consequently, users feel physically trapped and experience disruptive effects on cognitive performance (Bailenson, 2021). Because of the heightened cognitive efforts

necessary to even out these disruptions in communication and collaboration, the capacity to devote cognitive processes to knowledge (co-)construction is limited, hindering effective group work and social learning processes (Bailenson, 2021).

A new way to overcome the problems described above in the use of video-conferencing systems while fostering pre-service teachers' TPACK may be the use of virtual reality (VR). VR offers experiences to users that come closer to real-life sensations than those offered by other technologies, enabling experiences that may even encroach on physical reality (Slater et al., 2020). Regarding education, this ability allows new ways of turning classrooms into explorative and interactive learning environments that enable the inclusion of content that would be too costly, abstract, dangerous, or in general, not feasible to explore with a class in real life. VR and immersive technologies, such as augmented reality (AR) and mixed reality (XR), are expected to become increasingly feasible as communication and interaction media in the future. Outside of entertainment, education will probably be the second sector after health care to be affected by the implementation of immersive technologies (Perkins Coie, 2020). As a special form of VR, social VR "is a web-based social interaction paradigm, mediated by immersive technologies and taking place in predesigned three-dimensional virtual worlds where individuals, represented by an avatar, may engage in real-time interpersonal conversation and shared activities" (Dzardanova et al., 2018, p. 1). Since VR is immersive in nature, usersrepresented by avatars—can experience the stimuli of their surroundings as if they were real and can interpret them as such; hence, their responses are similar to what they would be if interactions were initiated in the real world. For remote teaching and learning, in particular, the possibility of communicating and collaborating in real time and the feeling of being present in the virtual learning environment can be vital for social and situated learning processes.

From a theoretical perspective, the term VR has been interpreted in different ways (see Lou et al., 2021, for an overview). Furthermore, studies often lack a theoretical basis or target-oriented alignment with the pedagogical concept under consideration (Pellas et al., 2021). Thus, it is not easy to generalize about VR's affordances for teaching and learning processes. The rapid development of VR hardware and software, in particular, means that requirements for the integration of VR in educational contexts change dynamically.

In the empirical scientific discourse, there is disagreement on VR's potential for teaching and learning processes and its effects on learning outcomes. Whereas some studies have shown that VR has a rather distracting and cognitive-overloading nature in the context of learning settings (e.g., Parong & Mayer, 2021), others have confirmed VR's potential to motivate students to

engage and perform in class and to create authentic learning environments (e.g., Liou & Chang, 2018; Southgate, 2020). Pellas et al.'s (2021) review emphasized that the successful implementation of VR in lesson designs is contextual and depends on various elements. Given the design, quality, and nature of the hardware and software, the main characteristics of VR can be either beneficial or disruptive to communication and collaboration processes—and thus, they can have related effects on learning processes. The degree of immersion, for example, varies among different kinds of VR systems. The immersion level depends on how the VR system represents the inclusive, extensive, surrounding, and vivid illusion of a virtual environment to the user (Slater & Wilbur, 1997). Immersion is the technical basis for the subjective sensation of being present in a virtual environment. The feeling of social presence, as the "perception of non-mediation" (Lombard et al., 2000, p. 77), refers to both artificial social actors (agents) and real social actors represented digitally in the virtual world.

Immersion and presence can make VR a valuable medium in education. As with any medium, however, merely integrating VR into teaching and learning does not guarantee additional value or improved learning success. It is essential to plan the implementation of the medium carefully and to consider several factors, such as prerequisites, requirements, and the process of iterative development, to ensure a successful and useful VR-supported learning process (Huang et al., 2010).

The use of VR in teacher education is gaining in importance. To date, observations in Virtual Learning scenarios have concentrated on the technical aspects of VR implementation (Dalgarno & Lee, 2010). Few studies have dealt with pedagogical aspects or have attempted to set standards for teaching and learning in VR. However, an example in initial teacher education (ITE) is the interdisciplinary project "Breaking Bad Behavior" (Lugrin et al., 2016). Using an action-oriented pedagogical approach in a fully immersive virtual classroom helped pre-service teachers develop their class management competencies (Seufert & Grafe, 2020; Seufert et al., 2022). Seufert and Grafe's (2020) study showed that the sound interplay of pedagogical concepts with the VR learning environment is important for the successful implementation of social VR and competency development.

Against this background, the aims of this dissertation are as follows: (a) to develop a pedagogical concept that promotes pre-service teachers' TPACK using a design-based research process and an iterative approach; (b) to investigate how social VR's attributes benefit or hinder learning processes; and (c) to map pre-service teachers' TPACK development to recommend

future concepts in ITE. Hence, four studies are conducted with a view to answering the following research questions:

- What do student teachers and teacher educators expect of a successful virtual reality application in Initial Teacher Education (ITE?
- How do pre-service teachers perceive teaching and learning activities in social VR?
- How should a pedagogical concept for remote initial teacher education be designed to promote the metacognitive learning processes of pre-service teachers?
- How do pre-service teachers perceive these learning processes in video-based communication and social VR? How did pre-service teachers' TPACK develop using a social VR learning environment prototype compared to a video-conferencing system?

The remainder of this work is structured as described below.

In Chapter 2, the context is presented, and theoretical approaches and concepts are provided. First, Section 2.1 introduces TPACK (Mishra & Koehler, 2006), a prominent theoretical framework suggesting that teachers' successful technology implementation in class goes back to the combination of the three main knowledge domains—namely, PK, TK, and CK (Mishra & Koehler, 2006). Based on the interpretation of the TPACK model as metaconceptual awareness (Krauskopf et al., 2018), this section identifies pre-service teachers' requirements for knowledge acquisition in ITE. To meet the requirements, Section 2.2 outlines the potential affordances offered by social VR for teaching and learning processes from a theoretical perspective. For the design and development of a pedagogical concept that considers theoretical foundations and in-practice experiences, Section 2.3 describes the design-based research (DBR) approach. In particular, it focuses on Euler's (2011, 2014) basic structure model, which allows the iterative analysis, exploration, design, and development of the pedagogical concept. The circular structure of Euler's model also serves as a frame for the studies carried out for this research (Chapters three to five), which build on each other by using insights gained to further create and develop the pedagogical concept, leading to its evaluation in Chapter 6.

Chapter 3 analyzes teacher educators' and pre-service teachers' expectations of teaching and learning processes in social VR learning environments to answer the first research question. Following the call to combine scientific and practical perspectives, it identifies relevant theories and addresses existing practical conditions. To understand the requirements needed to plan and design a pedagogical concept and a social VR prototype for pre-service teachers and teacher educators, a qualitative interview method is used to provide the first design requirements and hypotheses for the design and development process.

Considering the in-practice experience of teacher educators and pre-service teachers, Chapter 4 investigates how pre-service teachers perceive the design features of social VR in changing learning scenarios. A guided interview method is used to prompt students to talk about their experiences in VR. The new insights gained lead to the revision and adaptation of the design hypotheses, which serve as a template for the further development of the prototypical pedagogical concept.

Following the design demand to promote TPACK and considering the affordances of social VR for learning, Chapter 5 introduces the approach of peer group supervision (Tietze, 2010). Apparently, the core elements of this approach allow learners to co-construct knowledge via reflection and peer feedback (Tietze, 2010). In line with the assumption that as a type of metaconceptual awareness, TPACK can be achieved via higher order thinking, such as reflecting and problem solving (Krauskopf et al., 2015; Krauskopf et al., 2018), peer support may afford metacognitive strategies to implement VR successfully. For the first test of the adopted design features, the study is undertaken in an authentic teaching–learning environment. A qualitative interview method is carried out to give insights into further improvements of the design measures of the pedagogical concept to allow metacognitive processes and to use the benefits of the social VR learning environment to promote learning processes.

Chapter 6 focuses on the implementation and evaluation of the pedagogical concept in an authentic practice context in ITE. Based on the results, assumptions can be made regarding design principles for the pedagogical concept to promote TPACK as a metaconceptual awareness in social VR.

Chapter 7 summarizes the findings of all four studies, presents the work's limitations, and outlines implications for further research and practice.

2. Context

2.1 Technological Pedagogical Content Knowledge

To be able to promote students' media competencies, during ITE, pre-service teachers must acquire the knowledge and skills required to effectively plan and design classes that integrate the use of media (see Chapter 1). Education research has shown that teaching and learning processes only benefit from technological affordances if certain conditions are met (Conole & Dyke, 2004). Authors have agreed on the complexity of properly implementing educational technology in pedagogical concepts (Foulger et al., 2017; Tondeur et al., 2019). Although there is little agreement on what is required for teachers to ensure successful media integration, teachers play a crucial role in this implementation process. Four main factors influence and predict the effective integration of information and communication technology (ICT) into teacher practice; these are as follows: (a) teacher knowledge, (b) teacher self-efficacy, (c) teacher beliefs, and (d) school/subject culture (Ertmer & Ottenbreit-Leftwich, 2010; Sang et al., 2011).

Various theoretical models, frameworks, and approaches have discussed which competencies are necessary to develop media integration in teacher education (see Tiede, (2020) for an overview). Reviews of competency frameworks have shown that various aspects differ in the theoretical approaches, such as the emphasis of media didactics in the models, the target group for the promotion of competencies, the underlying concept of "competencies" itself, and the overall composition of the competence models (see McGarr & McDonagh, 2019; Schmid & Petko, 2020; Tiede, 2020). As there is no common denominator, it is difficult to compare international frameworks without facing contextual bias. In her comparative analysis of international models, Tiede (2020) chose the generic term "media-related educational competencies," which allows international comparison.

Three of the international models most prominently discussed in the literature are DigCompEdu (Redecker, 2017), UNESCO-ICT (UNESCO, 2018), and TPACK (Mishra & Koehler, 2006). Since the main objective of the current research is to promote the media-related educational competencies of pre-service teachers in a social VR learning environment, TPACK was considered an appropriate model for the following reasons: First, not only is it one of the most cited frameworks in the field of teacher professionalization, but it also has an empirical foundation. Second, it is particularly used in the context of ITE. Hence, the current research could build on the existing research of measurements and evaluations of pre-service teachers' TPACK.

TPACK is a theoretical framework built on the concept of PCK (Shulman, 1986, 1987). After

a nationwide call for new reforms in teacher education in 1986, Shulman (1986) discussed the efforts made and pointed out possible weak points. According to the reform's supporters, an ever-growing knowledge base derived from research justified the implementation and extension of standards that enabled evaluation and examination processes in teacher education (Shulman, 1986). Shulman criticized the superficial consideration of the term "knowledge base" and elaborated on the knowledge needed for a "novice teacher" to become an "expert" one (1986, p. 6). Expert teachers not only transmit CK to students but can also articulate why a particular statement is considered justified, why pupils might need this knowledge, and how it relates to other contexts (Shulman, 1986). It is not enough to acquire CK to develop the ability to teach in this way. Shulman (1987) recognized PCK as one vital source of knowledge that "identifies the distinctive bodies of knowledge for teaching" and "represents the blending of content and pedagogy into an understanding of how particular topics, problems, or issues are organized, represented, and adapted to the diverse interests and abilities of learners, and presented for instruction" (p. 9). With the ever-growing need to prepare pre- and in-service teachers for technology integration in educational contexts, several works have built on the notion of PCK and developed approaches to integrating technological knowledge (TK) into Shulman's PCK concept (see Mishra & Koehler, 2006; Niess, 2005; Pierson, 2001). For example, Pierson (2001) suggested adding TK to PK, CK, and PCK. Beyond the first two intersections, which Pierson labeled A (knowledge of content-related technology resources) and B (knowledge of contentrelated technology resources), the third intersection represents C, TPCK (combining TK, PK, and CK). According to Pierson (2001), TPCK comprises knowledge of "true technology integration" (p. 427). Inherent to all approaches adding TK to the PCK model is the assumption that teaching effectively with technology results from the dynamic interplay of PK, TK, and CK rather than from treating knowledge bases as isolated knowledge bodies.

One of the most prominent approaches is the theoretical framework of TPACK developed by Koehler and Koehler (2005, 2006, 2007, 2008, 2009); these researchers coined the term TPCK, which later evolved into TPACK. Noting that modern technologies develop rapidly and lack adjustability to educational contexts, the authors emphasized the necessity for pre-service teachers to prepare for effective technology integration in the classroom. For this, they need to acquire TPACK.

Figure 1

TPACK Venn diagram (Koehler & Mishra, 2009)



As displayed in the Venn diagram in Figure 1, there are three knowledge domains at the core of the model—namely, PK, CK, and TK (Koehler & Mishra, 2009). Four more knowledge areas—technological pedagogical knowledge (TPK), technological content knowledge (TCK), PCK, and TPACK, represent the intersections of the three primary knowledge domains (Koehler & Mishra, 2009).

The individual knowledge domains can be described in more detail as follows:

- **CK** is knowledge about the subject matter (Koehler & Mishra, 2009). Derived from Shulman's (1986) notion of CK, it comprises "the knowledge of theories, concepts, ideas, organizational frameworks, knowledge of evidence and proof, and practices and approaches to develop such knowledge" (Koehler & Mishra, 2009, p. 63).
- **PK** is knowledge about teaching and learning processes (Koehler & Mishra, 2009). In addition, it involves understanding students' learning processes, planning and designing classes, and classroom management skills (Koehler & Mishra, 2009).
- In line with Shulman (1986), PCK refers to knowledge of how to teach a particular subject (Koehler & Mishra, 2009). Going beyond the mere transfer of knowledge, PCK includes processes connected to the representation of content, such as designing the teaching and learning process in a way that is conducive to learning (Koehler & Mishra, 2009). According to the authors PCK "covers the core business of teaching, learning, curriculum, assessment and reporting, such as the conditions that promote learning and the links among curriculum, assessment, and pedagogy" (p. 64).

- TK goes beyond technical usage and includes the teacher's knowledge of how the technology supports or drives processes in a target-oriented manner (Koehler & Mishra, 2009). This requires profound knowledge about a technology's characteristics, affordances, and application scenarios (Koehler & Mishra, 2009). Moreover, given the rapid development and innovation of technology, this knowledge needs to adapt to current conditions flexibly and should thus involve various strategies (Koehler & Mishra, 2009).
- TCK refers to the fruitful combination of TK and CK (Koehler & Mishra, 2008, 2009). It represents teachers' knowledge of the various ways in which a technology can be used to deliver content (Koehler & Mishra, 2009). In particular, it involves knowledge of how the subject matter influences how a technology is applied (Koehler & Mishra, 2009).
- **TPK** involves the goal-oriented adaptation of technology to teaching and learning processes (Koehler & Mishra, 2009). In addition to a deep knowledge of the functionalities of technology's features and how to use them in pedagogical contexts, TPK "requires a forward-looking, creative, and open-minded seeking of technology use" (Koehler & Mishra, 2009, p. 66).
- **TPACK** represents the interplay of all the knowledge components above (Koehler & Mishra, 2009). The authors distinguish TPACK as a "unique body of knowledge" that requires "teachers' cognitive flexibility not just in each of the key domains, but also in the manner in which these domains and contextual parameters interrelate" (Koehler & Mishra, 2009, p. 66).

Mishra and Koehler (2006) also emphasize the importance of binding contextual factors, such as school infrastructure and learners' background. In 2019, Mishra labeled the outer circle line in the model as "Contextual Knowledge" (see Figure 1); this includes the teacher's awareness of the external factors influencing the teaching and learning processes, such as the availability of existing resources and regulations.

TPACK has gained wide recognition and prominence both nationally and internationally. Indeed, more than 1,418 journal articles, 318 chapters, 28 books, and 438 dissertations referring to TPACK have been published to date (Harris & Gallagher, 2021). Literature reviews (e.g., Irwanto, 2021; Wang et al., 2018; Willermark, 2017) have confirmed that most studies referring to TPACK are from North America; nevertheless, publications are increasingly emerging from other countries (Irwanto, 2021). The main topics on which works have focused are as follows: (a) whether TPACK is by nature integrative or transformative (Angeli & Valanides, 2013; Chai et al., 2010), (b) whether TPACK is domain-generic or domain-specific knowledge (Guzey & Roehrig, 2009; Lin et al., 2013), (c) the composition and interplay of the knowledge domains (Archambault & Barnett, 2010), (d) methods leading to TPACK (Harvey & Caro, 2017; Tseng et al., 2019), (e) possible extensions of TPACK (Porras-Hernández & Salinas-Amescua, 2013; Urban et al., 2018), and (f) how to measure and develop TPACK (Graham et al., 2012; Harris & Hofer, 2009). In this work, TPACK has the following attributes: (a) it serves as a target for pre-service teachers' learning processes, and accordingly, (b) it guides the iterative design and evaluation of the pedagogical concept.

The requirements for TPACK are diverse and complex (see Chapters 5 and 6 for more detail). Koehler and Mishra (2009) have emphasized the "dynamic" (p. 61) nature of the knowledge domains and their respective context. On the one hand, this dynamic nature comes from the different existing external and internal conditions pre-service teachers have to face at schools, and on the other, it stems from the continuous rapid development of technologies and their associated diverse sets of affordances for the teaching and learning processes (see Section 2.2). Thus, to go beyond a fixed body of knowledge that can be acquired, pre-service teachers need metacognitive strategies that involve higher order thinking skills (Krauskopf et al., 2015; Krauskopf et al., 2018). Such skills enable them to analyze, evaluate, and reflect on how technologies can be used in a target-oriented way to promote competencies in different contexts (Krauskopf et al., 2015; Krauskopf et al., 2018). The more pre-service teachers experience these strategies during their teacher education, the higher their perceived competence to use ICT for learning processes and to strengthen their instructional practice (Tondeur, 2018). In this work, TPACK acquisition is considered a dynamic learning approach that involves metacognitive learning processes (Krauskopf et al., 2015; Krauskopf et al., 2018). Krauskopf et al. (2018) built on the transformative view of TPACK and defined it as a

coherent scientific framework theory [that] is (1) a unitary shape with a clear application context (teaching with technology), (2) the assumption of a limited number of pre-suppositions about technology, pedagogy and content (ontological and epistemological) that constrain the construction of more specific theories (mental models) derived from them, (3) the idea of a metaconceptual frame for the systematic relations of these presuppositions and the teacher's knowledge of the sub-domains. (p. 22) These authors assumed that TPACK is achievable through a two-level process of constructing mental models (Krauskopf et al., 2018). The knowledge subdomains (TPK, PCK, TCK) translate to mental models on the first level. According to the authors, mental models are constructed through reflection on theory and considering others' teaching practice (Krauskopf et al., 2018). Thus, the second-level transformation results in the "higher mental model" of TPACK, which involves the meta-awareness of effectively implementing technology in teaching and learning processes (Krauskopf et al., 2015, pp. 4–5). This complex transformation process poses a challenge for the pedagogical concept of designing respective learning activities, allowing the construction of TPACK as metaconceptual awareness.

2.2 Social Virtual Reality's Affordances for Teaching and Learning Processes

Since social VR offers potential for fostering teaching and learning processes in ITE (see Chapter 1), this section focuses on the terminology and concepts concerning the VR learning environment and its affordances for teaching and learning scenarios.

With regard to the term "social VR," various interpretations of VR exist in the literature (e.g., Hamilton et al., 2020; Radianti et al., 2020; Riegler et al., 2021). In the current research, as initially stated, the definition of social VR proposed by Dzardanova et al. (2018) is used as a basis because it emphasizes the affordance of VR to allow social interactions between participants: "Social Virtual Reality is a web-based social interaction paradigm, mediated by immersive technologies and taking place in predesigned three-dimensional virtual worlds where individuals, represented by an avatar, may engage in real-time interpersonal conversation and shared activities." (p. 1)

Immersive technologies can refer either to semi-immersive systems, representing a combination of 2D and immersive extensions, or to highly immersive systems (Hamilton et al., 2020). In this research, the social VR learning environment ViLeArn (Latoschik et al., 2019) is used, which is a highly immersive system (for a description see Chapters 4-6).

Social VR is a research topic in many scientific disciplines, including human-computer interaction, medicine, education, and healthcare; the affordances for teaching and learning scenarios in social VR vary among these disciplines. Many interpretations of the concept of affordances can be traced back to two main strands. From the ecological perspective, the perceptual psychologist Gibson (1979) described affordances as follows: "The affordances of the environment are what it *offers* the animal, what it *provides* or *furnishes*, either for good or ill. ... It implies the complementarity of the animal and the environment" (p. 127, Gibson's italics). Investigating what the affordances of nature provide for the environment and its

creatures, Gibson (1979) assumed that particular objects stimulate their surroundings to enable them to take possible actions. Hence, he used the term *affordance* to refer to the relationship between an object and an agent. Gibson's (1979) concept of affordance implies that living beings perceive the objects in their environment against the background of their bodies and the resulting corresponding possibilities of action that an object offers. Thus, the affordance of an object is independent of the needs, attention, or evaluation of an observer (see Guski, 1996). For example, a chair "affords" sitting, regardless of whether someone wants to sit on it (Guski, 1996, p. 5).

Norman (1988) introduced another popular definition of *affordance* connected to the design of interfaces and their usability, mainly used in the field of Human Computer Interaction (HCI):

[...] the term *affordance* refers to the perceived and actual properties of the thing, primarily those fundamental properties that determine just how the thing could possibly be used. [...] Affordances provide strong clues to the operations of things. Plates are for pushing. Knobs are for turning. Slots are for inserting things into. Balls are for throwing or bouncing. When affordances are taken advantage of, the user knows what to do just by looking: no picture, label, or instruction needed. (Norman 1988, p.9)

In Norman's (1988) definition, the concept of affordance also comprises the user's perception, and it goes beyond the relationship between object and agent. In this sense, an affordance is a visible design aspect of an object (McGrenere & Ho, 2000). In contrast to Gibson (1979), who defined an affordance as the offering character of an object's utility, Norman (1988) focused on the user's perception of the visible design characteristics of an object as its usability.

In the discourse of affordances for social VR in education, several authors refer to Bower's (Bower, 2008; Bower & Sturman, 2015) term "learning affordances" (e.g., Dalgarno & Lee, 2010; Southgate, 2020). Bower built on both, Gibson's (1979) and Norman's (1988) use of affordances. Rather than evaluating technologies' effects on the user, he cited Gibson (1979) when investigating the potential of possible technologies for the design of learning tasks. Thus, Bower (2008) also emphasized the detachment of the object's affordance from any context such that it "avoids any contextual biases that could be caused by the experience or culture of the user" (p. 5). In contrast, when evaluating users' perceptions of affordances, Bower and Sturman (2015) followed the perceived usability aspect of technologies' affordances and based their

work on Norman's (1988) definition. They concluded that "if the use is there but not perceived, then it is of no educational benefit" (Bower & Sturman, 2015, p. 345). Bower's (2008; Bower & Sturman, 2015) approach to the use of the term *affordance*—that is, that it depends on the respective research focus—is employed in the current research throughout the DBR cycle. In terms of analyzing and exploring social VR's potential for teaching and learning processes in ITE, *affordances* serves as a generic term in this work (see Chapters 3 and 4) in the sense put forward by Gibson (1979) and Bower (2008). These affordances, derived objectively for the design of teaching and learning activities, are considered separately from users' perceptions and cultural context (Bower, 2008; Gibson, 1979). However, when the focus is on the evaluation of pre-service teachers' perception of social VR's use in teaching and learning activities, the term *affordances* is used in the sense put forward by Norman (1988) (Bower & Sturman, 2015). Here, the implementation and design testing of the pedagogical concept is inseparable from the perceived affordances of the social VR learning environment (see Chapters 6 and 7).

Against the background delineated above, three of social VR's affordances are central to the design of teaching and learning activities in this work; these are immersion, presence, and embodiment. The concept of immersion refers to the mere technical properties a social VR system offers that can be objectively assessed (Kilteni et al., 2012; Slater & Sanchez-Vives, 2016; Slater & Wilbur, 1997). The degree of experienced immersion depends on the technical characteristics transmitting the "illusion of reality" in a way as vivid, extensive, encompassing, and inclusive as possible (Slater & Wilbur, 1997, p. 3). *Presence* differs from *immersion* in that it is a psychological construct representing the "human reaction to immersion," leading to the "sense of being there" in the virtual environment (Slater & Wilbur, 1997, pp. 4–5). Based on the subjective perceptions of place and plausible illusions, users interpret their surroundings as realistic stimuli, leading to actions being carried out as they would be in the "real world" (Slater & Wilbur, 1997, p. 5). Embodiment results from the illusion of ownership of the virtual body and leads to the feeling of being able to carry out certain actions with gestures or movements, which are closely related to the construct of agency (Johnson-Glenberg, 2018).

In comparison with other educational technologies, affordances are unique features that benefit multimodal, adaptive, authentic, and interactive learning scenarios. Dalgarno and Lee (2010, pp. 18–21) saw the affordances of three-dimensional (3D) virtual environments in the facilitation of tasks that have the following characteristics:

- They lead to the development of enhanced spatial knowledge representation of the explored domain.
- They would be impractical or impossible to undertake in the real world.
- They foster increased intrinsic motivation and engagement.
- They lead to improved transfer of knowledge and skills to real situations through contextualization of learning.
- They allow richer and/or more effective collaborative learning than is possible with 2D alternatives (Dalgarno & Lee, 2010, pp. 18–21).

Similarly, Hellriegel and Čubela (2018) took Arnold's (2012) principles for successful learning as the basis of the potential of VR for the following elements: (a) learning as a self-directed, activating, and constructive process; (b) learning as a motivating process; (c) learning as a situated and hands-on process; and (d) learning as a social process. With the objective of developing a pedagogical concept to promote the TPACK of pre-service teachers, the current research implements social VR's affordances in teaching and learning processes in ITE (see Chapters 3 to 6).

2.3 Design-Based Research

The promotion of TPACK in a social VR learning environment represents a desideratum in research. Therefore, to generate insights for practice and further develop theory building to promote TPACK in this context, the DBR approach is a suitable framework.

Although the term *DBR* was coined by the Design-Based Research Collective (2003), it has received different accentuations in the literature from various perspectives. Over the past 20 years, DBR has become an established research approach in Anglo-American educational research, blending empirical educational research with the theory-driven design of learning environments. Plomp (2010) defined DBR as follows:

The systematic study of designing, developing and evaluating educational interventions (such as programs, teaching-learning strategies and materials, products and systems) as solutions for complex problems in educational practice, which also aims at advancing our knowledge about the characteristics of these interventions and the processes of designing and developing it. (p. 13)

In response to the existing gap between educational research on innovation and its relevance for actual in-practice teaching and learning scenarios, several concepts have emerged with the objective of focusing on the development and design of educational processes. In comparison with experimental or evaluation research, the new approaches do not treat design and research as two consecutive steps in the research process; rather, the design becomes the source for the development of theories (Euler, 2014). Under the umbrella term of "design research," Euler (2014) pointed to existing variations of the approach, such as "design-experiments" (Brown, 1992), "development research" (Van den Akker, 1999), "formative research" (Newman, 1990), and "educational design research" (McKenney & Reeves, 2012). Instead of speaking of a methodology, some authors (see Euler, 2011; McKenney & Reeves, 2013; Raatz, 2016) have agreed that design research is more like a frame for existing research methods (quantitative, qualitative, or mixed methods) than an inherently "new" method.

In line with the developments outlined above, German-speaking authors adopting the DBR perspective in media education have also emphasized the need to develop and design educational processes, including innovative media. Tulodziecki et al. (2018), following the action-oriented approach in media education, referred to the DBR method as an "accentuation" (p. 436) of design and development-oriented research. The authors differentiated action research (e.g., Altrichter & Posch, 2007), integrative research strategies (e.g., Brown, 1992), didactical development research (e.g., Einsiedler, 2010), and the practice- and theory-oriented development and evaluation of concepts for educational action (e.g., Tulodziecki, 1983), as well as the concept of development-oriented educational research (e.g., Reinmann & Sesink, 2011), as further variations of this line of research. All these approaches, including DBR, vary in some respects; however, at their core, they share the intention of improving the scientific foundation of action in practice and in educational processes (Tulodziecki et al., 2018).

The main characteristics of DBR are using an iterative and circular approach to develop solutions to problems, the theory-driven anchoring of the research and design process, the integrative use of research methods, and the continuous interplay of science and practice (see Euler, 2011; McKenney & Reeves, 2012; Raatz, 2016; Reinmann & Sesink, 2011; Tulodziecki et al., 2013). To derive implications for in-practice implementation, Euler (2014) reviewed the most prominent DBR process models (Anderson & Shattuck, 2012; Bannan-Ritland, 2003; McKenney & Reeves, 2013; Reeves, 2006), and based on their shared features, he developed a basic structure model. Then, "to develop a paradigm-constitutive normative structure for the scientific community of design researchers by providing the foundation and the justification of

the guidelines" (Euler, 2014, p. 22), he outlined each step of the model, explaining the *core requirements*, the *core questions and guidelines*, and what *result* is expected at the end of each process.

Along with the *iterative and circular structure* of Euler's (2014) process model, the main features of the DBR approach mentioned above require further consideration.

Figure 2

Research and development cycles in the design research context



The starting point of the research and development cycles is the specification of the problem at hand. With the goal "to identify the (key) objectives related to research and design issues" (Euler, 2014, p. 25), a precise analysis of the factors connected to the problem is necessary to lay the ground for subsequent procedures. However, it is insufficient to consider only the contextual factors connected to the problem observed in practice when it comes to specifying the problem. Moreover, for a holistic analysis, scientific theories must underpin practical knowledge. Therefore, the analysis follows a theory-driven derivation of possible design hypotheses, constructing a "theoretical frame of reference" (Euler, 2014, p. 26) that comprises problem statements, design requirements, and possible solutions. Multiple (iterative) cycles of development, testing, evaluation, and refinement of the design follow. Systematic documentation of the development process is necessary to comprehend the acquisition process of knowledge.

In the context outlined above, the *theory-driven anchoring of the research and design process* plays a twofold role throughout the procedure. Although on the one hand, theory serves as a foundation for the derivation of the conceptual framework, on the other, it allows conclusions

to be drawn for the further development, redefining, and evaluation cycle of the underlying design principles. Thus, theory application, verification, and development are interconnected (Euler, 2014).

Concerning *the integrative use of research methods*, following McKenney and Reeves (2012), Euler (2014) differentiated between the foci of various evaluations (alpha, beta, and gamma testing) throughout the development stages, hand in hand with corresponding evaluation strategies and methods. As possible evaluation strategies, he outlined the four following procedures: (a) discussion and screening of the intervention with focus groups, checklists, and document analysis; (b) approaching external experts; (c) conducting a pilot test with a small sample; and (d) a tryout in the actual situation. He particularly emphasized the importance of the acquisition and analysis of qualitative data to gain a holistic perspective on the contextual factors involved (Euler, 2014).

Throughout all phases of the process model, the *interplay of science and practice* is essential for the implementation of design in its application context (Euler, 2014). Incorporating practitioners into the research and development cycles enables the evolution of practical problem-solving approaches that can be transferred to real application scenarios (Euler, 2014). Further, with the help of practitioners' knowledge, interventions can be designed in a more objective and effective way, avoiding unnecessary detours for knowledge generation (Euler, 2014). However, Euler (2014) also warned that scientists' "critically evaluative attitude" and practitioners' "design-oriented and decision-oriented position" (p. 27) may diverge throughout the research and development process.

Euler's process model provides a suitable framing for the research process of the current work. Since the objective is to promote pre-service teachers' TPACK in a fully immersive learning environment, numerous contextual factors must be considered throughout the research process. Because of the desideratum in research, it is not clear what these contextual factors comprise. The two focuses on TPACK as a metaconceptual awareness and social VR for teaching and learning processes represent complex reference frames that are only complementary under certain conditions. Further, in the context of ITE, there are rarely best practices or hands-on experiences. However, to ensure that the designed pedagogical concept and social VR learning environment are suitable for in-practice implementation and knowledge generation, the close interplay of science and practice is crucial. Since the technical development of the social VR learning between scientists and practitioners is all the more important for effective teaching–learning

processes. Euler's (2014) model emphasizes both perspectives at each stage of the research and development cycle. Dismantling the complexity of this multi-layered research interest, Euler's (2014) three-part orientation starts with the core requirements, moves to core questions, and concludes with the expected results of each stage. Thus, the model offers a template that entails guided practices, allowing the design of a pedagogical concept for social VR that finds its way into in-practice implementation and knowledge acquisition beyond the research process.

3. Teaching and Learning Processes in Immersive VR – Comparing Expectations of Pre-service Teachers and Teacher Educators 3.1 Aims of Study 1

As per the first and second stages of Euler's (2014) process model, Study 1 specified a given problem as the starting point for the research and development cycles and evaluated the literature and experience (Section 3.2). The problem statement of this work was derived from two lacks in ITE. First, pre-service teachers lack media-related educational competencies and need the promotion of TPACK (see Foulger et al., 2017, Voogt et al., 2013). Second, there is a lack of innovative pedagogical concepts integrating social VR to promote TPACK in ITE (see Foulger et al. 2017, Tondeur et al., 2019). Consequently, pedagogical concepts for ITE are needed that promote pre-service teachers' TPACK, which is addressed in the current research. To ensure the practicability of the pedagogical concept for in-seminar implementation, an intervention must consider the close interplay of science and in-practice experiences (see Euler, 2014; Raatz, 2016). When reviewing literature and hands-on experiences, most works have focused on the technical perspective of VR implementation (e.g., Pellas et al., 2021; Radianti et al., 2020). For the successful implementation of fully immersive VR in practice, however, pedagogical factors play a crucial role (Huang et al., 2010). Hence, the theory-based design principles of a pedagogical concept and the in-practice experience of practitioners have to be considered. Teacher educators' and pre-service teachers' attitudes and beliefs about a given technology are indicators of successful future interaction with the medium in their teaching practice (Blömeke, 2017; Ertmer & Ottenbreit-Leftwich, 2010; Sang et al., 2011). Accordingly, the first study in this work (Section 3.2.) addressed the first research question:

• What do student teachers and teacher educators expect of a successful virtual reality application in ITE?

The results allowed us to derive the first requirements for the design of the pedagogical concept and an intervention as well as to extend the problem statement (Euler, 2014).

3.2 Study 1

SITE 2020 - Online, , April 7-10, 2020

Teaching and Learning Processes in Immersive VR – Comparing Expectations of Preservice Teachers and Teacher Educators

Jennifer Tiede	Silke Grafe	Marc Latoschik
Chair of School Pedagogy	Chair of School Pedagogy	Chair of HCI
University of Wuerzburg,	University of Wuerzburg,	University of Wuerzburg,
Germany	Germany	Germany
Jennifer.Tiede@uni- wuerzburg.de	Silke.Grafe@uni- wuerzburg.de	Marc.Latoschik@uni- wuerzburg.de
	Jennifer Tiede Chair of School Pedagogy University of Wuerzburg, Germany Jennifer.Tiede@uni- wuerzburg.de	Jennifer Tiede Silke Grafe Chair of School Pedagogy University of Wuerzburg, Germany Germany Jennifer Tiede@uni- wuerzburg.de Silke.Grafe@uni- wuerzburg.de

Abstract: The usage of VR in higher education is not uncommon anymore. However, concepts are mainly still focusing on technical rather than pedagogical aspects of VR in the classroom. The exploration of the expectations of teacher educators as well as of preservice teachers appears indispensable (1) to achieve a sound understanding of requirements, (2) to identify potential design spaces, and finally (3) to create and to derive suitable pedagogical approaches for VR in initial teacher education. This paper presents results of guideline-based qualitative interviews comparing the expectations of teacher educators and of preservice teachers regarding teaching and learning in immersive virtual learning environments. The results showed that preservice teachers and teacher educators expect VR to enrich classes through interactive engagement in situations that would otherwise be too costly or dangerous. Regarding the design, teacher educators put the emphasis on functionality. Student teachers emphasized that they do not want to miss social interactions with their peers. Furthermore, both groups stated preferred modes of collaboration and interaction taking into account the characteristics of a virtual learning surrounding such as being able to use diverse learning spaces for group work. Interviewees agreed on two vital factors for effective learning and teaching processes: flexibility and the possibility of customization considering technical properties that are to deal with. Apart from this, preservice teachers emphasized strongly their worries about data usage and the ethics regarding using avatars and agents for representation.

Keywords: initial teacher education, virtual reality, teacher education, educational technology, competency-based teaching, media pedagogical competencies

Introduction

The increased need to use digital media in classrooms induces a modification of the requirements and demands regarding teaching competencies. These competencies also have to cover a responsible and efficient media usage (e.g. KMK, 2016; U.S. Department of Education, Office of Educational Technology, 2016). The need to acquire such media pedagogical competencies for the beneficial pedagogical use of new technology in the classroom is comprehensive. It turns-up for most groups and stages in the teaching system, e.g., for preservice teachers as well as for teacher educators (Herring, Thomas, & Redmond, 2014). As a consequence, learning scenarios at universities need to integrate innovative concepts that promote the usage and reflection of digital media in initial teacher education (Borthwick & Hansen, 2017).

Media such as VR can offer a promising potential for fostering media pedagogical competencies in initial teacher education programs. Yet, as with any medium, the sheer integration of VR in teaching and learning does not guarantee an additional value or improved learning success. A growing number of research works confirms its affordances (Latoschik et al., 2019) and suggests additional values when it is included into educational settings reasonably (e.g. Lamb, Hand, Etopio, & Yoon, 2019). It is essential to plan the implementation of the medium carefully and to consider several factors such as prerequisites and requirements, but also the process of iterative development to ensure successful and useful learning processes supported by VR (Huang, Rauch, & Liaw, 2010). So far studies have dealt

SITE 2020 - Online, , April 7-10, 2020

with either the students' or the teachers' perspective. Seldom both views are taken into account (Radianti, Majchrzak, Fromm, & Wohlgenannt, 2018).

Against this background, this study, as the preliminary work of an iterative research process, seeks answers from both teacher educators as well as preservice teachers for the question: What do student teachers and teacher educators expect of a successful virtual reality application in Initial Teacher Education (ITE)? This research question is highly relevant as the design and use of Virtual Reality (VR) in the curriculum of initial teacher education as a learning and teaching tool is gaining in importance in higher education (Adams et al., 2017).

Implementing VR in Initial Teacher Education

Focusing on post-secondary Education, Concannon, Esmail and Roberts (2019) found that VR has mainly been implemented in educational disciplines like Science and Tech as well as Health Sciences. Regarding the pedagogical perspective, works miss setting standards for teaching and learning in VR (Fowler, 2014).

As the main focus of this study deals with learning and teaching processes, central components that are linked to them have to be taken into account: These include (1) learning prerequisites, (2) learning objectives, (3) teaching and learning activities, technology and social forms (Tulodziecki, Herzig, & Grafe 2019). These criteria will be used to systematize the following literature review adopting both perspectives, i.e., the teacher educators' and the students'.

Regarding the learning prerequisites, teacher educators often do not have sufficient knowledge, skills and competencies of how to use VR in seminars (Goktas, Yildirim, & Yildirim, 2009; Uerz, Volman, & Krai, 2018). They need continuous professional development to be able to foster preservice teachers' competencies in using digital media pedagogically in the classroom (Voogt, Fisser, Pareja Roblin, Tondeur, & van Braak, 2013; Foulger, Graziano, Schmidt-Crawford, & Slykhuis, 2017), for example in the form of further training courses. However, as research and practice of using immersive VR technology in teacher education is scarce, but expanding (Billingsley, Smith, Smith, & Merrit, 2019), at some departments teacher educators with rich experiences can be found, too. They are a rich source for learning about expectations on how to implement immersive VR successfully in teacher education programs and will be addressed in this study. With regard to preservice students it can be stated, that they see the advantages of implementing VR in the classroom, but they lack self-efficacy to use it themselves (Browne & Cooper, 2000).

With regard to learning objectives in teacher education, the demand for digital competencies increases with the rise of new technologies, such as VR (Borthwick & Hansen, 2017). Since preservice teachers have to be prepared for the pedagogical integration of media in class, also their educators should acquire the corresponding knowledge (Kay, 2006; U.S. Department of Education, Office of Educational Technology, 2017). Preservice teachers' and teacher educators' requirements differ in regard to the way of acquiring and fostering media pedagogical competencies (Krumsvik, 2014). Models addressing preservice teachers' and teacher educators' competency acquisition may serve as a theoretical basis for defining learning objectives for using fully immersive VR in teacher education. With regard to (student) teachers' competencies the Technological Pedagogical Content Knowledge Framework [TPACK] (Mishra & Koehler, 2006), the Digital Competence Framework for Educators [DigCompEdu] (Redecker, 2017), the UNESCO ICT Competency Framework for Teachers (UNESCO, 2018) and the Media Pedagogical Competencies Model [M³K] (Herzig, Martin, Schaper, & Ossenschmidt, 2015; Tiede & Grafe, 2016) can serve as systematic frameworks. With reference to teacher educators' competencies the Teacher Educator Technology Competencies [TETCs] framework (Foulger, Graziano, Schmidt-Crawford, & Slykhuis, 2017) and the Media Literacy Reference Framework for learners, teachers and teacher educators [Media Didactica] (Meeus, Van Ouytsel, Driesen, & T'Sas, 2014) are the only frameworks which address teacher educators as a target group explicitly. However, the authors of the DigCompEdu Framework claim to cover educators of all stages (Redecker, 2017). For the use of VR in teacher education learning objectives which address teaching and learning in fully immersive VR explicitly have to be derived from these frameworks

Having dealt with the learner's prerequisites and their learning objectives, the question arises how learning and teaching processes using different social forms can beneficially be influenced by the technology of fully immersive VR. To answer this question, firstly the technical main characteristic that distinguishes fully immersive VR from other VR systems, such as desktop-based VR, has to be mentioned. Fully immersive VR can be described by Biocca's and Delaney's (1995) definition of VR as "the sum of the hardware and software systems that seek to perfect an all-inclusive, sensory illusion of being present in another environment" (pp. 57- 124). A recent article by Skarbez et al. (2017) includes an up-to-date discussion of immersion and presence. They propose a model that distinguishes between different qualia of VR systems, i.e., presence being composed of and affected by the social presence, plausibility, and place illusions, where the place illusion is a function of immersion as an objective

SITE 2020 - Online, , April 7-10, 2020

characteristic of a virtual experience. The multi-sensory experience of VR leads to the brain's interpretation of the virtual stimuli as real world's stimuli (Kilteni et al., 2015). The resulting full immersion of the user into another world that pretends to be real brings with it the possibility for educators to create learning scenarios that otherwise would be difficult or impossible to integrate into the real world classroom (Grenier et al., 2015). To benefit from its nature, in higher education, fully immersive VR has mainly been implemented for the purpose of training skills (e.g. Moro et al., 2017) or promoting interactivity (e.g. Lamb et al. 2018). In their literature review Radianti et al. (2019) found that 68% of works with VR implemented in higher education did not state the underlying learning theories such as cognitivism (cf. Dede, 2009) or constructivism (cf. Sharma, Agada, & Ruffin, 2013). Also, very few authors give explicit suggestions or best practices regarding learning and teaching processes in fully immersive learning environments. Huang, Rauch and Liaw (2018) for example suggest, in their case study about learners' attitudes towards VR, to use the advantages of the immersive nature of VR with the help of a constructivist learning approach. Starting from the understanding of learning processes in virtual learning environments that is based upon elementary aspects of a constructivist learning theory (cf. Shih & Yang, 2008) the authors propose theory-guided five learning strategies for instructional designers: (1) Situated learning, (2) Role playing, (3) Cooperative/ collaborative learning, (4) Problem-based learning and (5) Creative learning. These derive from combining constructivist's elements with fundamental features of VR, such as immersion, interaction and imagination (cf. Burdea & Coiffet, 2013). The practical implementation and the effects of the learning strategies on the learning outcomes, however, have not been researched so far and still represent a desideratum in literature. In general, works focus on either the technical or pedagogical characteristics of VR. Radianti et al. (2019) suggest for future research to complement technology with pedagogy and vice versa as well as to combine the teachers' and students' perspectives. So far, works concentrated on one target group led to a one-sided evaluation (ibid.).

Interview Research Methodology

To systematically explore the requirements of preservice teachers and teacher educators an analysis of needs was conducted with the two target groups using guideline-based qualitative interviews (Przyborski & Wohlrab-Sahr, 2014). The central components of teaching and learning with technology (Tulodziecki et al., 2019) served as an orientation for the development of the guideline-based interviews. Furthermore, it was important to achieve a sound understanding of technical requirements for the VR system to identify potential design spaces. In addition, it was necessary to investigate possible pedagogical and technical forms of support for the successful implementation of VR in Initial Teacher Education:

- 1) Prior experiences of preservice teachers and teacher educators with VR;
- Assumptions about potential teaching and learning scenarios in VR to achieve different goals in initial teacher education;
- Assumptions about the characteristics of these teaching and learning scenarios in VR in initial teacher education;
- 4) Technical requirements for the VR system itself but also for the ways of communicating in a virtual environment; and
- 5) Support, technically and pedagogically, that both groups might need to use VR in class

The interviews were conducted with two convenience samples of $n_1 = 12$ preservice teachers and $n_2 = 10$ teacher educators from a university in Germany. The target groups were chosen intentionally from the same department as both groups experience the same pedagogical implementation of digital media in seminars.

The preservice teachers and the teacher educators were expected to have different levels of experiences regarding the use of VR in their leisure time and in teacher education. Based on the specific focus on digital media and VR in teacher education of this department, however, the prior experience of teacher educators with and exposure to digital media and VR in particular can be expected to be richer than the average of other German institutions. As mentioned above, they are a helpful source for learning about expectations on how to implement immersive VR successfully in teacher education. The varied expectations based on their differing experiences of teacher educators and student teachers in the sample is considered to be very valuable to get a holistic image of the combination of technical and pedagogical aspects involved in ongoing teaching and learning processes.

To be able to compare the perspectives of teacher educators and student teachers, the two interview guidelines for student teachers and teacher educators were designed widely congruently. Both guidelines share the same main categories but vary in elements because of the differing characteristics considering the nature of a teacher's and learner's role. For example, teacher educators were asked about possible scenarios for further education concepts

SITE 2020 - Online, , April 7-10, 2020

while student teachers' interview focused on full seminars. Each person of both groups was interviewed individually for the duration of an hour. Participants were contacted via email and chose to participate voluntarily. Due to organizational reasons, the teacher educators had to be interviewed via an online video call, while student teachers were interviewed in person. However, the interview procedure was identical to minimize possible differences between the online interview and the face-to-face interview.

The interviews were recorded, transcribed and analyzed by means of qualitative content analysis (Mayring, 2015). They were all coded using MAXQDA (Rädiker & Kuckartz, 2019). The following categories were determined deductively in advance, based on the shared main foci of the two interview guidelines and following the approach of Mayring (2015):

- 1. Experiences with VR
- 2. Potential teaching and learning scenarios in VR
- 3. Characteristics of teaching and learning scenarios in VR
- 4. Technical requirements
- 5. Forms of support for teacher educators and student teachers

Findings

The interviews conducted with preservice teachers and teacher educators revealed a number of findings on the expectations of both groups regarding the design of learning and teaching processes in VR in educational contexts. In the following, selected findings will be summarized systematically based on the derived categories.

1. Experiences with VR

The preservice teachers in the sample had few learning opportunities with VR in their studies. However, they used VR in their leisure time on gaming consoles. Having gained teaching experiences with digital media in the local initial teacher education program, teacher educators brought along multiple prior experiences with VR that range from implementing VR systems in seminars to promote competences (e.g classroom management strategies) on a regular basis, to using VR applications only from now and then as tools for designing classes. Approximately half of the sample of the teacher educators described using VR in their seminars with preservice teachers. The other half, however, although having basic knowledge about VR, has few or no practical experiences implementing VR in seminars.

2. Potential teaching and learning scenarios in VR

Preservice teachers imagined VR to be useful in the classroom for the presentation of information, e.g., "how does an active volcano work?", that otherwise would not be possible, too costly or too dangerous to show to students. In the teacher educator interviews, participants developed various potential scenarios for using VR in initial teacher education. Phases of regular seminars were mentioned, e.g., group work phases or presentations. The teacher educators also suggested using VR to display locations and objects which would otherwise not be accessible. Also, role plays in VR were mentioned to allow for practice in challenging situations.

3. Characteristics of teaching and learning scenarios in VR

Avatar representations

The design of the avatars as virtual representations of the participants was an issue controversially discussed for both preservice teachers and for teacher educators. Opinions in both target groups ranged from a preference of realistic representations to a tendency for abstract representations.

Several preservice teachers felt uncomfortable with the thought of an avatar that resembles too much a human being in its appearance and preferred abstract versions. They reported that being able to tell the virtual world apart from the real world gives them a feeling of security. The ethical and moral aspects of avatar representation was emphasized throughout the interviews. Main concerns referred to the possible consequences of data protection and personal struggles with discrepancies between self-image and avatar representation.
On the one hand teacher educators are in favor of a realistic avatar representation, they argued, e.g., that realistic representations could increase the potential for identification. On the other hand, abstract representations were preferred by some participants because they expected easier access to the virtual learning environment and a lower risk for cyberbullying.

Room design

For designing the learning environment, preservice teachers put spaces for collaboration as a priority. Several of them preferred a classroom that offers a range of learning spaces where groups could gather and work together on projects. They put an emphasis on the social interaction with their fellow students. In their opinion, the learning surroundings should be held simple and clean in style and thus would not distract from the central purpose of the scenario.

The superordinate room design criterion for teacher educators was flexibility. The environment should be easily adaptable according to their own needs and preferences, ranging from purist designs to playful, friendly and motivating. Examples mentioned in this context included references to subject contents, such as Ancient Rome environments for History, or a virtual gym for Physical Education students. Also, in the context of furnishing, flexibility plays a vital role for them. Overall, the virtual learning arrangement needs to reflect the pedagogical approaches and methods applied by teacher educators, which means that it should offer options for easy rearrangement and for spawning and de-spawning tables, chairs, media, and other furniture items.

Communication and interaction in the virtual room

From the preservice teachers' view, communication and interaction in the learning environment were important aspects that should work in VR. They suggested how the communication and interaction could be supported by measures like, for example, implementing a mechanism that signals to the participants, either visually or auditory, who is speaking, or giving the chance of using emojis to convey moods or feelings.

With regards to communication, teacher educators emphasized the necessity to control and regulate the auditive range for specific participants or groups. For example, it was suggested to have members of a group only hear each other without distractions from other groups. Also, teacher educators wanted to be able to limit or extend their range of addressees. More specifically, this means being able to select whether an input is to be heard by a single addressee, by a group or by the whole audience.

Teaching and learning methods

Basically, teacher educators and preservice teachers expressed the need to replicate practices known from face to face teaching settings also in the virtual environment. This applies to teaching and learning methods, such as open and constructivist learning formats, and changing social formats.

Preservice teachers stated the advantage of dividing learning content into several workload units and the importance of using learning methods that involve social interaction. They see the benefit of working closely together with preservice teachers that focus on other school forms than themselves in their studies. Teacher educators again emphasized the importance of flexibility and wanted to have their student teachers work collaboratively in flexible social forms in the virtual environment. This includes working on one's own, in groups or in a plenary with the possibility to change formats easily. However, group works were prioritized in a majority of cases.

Media

A central matter of concern for both groups, teacher educators as well as preservice teachers, referred to the question of writing in the virtual room. As participants do not have a keyboard at their disposal when moving in the virtual space, suggestions to compensate for this include, e.g., virtual keyboards, speech to text transcription, or handwriting recognition. Some teacher educators also thought about designing learning scenarios without writing and notes at all, while others considered this a serious constraint.

-1840-

4. Technical requirements

Both for preservice teachers and teacher educators, technical feasibility was a frequently mentioned request. According to the interviews, hardware and software have to be intuitive and user-friendly. The hardware should be portable and ergonomic. Furthermore, the software should be stable and well designed and support immersion. It is also desirable to acknowledge aspects of inclusive design, e.g., to account for visual and auditory impairments.

5. Forms of support/ desirable knowledge and competency acquisition

Overall, both preservice teachers and teacher educators expressed a need for manifold support. Formats mentioned include tutorials, workshops, administrative support, and supportive feedback and helping functions within the virtual environment.

The preservice teachers wished to be accompanied closely in their exploration and use of the virtual environment, e.g., by supportive staff. Great importance was put on the face-to-face introduction and supervision by teacher educators.

For teacher educators, the following areas had been identified as potentially important for their own continuing education: 1) Technological skills, i.e., the skills necessary to operate and handle the respective devices and troubleshooting; 2) Application scenarios and best practice examples; 3) Methodology and teaching and learning approaches applicable in VR; 4) Attitudes and knowledge concerning VR in education; and 5) Legal, social and ethical aspects. Teacher educators discussed these proposed contents for continuing education rather controversially. There was common consent towards the importance of technological skills, application scenarios and best practice examples, and attitudes and knowledge concerning VR. However, opinions diverged concerning covering the topic of methodology and teaching and learning methods. While some teacher educators considered this a core constituent of their continuing education, others were confident to be able to integrate the virtual reality application also on their own without specific pedagogical training, as long as other aspects such as handling and technical skills are assured.

Discussion and Implications

The findings summarized above are subject to certain limitations. With regards to the samples of the study, it is important to note that both groups were convenience samples of the local department of educational sciences. Hence, against the background of the qualitative research approach and sampling method the interviewees are not representative of their respective groups and thus the results may not apply to other preservice teachers and teacher educators in the same way.

Against this background, a number of conclusions can be drawn from the interview results. Overall, it becomes evident that both the samples of preservice teachers and of teacher educators show considerable heterogeneity in certain cases with regards to their ideas and requirements. With regards to application scenarios, this heterogeneity led both samples to construct familiar learning settings on the one hand and to extend these to new contexts on the other hand. Also, the references to avatar representation illustrate how personal ideas and preferences shape the demands future users of a virtual reality environment bring along. There is no clear tendency in either of the groups to prefer abstract or realistic avatars. Notably, this finding corresponds to controversial findings from related research, where the effects of the avatar design vary as well depending on the outcome of interest (Latoschik et al., 2017). Additionally, in case of room design, personal preferences significantly shaped heterogeneous ideas of an ideal learning environment within the virtual room.

These observations substantiate the conclusion – which was also suggested by teacher educators in particular – that flexibility appears as a key criterion for the design and feasibility of a virtual learning environment to be used in initial teacher education. Teaching and learning scenarios are highly diverse and depend on a dense network of factors with regards to claims that the virtual environment has to fulfill.

Furthermore, it is noteworthy that both preservice teachers and teacher educators described teaching and learning scenarios in close connection to the conditions known from face to face teaching settings. This refers, e.g., to the design of the classroom, to learning formats, and to tools and media required, where the participants favored a realistic replication of face to face settings. Consequently, restraints from a technical perspective, e.g., with regards to the issue of writing or handwriting, were partly considered a serious limitation. Several teacher educators claimed to

refuse a redesign of their methods and approaches to account for the new circumstances proposed by the virtual environment. Also, innovative ideas and approaches to teaching and learning were comparably scarce.

Hence, it appears a research desideratum for future studies to balance the innovative potential of learning scenarios in virtual reality and to respond to the demands expressed by potential users at the same time. Designing the environment in accordance with these demands is likely to increase acceptance and feasibility but at the same time limits the potential inherent in virtual reality for teaching and learning purposes.

In terms of technical requirements, both samples emphasized the importance of accessibility and user friendliness. Acknowledging this focus will account for the varying levels of skills both student teachers and teacher educators bring along with regards to the operation of digital media. Finally, considering the results on desirable forms of support, it became evident that both preservice teachers and teacher educators need to be supported and accompanied closely in their acquisition of respective competencies. As identified in the interviews, there are multiple ways to ensure that the competencies needed are acquired adequately, as, e.g., continuing professional development or administrative support. In the case of virtual reality in teacher education, such an extensive support appears especially important due to the different facets of competencies that need to be addressed with regards to educational and pedagogical competencies on the one hand and technical skills on the other.

Further research perspectives are conceivable. To substantiate the findings that build on subjective opinions, it will be necessary to triangulate methods and to contextualize the initial requirements of preservice teachers and teacher educators with experiences and data collected in the actual implementation of a virtual reality environment in initial teacher education programs. Against the background of the interview findings, it appears relevant not only to design immersive virtual learning environments but also to develop pedagogical concepts based on theory and empirical data to advance the competencies of preservice teachers and teacher educators have for teaching and learning in virtual reality offer significant insights into aspects to be considered in this context. In accordance with emerging literature from the perspective of educational research (Merchant, Goetz, Cifuentes, Keeney-Kennicutt, & Davis, 2014; Southgate et al., 2019), it will be insightful to expand the perspective and to evaluate teaching and learning processes in virtual reality not just from a technical, but also from an educational and pedagogical perspective, based on the foci proposed above. This way, a pedagogical design-based research approach (e.g. Tulodziecki et al., 2013) can contribute to the further exploration of virtual reality in teaching and learning processes to make sure that future teachers can benefit from the potential VR offers.

References

Adams, B. S., Cummins, M., Davis, A., Freeman, A., Hall Giesinger, C., & Anantharayanan, V. (2017). NMC Horizon Report: 2017 Higher Education Edition. Austin, TX: The New Media Consortium.

Billingsley, G., Smith, S., Smith, S., Smith, S., & Meritt, J. (2019). A Systematic Literature Review of Using Immersive Virtual Reality Technology in Teacher Education. *Journal of Interactive Learning Research* 30(1), 65–90. Waynesville, NC: Association for the Advancement of Computing in Education (AACE). Retrieved March 14, 2020 from https://www.learntechlib.org/primary/p/176261/.

Borthwick, A. C., & Hansen, R. (2017). Digital Literacy in Teacher Education: Are Teacher Educators Competent? *Journal of Digital Learning in Teacher Education*, 33(2), 46–48. https://doi.org/10.1080/21532974.2017. 1291249

Browne, N., & Cooper, E. (2000). Pedagogically appropriate integration of informational technology in an elementary preservice teacher education program. *Journal of Technology and Teacher Education*, 8(3), 219–229.

Burdea, G. C., & Coiffet, P. (2003). *Virtual reality technology* (2nd ed.). New York, NY: John Wiley & Sons. Concannon, B. J., Esmail, S., & Roduta Roberts, M. (2019). Head-Mounted Display Virtual Reality in Post-

secondary Education and Skill Training. Frontiers in Education, 4. https://doi.org/10.3389/feduc.2019.00080
Dawley, L., & Dede, C. (2013). Situated Learning in Virtual Worlds and Immersive Simulations. In J. M. Spector,
M. D. Merrill, J. Elen, & M. J. Bishop (Eds.), Handbook of Research on Educational Communications and

Technology (pp. 723–734). New York, NY: Springer. https://doi.org/10.1007/978-1-4614-3185-5_58 Dede, C. (2009). Immersive interfaces for engagement and learning. *Science* 323, 66–69. doi: 10.1126/science.1167311

Fowler, C. (2014). Virtual reality and learning: Where is the pedagogy? British Journal of Educational Technology, 46(2), 412–422. https://doi.org/10.1111/bjet.12135

- Foulger, T., Graziano, K. J., Schmidt-Crawford, D. A., & Slykhuis, D. A. (2017). Teacher Educator Technology Competencies. Journal of Technology and Teacher Education, 25(4), 413–448.
- Goktas, Y., Yildirim, Z., & Yildirim, S. (2009). Investigation of K-12 Teachers' ICT Competencies and the Contributing Factors in Acquiring these Competencies. *The New Educational Review*, *17*(1), 276–294.
- Grenier, S., Forget, H., Bouchard, S., Isere, S., Belleville, S., Potvin, O., & Talbot, M. (2015). Using virtual reality to improve the efficacy of cognitive-behavioral therapy (CBT) in the treatment of late-life anxiety: preliminary recommendations for future research. *Int. Psychogeriatr.*, 27, 1217–1225. doi: 10.1017/S1041610214002300
- Herring, M., Thomas, T., & Redmond, P. (2014). Special Editorial: Technology Leadership for Preparing Tomorrow's Teachers to Use Technology. *Journal of Digital Learning in Teacher Education*, 30(3), 76–80. https://doi.org/10.1080/21532974.2014.891875
- Herzig, B., Martin, A., Schaper, N., & Ossenschmidt, D. (2015). Modellierung und Messung medienpädagogischer Kompetenz – Grundlagen und erste Ergebnisse. In B. Koch-Priewe, A. Köker, J. Seifried, & E. Wuttke (Eds.), Kompetenzerwerb an Hochschulen: Modellierung und Messung. Zur Professionalisierung angehender Lehrerinnen und Lehrer sowie frühpädagogischer Fachkräfte (pp. 153–176). Bad Heilbrunn, Germany: Klinkhardt.
- Herzig, B., Schaper, N., Martin, A., & Ossenschmidt, D. (2016). Schlussbericht zum BMBF Verbundprojekt M³K Modellierung und Messung medienpädagogischer Kompetenz, Teilprojekt Medienerzieherische und mediendidaktische Facetten und handlungsleitende Einstellungen. Paderborn, Germany: Universität Paderborn.
- Huang, H. -M., Rauch, U., & Liaw, S. -S. (2010). Investigating learners' attitudes toward virtual reality learning environments: Based on a constructivist approach. *Computers & Education*, 55(3), 1171–1182. https://doi.org/10.1016/j.compedu.2010.05.014
- Kay, R. H. (2006). Evaluating strategies used to incorporate technology into pre-service education: A review of the literature. *Journal of Research on Technology in Education and Information Technologies*, 38(4), 383–408.
- Kilteni, K., Bergstrom, I., & Slater, M. (2013). Drumming in immersive virtual reality: the body shapes the way we play. *IEEE Trans. Visual. Comput. Graphics*, 19, 597–605. doi: 10.1109/VR.2013.6549442
- Krueger, R. A., & Casey, M. A. (2015). Focus Groups. A Practical Guide for Applied Research (5th ed.). Los Angeles, CA: Sage.
- Krumsvik, R. J. (2014). Teacher Educators' Digital Competence. Scandinavian Journal of Educational Research, 58(3), 269–280. https://doi.org/10.1080/00313831.2012.726273
- Lamb, R., Antonenko, P., Etopio, E., & Seccia, A. (2018). Comparison of virtual reality and hands on activities in science education via functional near infrared spectroscopy. *Computers & Education* 124, 14–26. doi: 10.1016/ j.compedu.2018.05.014
- Lamb, R., Etopio, E., Hand, B., & Yoon, S. (2019). Virtual reality simulation: Effects on academic performance within two domains of writing in science. *Journal of Science Education and Technology*, 1-11.
- Latoschik, M. E., Roth, D., Gall, D., Achenbach, J., Waltemate, T., & Botsch, M. (2017). The effect of avatar realism in immersive social virtual realities. *Proceedings of the 23rd ACM Symposium on Virtual Reality* Software and Technology - VRST '17. https://doi.org/10.1145/3139131.3139156

Latoschik, M. E., Kern, F., Stauffert, J.-P., Bartl, A., Botsch, M., & Lugrin, J.-L. (2019). Not Alone Here?! Scalability and User Experience of Embodied Ambient Crowds in Distributed Social Virtual Reality. *IEEE Transactions on Visualization and Computer Graphics*, 25(5), 2134–2144. https://doi.org/10.1109/tvcg.2019.2899250

Mayring, P. (2015). Qualitative Content Analysis: Theoretical Background and Procedures. In A. Bikner-Ahsbahs, C. Knipping, & N. Presmeg (Eds.), *Approaches to Qualitative Research in Mathematics Education* (pp. 365– 380). Dordrecht, Netherlands: Springer. https://doi.org/10.1007/978-94-017-9181-6 13

- Meeus, W., Van Ouytsel, J., Driesen, A., & T'Sas, J. (2014). Media Didactica: A Media Literacy Reference Framework for Learners, Teachers and Teacher Educators. *Merz medien + erziehung*, 58(6), 41–49.
- Merchant, Z., Goetz, E. T., Cifuentes, L., Keeney-Kennicutt, W., & Davis, T. J. (2014). Effectiveness of virtual reality-based instruction on students' learning outcomes in K-12 and higher education: A meta-analysis. *Computers & Education*, 70, 29–40. http://dx.doi.org/10.1016/j.compedu.2013.07.033
- Mishra, P., & Koehler, M. J. (2006). Technological pedagogical content knowledge: A framework for integrating technology in teachers' knowledge. *Teachers College Record*, 108(6), 1017–1054. https://doi.org/10.1111/j.1467-9620.2006.00684.x
- Moro, C., Štromberga, Z., Raikos, A., & Stirling, A. (2017). The effectiveness of virtual and augmented reality in health sciences and medical anatomy. *Anat. Sci. Educ.* 10, 549–559. https://doi: 10.1002/ase.1696
- Parong, J., & Mayer, R. E. (2018). Learning science in immersive virtual reality. *Journal of Educational Psychology*, 110(6), 785–797. https://doi.org/10.1037/edu0000241

Przyborski, A., & Wohlrab-Sahr, M. (2013). Qualitative Sozialforschung: Ein Arbeitsbuch. München, Germany: Oldenbourg.

Radianti, J., Majchrzak, T. A., Fromm, J., & Wohlgenannt, I. (2020). A systematic review of immersive virtual reality applications for higher education: Design elements, lessons learned, and research agenda. *Computers & Education*, 147, 103778. https://doi.org/10.1016/j.compedu.2019.103778

Rädiker, S., & Kuckartz, U. (2018). Analyse qualitativer Daten mit MAXQDA: Text, Audio und Video. Wiesbaden, Germany: Springer Fachmedien.

Redecker, C. (2017). European Framework for the Digital Competence of Educators: DigCompEdu. Punie, Y. (ed). Luxembourg, Luxembourg: Publications Office of the European Union. https://doi.org/10.2760/159770

Sharma, S., Agada, R., & Ruffin, J. (2013). Virtual reality classroom as a constructivist approach. In 2013 proceedings of IEEE southeastcon (pp. 1–5). http://dx.doi.org/10.1109/SECON.2013.6567441

Shih, Y.-C., & Yang, M.-T. (2008). A collaborative virtual environment for situated language learning using VEC3D. Educational Technology & Society, 11(1), 56–68.

Skarbez, R., Brooks, Jr., F. P., and Whitton, M. C. (2017). A survey of presence and related concepts. ACM Comput. Surv., 50(6):96:1–96:39.

Southgate, E., Smith, S. P., Cividino, C., Saxby, S., Kilham, Eather, G., ... Bergin, C. (2019). Embedding immersive virtual reality in classrooms: Ethical, organisational and educational lessons in bridging research and practice. *International Journal of Child-Computer Interaction*, 19, 19–29. https://doi.org/10.1016/j.ijcci.2018.10.002

Ständige Konferenz der Kultusminister der Länder in der Bundesrepublik Deutschland [KMK]. (2016). Bildung in der digitalen Welt. Strategie der Kultusministerkonferenz. Berlin, Germany.

Taylor, S. J., Bogdan, R., & DeVault, M. (2015). *Introduction to qualitative research methods: A guidebook and resource*. Hoboken, NJ: John Wiley & Sons.

Tiede, J., & Grafe, S. (2016). Media Pedagogy in German and U.S. Teacher Education. Comunicar, 24(49), 19–28. https://doi.org/10.3916/c49-2016-02

Tulodziecki, G., Grafe, S., & Herzig, B. (2013). Gestaltungsorientierte Bildungsforschung und Didaktik: Theorie -Emprie - Praxis. Bad Heilbrunn, Germany: Klinkhardt.

Tulodziecki, G., Herzig, B., & Grafe, S. (2019). Medienbildung in Schule und Unterricht: Grundlagen und Beispiele. 2nd ed. Bad Heilbrunn, Germany: Klinkhardt/ UTB.

- Uerz, D., Volman, M., & Kral, M. (2018). Teacher educators' competences in fostering student teachers' proficiency in teaching and learning with technology: An overview of relevant research literature. *Teaching and Teacher Education*, 70, 12–23. https://doi.org/10.1016/j.tate.2017.11.005
- United Nations Educational, Scientific and Cultural Organization [UNESCO]. (2018). UNESCO ICT Competency Framework for Teachers. Paris, France: Author. Retrieved from https://en.unesco.org/
- U.S. Department of Education, Office of Educational Technology. (2016). Future Ready Learning: Reimagining the Role of Technology in Education. 2016 National Education Technology Plan. Washington, DC: Author. Retrieved from https://tech.ed.gov/files/2015/12/NETP16.pdf

Voogt, J., Fisser, P., Pareja Roblin, N., Tondeur, J., & van Braak, J. (2013). Technological pedagogical content knowledge–a review of the literature. *Journal of computer assisted learning*, 29(2), 109–121.

Yeh, S. -C., Tsai, C. -F., Fan, Y. -C., Liu, P. -C., & Rizzo, A. (2012). An innovative ADHD assessment system using virtual reality. 2012 IEEE-EMBS Conference on Biomedical Engineering and Sciences. https://doi.org/10.1109/IECBES.2012.6498026

Zhang, L. -L., Wang, J. -Q., Qi, R. -R., Pan, L. -L., Li, M., & Cai, Y. -L. (2015). Motion Sickness: Current Knowledge and Recent Advance. CNS Neuroscience & Therapeutics, 22(1), 15–24. https://doi.org/10.1111/cns.12468

3.3 Conclusion

In this chapter, teacher educators' and pre-service teachers' requirements for a pedagogical concept to prepare both target groups for the successful implementation of VR in ITE were analyzed and evaluated. Insights gained from existing theoretical and empirical research, as well as from conducting qualitative interviews with teacher educators and pre-service students, allowed the first design hypotheses for the pedagogical concept to be derived (see Euler, 2014) using social VR in ITE. The results of the study showed that if neither educators nor students have gained sufficient experience of fully immersive VR in class, there should be close guidance and support when integrating VR hardware and software in seminars. The material and learning surroundings should be accessible and user-friendly. Moreover, teacher educators agreed that the pedagogical concept should consider elements of constructivist learning theory (Section 3.2); these empirical results are in accordance with the existing scientific literature (Huang & Liaw, 2018). Furthermore, both groups emphasized the need for disruption-free communication, collaboration, and interaction in social VR, imagining the possibility of varying social formats in group work (Section 3.2). Further, in line with cognitive load theory (Chandler & Sweller, 1991; Parong & Mayer, 2021), the teacher educators pointed out that design elements should be kept minor and clean so the social VR's surroundings do not distract students from learning (Section 3.2).

Since the objective of this research was to develop a pedagogical concept fostering pre-service teachers' TPACK in social VR, the first study explored important contextual factors, in particular those concerning communication and interaction processes, which allowed the first design elements of the pedagogical concept to be derived. However, the design principles were preliminary and required further research and development. Necessary adaptations and improvements followed the application of the design measures in practice.

4. Preservice Teachers' Encounter with Social VR – Exploring Virtual Teaching and Learning Processes in Initial Teacher Education

4.1 Aims of Study 2

Conclusions were drawn from the results of the needs analysis (Chapter 3) regarding the design of the pedagogical concept and the social VR environment. Following Euler's (2014) recommendation, an important next step was to develop a targeted and robust intervention before evaluating the design in a broader context. Accordingly, in the second study, a pedagogical concept was developed and implemented in a seminar session in a social VR learning environment with a convenience sample of pre-service teachers at a German university (Section 4.2). Doing so enabled an exploration of how the pedagogical concept could be implemented into a fully social VR learning environment in practice, as well as the consequences for the further development of the pedagogical concept and the social VR learning environment and their interplay. Furthermore, and in line with Bower and Sturman's (2015) reminder that "if the use is there but not perceived, then it is of no educational benefit" (p. 345), the pre-service teachers' perception of the teaching and learning activities in the social VR learning environment and its affordances are of central importance. Thus, the second study investigated the following research questions:

- How do pre-service teachers perceive teaching and learning activities in fully immersive VR?
- How should teaching and learning processes using social VR in teacher education be designed?

The results led to further insights into the development cycle of the pedagogical concept to foster the TPACK of pre-service teachers and its implementation in social VR.

4.2 Study 2

SITE Interactive Online 2020 Conference - Online, , October 26-28, 2020

Preservice Teachers' encounter with Social VR -

Exploring Virtual Teaching and Learning Processes

in Initial Teacher Education

Silke Grafe	Marc Erich Latoschik
Chair of School Pedagogy	Chair of HCI
University of Wuerzburg	University of Wuerzburg
Germany	Germany
Silke.Grafe@ uni- wuerzburg.de	Marc.Latoschik@uni- wuerzburg.de
	Silke Grafe Chair of School Pedagogy University of Wuerzburg Germany Silke.Grafe@ uni- wuerzburg.de

Abstract: With 21st century challenges ahead, higher education teaching and learning need new pedagogical concepts. Technologies like social VR enable student-centered. action-oriented, and situated learning. This paper presents findings of the pedagogical implementation of a distributed social VR prototype, a fully immersive VR learning environment, into an Initial Teacher Education program in Germany. The exploratory study addressed the following research questions: 1) How do preservice teachers perceive teaching and learning activities in fully immersive VR and 2) how should teaching and learning processes using social VR in Teacher Education be designed? It followed a design-based research approach. The pedagogical concept for teaching and learning in social VR was based on principles of action-orientation. A convenience sample of three groups of five students each took part in a 90-minute teaching and learning scenario using a fully immersive VR learning environment. During these seminar units, students engaged in qualitative group interviews and shared their perception of the action-oriented teaching and learning activities in VR. The results showed that preservice teachers had the feeling of being less distracted in social VR. Additionally, during group activities, missing social and behavioral cues made communication procedures more challenging for participants. However, some participants noticed a stronger sense of community while collaborating with others.

Introduction/ Study context

During the pandemic, the transition from classroom teaching to virtual instruction has emphasized the necessity to promote teachers' pedagogical media competencies from early stages on in Teacher Education (Ferdig et al., 2020).. Already existing demands for Information and Communication Technology infrastructures, innovative concepts embracing technology infusion, and action orientation in seminars are still pressing (Foulger et al., 2019). However, to meet expectations that envision teachers to be "fully capable of taking advantage of technology to transform learning" (US Department of Education, 2017) and for additional pedagogical value and improved learning outcomes, the sole implementation of technology in classes is not sufficient (Blömeke, 2017; Eickelmann et al., 2019, Foulger et al., 2017, Krumsvik, 2014). The effective use of emerging technologies for pedagogical purposes also includes promoting preservice teachers' media-pedagogical and ICT-related competencies (Blömeke, 2017; Eichelberger & Leong, 2019; Tondeur et al., 2019; Tulodziecki et al., 2017, Tiede & Grafe 2019). Consequently, learning scenarios in higher education need to offer opportunities for teacher candidates to gain in-practice experiences using and

reflecting on digital media in initial teacher education (Borthwick & Hansen, 2017). Social VR is already being used actively in education and training programs and could promote constructivist learning environments that embrace situated and action-oriented learning in teacher education (Dawley & Dede, 2014; Hellriegel & Cubela, 2018). However, there is still a lack of studies dealing with the design and development of pedagogical concepts and best practices for social VR in teacher education.

Against this background, this exploratory research study focuses on the pedagogical integration of a fully immersive social virtual environment into a teacher education course in Germany. We aim to answer the following questions:

1) How do preservice teachers perceive teaching and learning activities in fully immersive VR?

2) How should teaching and learning processes using social VR in Teacher Education be designed?

We hope the results may be valuable for elaborating pedagogical concepts using social VR in initial teacher education.

Literature Review

Social VR

Years of enhancing immersive virtual reality technology lead to the use of VR on a mass-market scale (Perry, 2016b; Scavarelli et al., 2020). Developers and companies are still facing, however, ongoing challenges such as developing VR headsets with increased mobility, designing VR applications according to current standards, and considering ethical concerns (Slater & Sanchez-Vives, 2016). With userfriendly hardware and software becoming affordable, immersive VR attracts a wide range of consumers from different backgrounds approaching the technology with varying demands such as gaming, socializing, or collaborating in formal or informal contexts.

When emphasizing the use of VR's potentials to promote virtual and synchronous social collaboration and interaction in diverse contexts, the terms "social VR", "collaborative virtual environments", "immersive multi-user virtual environments or "collaborative VR" are sometimes used interchangeably. The virtual environments which are described by these terms share the characteristics of being " a web-based social interaction paradigm, mediated by immersive technologies and taking place in predesigned three-dimensional virtual worlds where individuals, represented by an avatar, may engage in real-time interpersonal conversation and shared activities" (Dzardanova et al., 2018).

Sharing the same definition, the term social VR, when used by commercial companies, underlines the immediate virtual environment's attachment to social media and its primary purpose of social networking. Dzardanova et al. (2018) describe the change towards social media's use in VR as "the second generation of social networking." Social VR applications like AltSpace, VRChat, or RecRoom attract increasing numbers of users that enjoy interacting with the VR experience (Gunkel, 2019). According to the authors the four key uses for social VR, in the understanding of a social networking platform, are "video conferencing, education, gaming, and watching

-550-

movies." The question, which one of the key uses, does profit the most from social VR is open to discussion (*ibid*.).

Regarding VR's usefulness for education, there is disagreement about whether the additional effort and expenses required when implementing VR in class do justice to the added value on learning outcomes (Radianti et al., 2020). Regardless of their purpose, technically, social VR platforms seek to improve as best as possible the feeling of "being there together" in VR. Whereas 2D digital media and single-user VR systems lack in mediating the whole sensory impression of social encounters, immersive embodied multi-user virtual environments (MVEs) provide the possibility of communicating and interacting with each other via avatars, thus coming close to "real/offline" social communication and interaction. People can deal with and successfully interpret a wide variety of artificial and/or augmented signals to carry social meaning (Roth et al., 2018). However, implementing head and body tracking, a faithful avatar representation, spatialized audio, and natural nonverbal communication cues lead to reinforcing a "close-to-reality "experience (Latoschik et al., 2019). Thus, the sensory impression of diving into a simulated "real" world including artificial bodies intensifies prominent VR factors like the degree of immersion, presence, and virtual body ownership (Roth et al., 2017; Roth & Latoschik, 2020). If the sensory perception of the virtual environment is designed as closely as possible to reality, the brain does not differentiate between "real" or "virtual" sensory perception and, as a consequence, interpret them as real stimuli in VR (Slater & Sanchez-Vives, 2016; Slater & Steed, 2000).

Teaching and learning processes in social VR

Social VR's potential for teaching and learning processes lies within offering authentic learning environments that enable to interact remotely and synchronously as well as to permit learning experiences that affect learners in a multi-sensory way. Schwan and Buder (2006), stressing the advantages of the authenticity and interactivity of virtual environments, categorize virtual environments according to the user's level of interactivity with the virtual world into three types (Schwan & Buder, 2006): (1) In the focus of the Exploration Worlds is the sensory experience and self-paced exploration of objects, places or rooms that otherwise would be too difficult, not feasible or too dangerous to access. Learners can walk through virtual museums or travel to places they have never been before (*ibid.*), gathering information on their own or in groups. The National Geographic "Explore VR" App, for example, enables the user to walk through the historical site of Machu Picchu, and while fulfilling short tasks on the way, the user learns about Incas' history. The virtual environment provides information in multimodal ways and can thus promote declarative knowledge acquisition. Whereas interactivity is restricted in Exploration Worlds, (2) Training Worlds offer more potential to train procedural knowledge and psychomotor skills through action-based sequences in scenarios like evacuation drills (Feng et al., 2018). The advantage of using VR to train psychomotor skills is that situations can repeatedly be simulated under the same circumstances over and over again. Otherwise, in real life, this procedure would not be possible due to high costs, efforts, or the risk to harm people. (3) Construction Worlds have the highest possibility of interactivity for learners and virtual environments. In these worlds, users themselves can create and design objects inside the virtual environment.

Increasingly, software is released that enables VR's users to create their own 3D objects or express their ideas with drawing and modeling tools. The results are available in different formats and exportable to other media. Such a categorization might be helpful to distinguish between different virtual worlds. However, the chosen terminology refers only to exploration, training, and construction, therefore concealing its potential for important other learning activities, such as problem-solving, decision making, and critical thinking. Designing pedagogical concepts for teaching and learning processes in social VR on the grounds of constructivist learning theories offer great potential for promoting various competencies of learners (Burdea & Coiffet, 2003; Dawley & Dede, 2014; Scavarelli et al., 2020; Wang, 2020).

The integration and implementation of social VR for teaching and learning purposes are connected to some challenges. For one, the development of social VR platforms takes time and is extensive regarding resources. VR software and content suitable for teaching and learning settings are rare and need further development. Concepts and best practices that introduce teachers to how to use social VR in classes with materials explaining step-by-step procedures are missing (Stavroulia et al., 2019). Also, there are technical challenges concerning the choice of hard- and software. With the development of social VR platforms and thus making social VR commercially accessible for a bigger audience, also ethical questions arise in the context of users' data protection, harassments in open social VR places, fraud or avatars' influence on one's identity (Shiram & Schwarz, 2017). This makes designing teaching and learning processes in social VR quite challenging, as not only the pedagogical view has to be considered but also the technical characteristics of social VR have to be taken into account.

By exploring teaching and learning processes in social VR and how they are perceived by preservice teachers, the present study seeks to give valuable insights for further developments of pedagogical concepts for teaching and learning in social VR. Based on the needs analysis of Ripka et al. (2020) the following conclusions have served for the development of the pedagogical concept and the social VR prototype used this study:

- 1) Design of action-oriented and constructivist teaching and learning scenarios with changing social formats
- 2) Preservice teachers are supported in their exploration and the use of the virtual environment.
- Room design of social VR must be clean and simple, allowing interactivity and social collaboration
- 4) Communication and interaction in social VR should work easily without interruptions
- 5) Avatar representations can either be abstract or real

Research Methodology

1. Study Context

The overall goals of the research project are the design of a theory and practice-based pedagogical concept for promoting media-pedagogical competencies

using social VR in Initial Teacher Education (ITE) and the interdisciplinary development of an according social VR platform. As a result of this iterative design process, the social VR platform will be integrated into seminars in ITE at university and used as a distance teaching and learning tool that promotes media-pedagogical competencies of preservice teachers. In this developing process, following up on the needs analysis of teacher educators and preservice teachers' requirements for teaching and learning in social VR (Ripka et al., 2020), this study focuses on the following research questions:

1) How do preservice teachers perceive teaching and learning activities in fully immersive VR?

2) How should teaching and learning processes using social VR in Teacher Education be designed?

2. Study design

Social VR prototype

Regarding room design, the social VR prototype, based on the game engine Unreal Engine 4.24 and optimized for Microsoft Windows 10, offered a fully immersive seminar room. The *Oculus Rift S*, a Head Mounted Display, and a Laptop served as VR hardware. To avoid unnecessary distractions for the students, the virtual seminar room is kept clean and simple, offering room for collaboration and interaction.

The implemented virtual whiteboard enables the teacher to use webpages, presentation slides, and collaborative web tools throughout the seminar session. The board can be



operated with a control board integrated into one's avatar's virtual wristband. For avatar representation, a comic-alike abstract avatar is used. Outer appearances like the color of the upper body, gender, and name are customizable for the users. After entering the preferred avatar choice, representations' names appear over the avatar's head, visible for all participants. Students are able to move around via teleporting.

Pedagogical Concept

For the preparation of students using the VR hard-and software, a face-to-face introduction had been performed before the seminar started. In addition to a hands-on approach throughout the process, with the aim in mind that students need to get used to

the VR user interface, a collaborative and interactive group activity in the virtual environment was designed. A planetary system was simulated inside the virtual seminar room, and participants would have to communicate with each other to put the planets into the right spot. To be able to closely observe teaching and learning processes and to support students with getting used to VR's hard-and software, two main choices were made. First, the group of participants for one seminar session was restricted to five. Second, the intervention took place in one big lab instead of locally distributed to support students in case of having difficulties with the technology and to ease the detection of technical malfunctions and troubleshooting.

The pedagogical concept for the 90-minute seminar session was designed based on action- and development-oriented didactics (Tulodziecki et al., 2017; Tulodziecki et al., 2019) combined with flipped classroom principles. This involved a task-based approach and collaborative learning processes in the virtual learning environment. Students prepared the learning content for the seminar session in advance according to flipped classroom principles. The seminar session was divided into two units:

(1) a teacher-centered introduction of the topic and the presentation of a situated task asking the students to design a concept for using VR in schools which was followed by a joint brainstorming (10 minutes),

(2) a learner-centered group activity in which preservice teachers designed a concept for using VR in the classroom with pupils and presented their results at the end of the session (30 minutes).

3. Methodology and data collection

The convenience sample consisted of 15 preservice teachers (3 groups of 5 students each). Due to sickness one of the students did not appear. The students had the opportunity to sign up voluntarily for the intervention which was promoted in school pedagogy courses focusing on the use of digital media in educational contexts. The participants represented a mixed group of 12 female and 3 male teacher candidates. Group 1 and group 2 consisted each of five female students. Three male and two female preservice teachers participated in group 3. Data was collected qualitatively in group interviews.

The development of the interview guide was based on the central components of teaching and learning with technology, according to Tulodziecki et al. (2019), and the requirements of teacher educators and preservice teachers (Ripka et al., 2020). Ι.

- Teacher-centered teaching and learning processes
 - a. What is your experience with the social VR application?
 - a.i. When you think of your seminars in "real" classrooms, how did you perceive the interaction with the lecturer in comparison?
 - a.ii. What advantages and disadvantages do you see in learning in social VR?
- Π. Learner-centered teaching and learning processes

- a. You have worked together in the group activity in VR. The sketch of the design task is the result of this group work. What was your experience during the group work in VR?
 - a.i. What experiences have you had during the process of finding mutual consent in VR?
 - a.ii. What was your experience in giving mutual feedback?
 - a.iii. What was your experience with consultations within the group?
 - a.iv. Which other functions in VR would have been helpful in the group activity?

The group interviews took place in-between seminar units and outside of VR. The interviews were recorded, transcribed, and analyzed through qualitative content analysis (Mayring, 2015). They were coded using MAXQDA (Rädiker & Kuckartz, 2019). The following categories were determined deductively following the approach of Mayring (2015):

- 1) Teacher-centered teaching and learning processes
 - a. Interaction between lecturers and students/ students and students
 - b. Interaction with social VR environment
 - c. Communication and language in social VR
 - d. Media pedagogical content
 - Learner-centered teaching and learning processes
 - a. Interaction between lecturers and students/ students and students
 - b. Interaction with social VR environment
 - c. Communication and language in social VR
 - d. Media pedagogical content

Also, an exploratory participatory observation was carried out to help interpret the results of the group interviews.

Results

2)

1) Teacher-centered teaching and learning processes

a. Interaction between lecturers and students/students and students When asked about how they perceived the interaction with the lecturer in social VR in comparison to "real" classrooms, one student said that it would not differ so much, "Well, I think it was not much different [...] I think you need a few minutes to get used to it. But you see and hear quite good who is talking and what it is all about" (Group 1_C). Some participants experienced a shift of focus in VR, and one explained, "I had the feeling that somehow the attention of us lay much more crassly on what was said, [...] I think the focus was just completely different because I cannot be distracted" (Group 2_C). Further, describing the perceived relationship between the students and the lecturer in social VR students commented for example: "I don't know if this is due to the format, but I thought that it wasn't a real teacher-student relationship because we were standing in a circle and also much closer together. The classroom is usually bigger, and you have more distance to the teacher and he usually stands in the front. I thought that was the group feeling that was addressed, not only including the students but also the teacher" (Group 3_D).

b. Interaction with social VR environment

Regarding their experiences with the social VR environment, some students reported unease while getting used to the VR environment: "I thought it took a lot of time to get used to it. In the beginning, I didn't know at all if this is a blackboard" (Group 1 E). Concerning the change from the "real classroom" to the VR environment one student stated: "So it was just a strange feeling to be in another room from one second to the next" (Group 2 A). Others valued that there were no distractions in the virtual seminar room: "There is no window, where I can distract myself from any people who might be walking around, [...] I don't have anything that tempts me" (Group 3 C). Emphasizing the positive effects of a mobile-phone-free learning surrounding a student said: "If I sit in a face-to-face seminar for example, then 80% of the students have their mobile phones at the table [...] but this doesn't work here, and because of that, I don't have the possibility at all, I think the focus was completely different because I can't be distracted" (Group 1 C). Referring to the effects of wearing VR headsets over a specific timeperiod, most participants agreed on the following impression: "At the beginning, I thought: Super interesting. And at some point, you realize: It's getting super tiring to look at the virtual board and to concentrate" (Group 2 C).

c. Communication and language in social VR

As abstract avatars represented participants, facial expressions, and full body gestures were missing. The consequences influenced the way of communicating and interacting in social VR. Several participants observed positive effects of missing communication cues, as these examples show: "I also agree with what you said about concentrating on the language because somehow, I don't have to concentrate on facial expressions and gestures and my appearance, and everybody is looking at me" (Group 3_B) and some felt more comfortable to talk, pointing out: "But I think you also have a little less inhibition to say something because you feel so anonymous [...]" (Group 2_B).

d. Media pedagogical content

There were no comments regarding the media pedagogical content.

2) Learner-centered teaching and learning processes

a. Interaction between lecturers and students/ students and students Regarding the interaction between students in the second collaborative unit of the session, most of the participants thought that the group work in social VR was not too challenging, as this statement shows: "Well, I thought it went very well, I didn't think that we would come to a result so quickly. I know it from other group work, where you discuss forever and then I heard a quarter of an hour, I was like "Oh god oh god", four of us, four opinions in a quarter of an hour, but it went well, so we agreed relatively quickly, and were all happy with it" (Group 3_C). Referring to the learners' proximity of the learning setting in VR compared to a real classroom one, a student thought, "it's better because somehow you have the feeling that you can't retreat. Because we all see each other. We are all standing in this circle and it would also be noticeable if one of us just wouldn't do anything" (Group 1_C). Also, during this process, the missing social cues influenced the communication between students. As one participant expressed: "I

can also imagine that when you do this in larger groups, I mean we are a really small group now, it can sometimes be a bit confusing when you have to assign the voices, because you don't see that someone is speaking, and then you are a bit confused" (Group 3_D).

b. Interaction with social VR environment

Several participants described positive consequences using the social VR environment, stating for example, "Well, I would say that I think: I'll just profit more of it. Because somehow - I don't know - I feel more like I'm present" (Group 1_C). Additionally, one participant pointed out social VR's potential being a "safe place" in that one could act without fearing consequences: "I think it is really good if you don't have any direct consequences, so you can really test a lot. And if it goes wrong, then you just do it again or something like that. Things that you could not do at all, or if you have to do them once, you do them once. I think that's a very, very big opportunity" (Group 1_A). However, other students expressed their concerns: "I think it's a difference if you put on your glasses and you have everything already in place than having to create it yourself, especially if you are technically a bit insecure [...] I think I would - personally I'm afraid of it - so I'd be really afraid, if I really could manage that - especially with students" (Group 2_A). Also, some wished to be able to take some notes: "...what I missed, was that you can't take notes. Because it takes a bit longer and everyone gives input, and so on - you can't take notes" (Group 2_D).

c. Communication and language in social VR

Similar to the teacher-centered phase, the missing social cues lead to more challenging communication: "So, what you really have to get used to: You really have to approach people directly. By name, really. You really have to get into the habit of addressing people directly by name. Because otherwise you really don't feel addressed" (Group 2_D).

d. Media pedagogical content

Regarding the lessons' content and its extent, several participants commented like this: "It would have been hard to memorize all this or not to write it down on the side. Especially when it is something bigger and longer that you can take notes on the side" (Group 2 A). Another student added:" I got more out of it than if I would have just sat there and written it down" (Group 2 C). Concerning the potential of using VR in their future classroom, students described examples of how they can imagine getting their pupils involved in learning content: "[...] for example, when I imagine that I somehow deal with the topic of factory farming in class, and then just say that the chickens are totally cramped in cages, but I only have a picture, which perhaps doesn't concern anyone that much. But if I can really stand there in this farm with the help of social VR and look around and see how cramped these cages are, then it might be more enduring and impressive for me, and it keeps me more engaged, and then it might have more impact on my future actions and consumer behavior" (Group 3 B). Additionally, one student commented on the use of social VR from a more differentiated perspective: "I can perhaps imagine it quite well in the didactics, less in other disciplines, [...]but I think it's a good opportunity to try it out if you're designing a lesson with a group [...]" (Group 3_C).

Discussion and Implications

Concerning the further development of the social VR platform and the pedagogical concept for teacher education, the results and the participatory observation of the study are leading to the following conclusions and implications for the research questions:

RQ 1) How do preservice teachers perceive teaching and learning activities in fully immersive VR?

Results show that several participants pointed out that based on the virtual surroundings they focused more on the content shared. However, most of them agreed on the efforts that had to be made to read the presentation on the virtual whiteboard. As an improvement, few students would like to have the possibility of taking notes. Throughout the design task in the group activity, communication cues like facial expressions and gestures were missing leading to more time spent with managing communication flow. Observation showed that the speaker turns posed difficulty for two groups to anticipate. However, both groups started to compensate for the missing social cues. One group started to use hand signs. The other group concentrated on the avatar's upper body as a signal when the speaker's turn took place. Students commented on this and agreed that they felt losing valuable time to work together productively. Some recognized, however, that they had the urge to be more attentive and actively participating in keeping the workflow going. Shy students even felt they could participate in social VR more freely because, in their avatar representation appearance, they felt more anonymous and safer to speak.

Close to the end of the session, the majority agreed that wearing the VR headset would be exhausting, in particular when focusing on the digital whiteboard panel.

The exploratory observation of participants throughout the teaching and learning activities shows that preservice teachers were curious and open to the VR experience. Most of the preservice teachers got used to VR's hard-and software and did not need any technical support. Furthermore, the observation showed that the group task and the results' presentation were carried out from all groups without problems. However, taken from the results, some remarked that group size might be a limiting factor regarding fulfillment as communication was challenging.

RQ 2) How should teaching and learning processes using social VR in Teacher Education be designed?

Concerning the pedagogical concept, the observation showed that combining the seminar concept with the flipped classroom approach has proven beneficial for the action-oriented and situational teaching and learning processes in VR. As integrating social VR in teaching and learning processes takes more time, it was possible to use the seminar units in VR intensively to work on the subject matter without mentally overloading the participants. As results show, the lesson's extent of content was described as manageable; otherwise few participants stated they would have had to take notes. Overall, observations showed that all groups could perform the tasks. Technical limitations, such as the resolution of virtual media presentation, headset quality, and mobility as well as missing communication cues, made the seminar session

in VR more challenging, however, did they not prevent the teaching and learning processes overall. Combining seminar sessions concerning 1) location (online/face-to-face) and 2) teacher-or learner-centered, on a selective basis, teacher educators and preservice teachers could profit from the positive effects on motivation to use VR in class themselves, group dynamics, and participation in VR, without being dependent on technical limitations. Results showed that most preservice teachers thought the VR experience to be exciting. However, this is possibly due to novelty effects. Also, some stated that they felt a sense of community and group feeling including the lecturer. Concerning the social VR environment, distraction-free surroundings seemed to help that students' kept focus. Keeping the balance between functionality and usability, the room design itself needs no change. However, it might be interesting to observe in future studies what effect different sceneries would have on preservice teachers. As missing social cues lead to challenges in communication and interaction during a group activity, supporting social cues should be implemented into social VR's functionalities.

On the one hand, restricted communication cues in social VR lead to positive effects. Interview results showed that several participants focused more on the spoken word and thus felt more involved in activities. Also, observation and results show that some of them tend to pay more attention to addressing their peers, and attentively followed the communication.

On the other hand, social cues are vital for communicating with each other and interpreting non-spoken context. Without the extra information included in communication clues, it takes more time to keep up the communication flow during group activities and might influence the learning process. According to Roth et al. (2016; 2018) the absence of social or behavioral cues can have impacts on social interaction in VR, such as decreased communication efficiency or decreased affective understanding. Leading to the assumption that learning processes might profit from the implementation of social and behavioral cues, further studies in this regard are needed.

The results summarized above are subject to certain limitations. Regarding the samples of this exploratory study, it is essential to note that a convenience sample was used. Hence, against the background of the qualitative research approach and sampling method, the interviewees are not representative of their respective groups. Thus, the results may not apply to other preservice teachers in the same way.

To substantiate the findings, more investigation is needed to observe how teaching and learning processes in social VR can be designed effectively. At this moment, the close interdisciplinary work between pedagogical and technical disciplines plays a vital role, as necessary pedagogical processes need to be enabled by the technical environment of social VR. It will be insightful to extend the observation to more seminar sessions with changing learning environments as well as changing teaching and learning scenarios. Moreover, a further step will be that interventions take place locally distributed. Regarding the shift to distance teaching and learning formats in higher education during COVID 19 pandemic, in connection to the most commonly used videomediated platforms, social VR represents a possible extension tool to integrate actionoriented and situated learning in new pedagogical concepts for distance learning.

The findings from this exploratory study are currently incorporated into the further development of pedagogical concepts using social VR in teacher education. The provision of the pedagogical approaches and the developed materials as well as the

social virtual environment as open source will contribute to the dissemination of social VR scenarios in different educational contexts. Thus, the presented social VR teaching and learning activities offer a wide range of transfer and possible uses for students and lecturers of different disciplines in higher education in the future, also promoting international mobility and inclusion.

The study is part of the research project ViLeArn funded by the Federal Ministry of Education in Germany in the funding line "Research on digital higher education" (reference number 16DHB2111).

References

- Blömeke, S. (2017). Erwerb medienpädagogischer Kompetenz in der Lehrerausbildung. Modell der Zielqualifikation, Lernvoraussetzungen der Studierenden und Folgerungen für Struktur und Inhalte des medienpädagogischen Lehramtsstudiums. Advance online publication. https://doi.org/10.21240/mpaed/retro/2017.07.13.X (231-244 Seiten / MedienPädagogik: Zeitschrift für Theorie und Praxis der Medienbildung, Jahrbuch Medienpädagogik 3)
- Borthwick, A. C., & Hansen, R. (2017). Digital Literacy in Teacher Education: Are Teacher Educators Competent? *Journal of Digital Learning in Teacher Education*, 33(2), 46–48. https://doi.org/10.1080/21532974.2017.1291249
- Burdea, G., & Coiffet, P. (2003). Virtual reality technology (2nd ed.) [Wiley-IEEE Press].
- Dawley, L., & Dede, C. (2014). Situated Learning in Virtual Worlds and Immersive Simulations. In J. M. Spector, M. D. Merrill, J. Elen, & M. J. Bishop (Eds.), *Handbook* of Research on Educational Communications and Technology (Vol. 16, pp. 723– 734). Springer New York. https://doi.org/10.1007/978-1-4614-3185-5 58
- Dzardanova, E., Kasapakis, V., Gavalas, D., & Sylaiou, S. Exploring aspects of obedience in VR-mediated communication. 2019 Eleventh International Conference on Quality of Multimedia Experience (QoMEX)(IKEECONF-2019-369), 1–3. https://doi.org/10.1109/QoMEX.2019.8743196
- Eichelberger, A., & Leong, P. (2019). Using TPACK as a framework to study the influence of college faculty's beliefs on online teaching. *Educational Media International*, 56(2), 116–133. https://doi.org/10.1080/09523987.2019.1614246
- Eickelmann, B., Bos, W., Gerick, J., Goldhammer, F., Schaumburg, H., Schwippert, K., Senkbeil, M., & Vahrenhold, J. (Eds.). (2019). *ICILS 2018 #Deutschland: Computerund informationsbezogene Kompetenzen von Schülerinnen und Schülern im zweiten internationalen Vergleich und Kompetenzen im Bereich Computational Thinking* [1. Auflage]. Waxmann.
- Feng, Z., González, V. A., Amor, R., Lovreglio, R., & Cabrera-Guerrero, G. (2018). Immersive virtual reality serious games for evacuation training and research: A systematic literature review. *Computers & Education*, 127, 252–266.
- Ferdig, R.E., Baumgartner, E., Hartshorne, R., Kaplan-Rakowski, R. & Mouza, C. (2020). Teaching, Technology, and Teacher Education during the COVID-19
 Pandemic: Stories from the Field. Association for the Advancement of Computing in

Education (AACE). Retrieved August 3, 2020 from https://www.learntechlib.org/p/216903/.

- Foulger, T. S., Wetzel, K., & Buss, R. R. (2019). Moving Toward a Technology Infusion Approach: Considerations for Teacher Preparation Programs. *Journal of Digital Learning in Teacher Education*, 35(2), 79–91. https://doi.org/10.1080/21532974.2019.1568325
- Hellriegel, J., & Čubela, D. (2018). Das Potenzial von Virtual Reality für den schulischen Unterricht - Eine konstruktivistische Sicht. Advance online publication. https://doi.org/10.21240/mpaed/00/2018.12.11.X (MedienPädagogik: Zeitschrift für Theorie und Praxis der Medienbildung, Einzelbeiträge 2018 / MedienPädagogik: Zeitschrift für Theorie und Praxis der Medienbildung, Einzelbeiträge 2018).

Krumsvik, R. J. (2014). Teacher educators' digital competence. Scandinavian Journal of Educational Research, 58(3), 269–280. https://doi.org/10.1080/00313831.2012.726273

- Latoschik, M. E., Kern, F., Stauffert, J.-P., Bartl, A., Botsch, M. & Lugrin, J.-L. (2019). Not Alone Here?! Scalability and User Experience of Embodied Ambient Crowds in Distributed Social Virtual Reality. IEEE Transactions on Visualization and Computer Graphics, 25(5), 2134–2144. https://doi.org/10.1109/tvcg.2019.2899250
- Mayring, P. (2015). Qualitative Inhaltsanalyse. Grundlagen und Techniken. 12. überarbeitete Auflage. Weinheim: Beltz.
- Perry, T. S. (2016). Virtual reality goes social. *IEEE Spectrum*, 53(1), 56–57. https://doi.org/10.1109/MSPEC.2016.7367470
- Radianti, J., Majchrzak, T. A., Fromm, J., & Wohlgenannt, I. (2020). A systematic review of immersive virtual reality applications for higher education: Design elements, lessons learned, and research agenda. *Computers & Education*, 147, 103778. https://doi.org/10.1016/j.compedu.2019.103778
- Rädiker, S. & Kuckartz, U. (2019). Analyse qualitativer Daten. Wiesbaden: Springer VS.
- Ripka, G., Tiede, J., Grafe, S., & Latoschik, E. (2020). Teaching and Learning Processes in Immersive VR – Comparing Expectations of Preservice Teachers and Teacher Educators. In D. Schmidt-Crawford (Ed.), *Proceedings of Society for Information Technology & Teacher Education International Conference* (pp. 1836– 1844)
- Roth, D., Lugrin, J.-L., Galakhov, D., Hofmann, A., Bente, G., Latoschik, M. E. & Fuhrmann, A. (2016). Avatar realism and social interaction quality in virtual reality. 2016 IEEE Virtual Reality (VR), 277–278. <u>https://doi.org/10.1109/vr.2016.7504761</u>
- Roth, D., Latoschik, M. E., Bloch, C., & Bente, G. (2017). When Some Things are Missing: The Quality of Interpersonal Communication in Social Virtual Reality.
 Presentation at the 68th Annual Conference of the International Communication Association (ICA), May 24-28 2018, Prague, Czech Republic.
- Roth, D., Lugrin, J.-L., Latoschik, M. E. & Huber, S. (2017). Alpha IVBO Construction of a Scale to Measure the Illusion of Virtual Body Ownership. Proceedings of the 2017 CHI Conference Extended Abstracts on Human Factors in Computing Systems - CHI EA '17, 2875–2883. https://doi.org/10.1145/3027063.3053272

- Roth, D. & Latoschik, M. E. (2020). Construction of the Virtual Embodiment Questionnaire (VEQ). IEEE Transactions on Visualization and Computer Graphics, 1. https://doi.org/10.1109/tvcg.2020.3023603
- Scavarelli, A., Arya, A., & Teather, R. J. (2020). Virtual reality and augmented reality in social learning spaces: A literature review. *Virtual Reality*. Advance online publication. https://doi.org/10.1007/s10055-020-00444-8
- Schwan, S. & Buder, J. (2006). Virtuelle Realität und E-Learning. Tübingen. https://www.e-teaching.org/materialien/literatur/schwan-buder-2005
- Shiram, K. & Schwarz, R. (2017). All Are Welcome: Using VR Ethnography to Explore Harassment Behavior in Immersive Social Virtual Reality. IEEE Virtual Reality (VR), Los Angeles, CA, 2017, pp. 225-226, doi:10.1109/VR.2017.7892258.
- Slater, M., & Sanchez-Vives, M. V. (2016). Enhancing Our Lives with Immersive Virtual Reality. *Frontiers in Robotics and AI*, *3.* https://doi.org/10.3389/frobt.2016.00074
- Slater, M., & Steed, A. (2000). A Virtual Presence Counter. Presence: Teleoperators and Virtual Environments, 9(5), 413–434. https://doi.org/10.1162/105474600566925
- Stavroulia, Kalliopi-Evangelia; Christofi, Maria; Zarraonandia, Telmo; Michael-Grigoriou, Despina; Lanitis, Andreas (2019): Virtual Reality Environments (VREs) for Training and Learning. In: Paloma Díaz, Andri Ioannou, Kaushal Kumar Bhagat und J. Michael Spector (Hg.): Learning in a Digital World, Bd. 9. Singapore: Springer Singapore (Smart Computing and Intelligence), S. 195–211.
- Tondeur, J., Scherer, R., Baran, E., Siddiq, F., Valtonen, T., & Sointu, E. (2019). Teacher educators as gatekeepers: Preparing the next generation of teachers for technology integration in education. *British Journal of Educational Technology*, 50(3), 1189–1209. <u>https://doi.org/10.1111/bjet.12748</u>
- Tiede, J., & Grafe, S. (2019). The Integration of Media-Related Studies and Competencies into US and German Initial Teacher Education. A Cross-National Analysis of Contemporary Practices and Trends. (A. for the Advancement of Computing in Education (AACE), (Hrsg.)
- Tulodziecki, G., Herzig, B., & Blömeke, S. (2017). Gestaltung von Unterricht: Eine Einführung in die Didaktik (3., überarbeitete und erweiterte Auflage). UTB Erziehungswissenschaft, Schulpädagogik, Allgemeine Didaktik: Vol. 3311. Verlag Julius Klinkhardt. <u>http://www.utb-studi-e-book.de/9783838547947</u>
- Tulodziecki, G., Herzig, B., & Grafe, S. (2019). *Medienbildung in Schule und Unterricht. Grundlagen und Beispiele*. Bad Heilbrunn, Germany: UTB.
- US Department of Education. (2017). *Reimagining the role of technology in education:* 2017 national education technology plan update. Office of Educational Technology Washington, DC.
- Wang, M. (2020). Social VR: A New Form of Social Communication in the Future or a Beautiful Illusion? Journal of Physics: Conference Series, 1518(1), 12.032. hpps://doi.org/10.1088/1742-6596/1518/1/012032

4.3 Conclusion

The aim of the second study was to analyze how pre-service teachers perceive teaching and learning activities in fully immersive social VR and to derive consequences for further refinement of the pedagogical concept to promote the TPACK of pre-service teachers. As indicated at the beginning of Section 4.1, the in-practice implementation of social VR for teaching and learning processes in ITE requires the close adaptation and interplay of pedagogical and technical components. Regarding pedagogical design elements, the results of the second study showed that flipped classroom principles prove beneficial for action-oriented activities and enable effective use of limited seminar time (see Section 4.2). Thus, it was possible to integrate more breaks in the seminar units, which may have relieved the strenuous cognitive processing (Chandler & Sweller, 1991) that can occur with social VR stimuli (Parong & Mayer, 2021). Further, study participants described a sense of community; such an experience can result from feeling co-present in social VR (see Dalgarno & Lee, 2010; Hacker et al., 2020; Slater & Wilbur, 1997). The results showed that they enjoyed the social interactions and performance of the group task (see Section 4.2).

Regarding the social VR learning environment, the results of the second study indicated that social interactions without disruptions are highly important. Although the successful performance of the group task was closely linked to the characteristics of social VR, such technical restrictions as missing social cues, avatar representations, or content representation made social interactions more difficult. However, the missing cues also had beneficial effects, such as increased focus on spoken words by learners. Further, the results suggested that many breaks, close supervision, and monitoring in social VR can build a supportive frame for learners. In line with this, investing time in providing IT support can prevent major disruptions during seminars, particularly when done at the start of the session (Section 4.2).

In summary, considering the pedagogical and technical implications derived from the literature as a theoretical frame of reference (Euler, 2014), as well as the empirical results of the second study regarding students' perceptions, the following design principles of the pedagogical concept using social VR were implemented in the next design circle (Euler, 2014):

- Active learner engagement with the subject matter.
- Group tasks for small groups.
- Flipped classroom principles and action orientation.

- Clear and structured content representation to prevent students' cognitive overload.
- Use of the potential of the feeling of being together in social VR.
- Close support from a pedagogical and technical perspective.

5. Peer Group Supervision in Zoom and Social VR – Preparing Preservice Teachers for Planning and Designing Digital media integrated Classes

5.1 Aims of Study 3

In the first two studies carried out for the current research, contextual factors for a pedagogical concept to foster the TPACK of pre-service teachers in social VR were analyzed and explored by implementing a prototypical pedagogical concept to promote such TPACK using social VR on campus (Sections 3.2 and 4.2). In the third study, the pedagogical concept was refined by integrating further design principles and shifting from the lab setting to distributed teaching and learning processes in social VR (see Section 5.2).

Following the key objective of this work—namely, to promote pre-service teachers' TPACK as metaconceptual awareness (see Section 2.1)—the pedagogical concept aimed to foster metaconceptual learning processes, such as scaffolding, reflecting, and modeling instructional designs (see Section 5.2). In this vein, at the core of the investigation is the integration of the peer group supervision approach, as Tietze (2010) described.

Starting a new iterative cycle of refining, testing, and evaluating the implemented design principles (see Euler, 2014), Study 3 aimed to create a robust intervention before the final evaluation of the pedagogical concept for the promotion of pre-service teachers' TPACK in social VR (see Section 6.2). Further improvements in the design of the pedagogical concept were investigated in further pilot testing (see Euler, 2014). Furthermore, the implementation of the pedagogical concept in social VR was compared with the use of a video-based conference system, a tool commonly used in higher education (Pelletier et al., 2021). Thus, the study addresses the following research questions:

- How should a pedagogical concept for remote ITE be designed to promote the metacognitive learning processes of pre-service teachers?
- How do pre-service teachers perceive these learning processes in video-based communication and social VR?

The results of the study were used in the final evaluation of the pedagogical concept of fostering the TPACK of pre-service teachers in social VR.

5.2 Study 3

EdMedia + Innovate Learning 2021 - Online, United States, July 6-8, 2021

Peer group supervision in Zoom and social VR- Preparing preservice teachers for planning and designing digital media integrated classes

Gabriela Ripka Chair of School Pedagogy University of Wuerzburg Germany Gabriela.Ripka@uni-wuerzburg.de Silke Grafe Chair of School Pedagogy University of Wuerzburg Germany Silke.Grafe@ uni-wuerzburg.de Marc Erich Latoschik Chair of HCI University of Wuerzburg Germany Marc.Latoschik@uni-wuerzburg.de

Abstract: 21-century challenges demand a change towards collaborative and constructive seminar designs in initial teacher education regarding preservice teachers acquiring meta-conceptual awareness (TPACK) about how to implement emerging technologies in their future profession. Against this background the paper addresses the following research questions: 1) How should a pedagogical concept for remote initial teacher education be designed to promote metacognitive learning processes of preservice teachers? 2) How do preservice teachers perceive these learning processes in video-based communication and social VR? Regarding the pedagogical concept, peer group supervision and an action- and development-oriented approach using Zoom and social VR were identified as relevant for an instructional design that provides collaborative and constructive learning processes for students. In this exploratory study, 17 students participated in two iterative cycles of peer group supervision performing design tasks in groups. A content analysis of reflective video statements and qualitative group interviews was carried out using a qualitative research design. Results indicate the successful implementation of peer group supervision. Regarding media's implementation, Zoom's screen-sharing option and breakout session benefitted the consultation process as well as social VR's "realistic" experience of creating a "sense of community".

1. Introduction

The pandemic emphasized the need for new concepts in initial teacher education to prepare preservice teachers for 21st-century challenges. Shifting from f2f to remote teaching in education, teachers at all school types had to promote students' competencies with a set of available digital tools. Social distancing has restricted communication and collaboration to a specific limit with the consequence that teachers must put more effort in adapting, especially collaborative constructive learning activities to foster students' engagement in online learning scenarios. As most instructional designs in teacher education rarely offer constructive learning scenarios allowing technology integration (Foulger Teresa S. et al., 2017), there was a rise in teachers networking on social media platforms to support each other as a community of interest with methodological know-how for effective media integration into remote teaching and learning scenarios (Hacker et al., 2020). Most teachers did not feel prepared well enough for the complex task of planning, designing, and reflecting learning scenarios with emerging technologies, as this implies a sound knowledge of technology, pedagogy, content, and, more importantly, about how to transfer this knowledge into action. This in turn involves higher-order thinking skills such as reflective and problem-solving thinking processes that offer the potential to promote collaborative, constructive, and meaningful teaching and learning.

Hence, there is the need to design seminar concepts that provide student teachers with diverse learning opportunities, as early as possible in initial teacher education (ITE), offering incentives for the development of metacognitive learning processes as well as promoting, urgingly, media pedagogical competencies that prepare them for their complex tasks in their future profession (Blömeke, 2017; Foulger Teresa S. et al., 2017; Ripka, Tiede et al. 2020).

Against this background, this work will investigate the following two research questions:

1) How should a pedagogical concept for remote initial teacher education be designed to promote the metacognitive learning processes of preservice teachers?

2) How do preservice teachers perceive these learning processes in video-based communication and social VR?

2. Literature Review

TPACK as meta-conceptual knowledge

Between 2009 and 2020, more than 844 works have been published that contribute to the research on TPACK (Tseng et al., 2020). As an extension to Shulman's pedagogical content knowledge concept (Shulman, 1987), Mishra and Koehler (2006) introduced the conceptual framework TPACK. It addresses not only the mere technical aspects of using technology in educational contexts but also the multifaceted and complex pedagogical and content-related implications that go along with it. According to the authors, the framework comprises three core knowledge bases (technological, pedagogical, and content knowledge (TK, PK, CK)) and its intersectional components (technological pedagogical knowledge (TPK), technological content knowledge (TCK), pedagogical content knowledge (TCK), technological pedagogical content knowledge (TPACK)) that describe teachers' knowledge of using emerging technologies effectively in educational contexts *(ibid.)*. To explain how the knowledge domains interplay with each other and how TPACK is constituted there are two main perspectives in literature: the integrative and transformative view. In the integrative view, "high levels of TPCK will be constituted by high levels of TPK, TCK, PCK, TK, PK, and CK" whereas in the transformative view "TPCK cannot simply be accounted for by summing all other TPACK components, but rather it is a distinct form of knowledge which transforms beyond the components at its base" (Schmid et al., 2020) This work, however, deals with TPACK as a meta-conceptual knowledge as outlined in the following.

Regarding learners' development of TPACK, Mishra and Koehler (2006) point out that "learning through design embodies a process that is present in the construction of artifacts". In this "learning-by-doing" process, learners are supposed to engage actively "in practices of inquiry, research, and design in collaborative groups" (*ibid*.). Taking the design of lesson plans into focus, Zohar and Schwartzer (2005) indicate that design tasks are complex tasks that "require higher-order thinking in TPCK". Preservice teachers are supposed to make multiple decisions when integrating technology in class, considering contextual factors involving critical thinking and problem-solving. They need to (a) plan and design appropriate learning activities for teaching and learning scenarios with technology (b) choose digital media and content to use in teaching/learning and why; (c) embedding it in the pedagogical method to support that choice (d), deciding when and how to use it (Kramarski & Michalsky, 2010). As an elaboration of the transformative view on TPACK, based on the assumption that the TPACK framework also comprises the metacognitive learning processes involving higher-order thinking, Krauskopf et al. (2012) see TPACK as a meta-conceptual awareness that considers metacognitive aspects integrated into TPACK. This coherent theory is based on the notions that constructing mental models, that comprise a variety of aspects needed to design a lesson plan serve as "mediating variables between a teacher's abstract knowledge and planning the integration of the respective tool into their teaching", and thus lead to TPACK as a "higher mental model" (Krauskopf et al., 2012).

Against this background of TPACK being considered as a meta-conceptual awareness (Krauskopf et al., 2012; Krauskopf et al., 2012; Krauskopf et al., 2018) this work follows two lines of thought that set the baseline for the design of a pedagogical concept in remote initial teacher education. First, teachers need to construct complex mental models of integrating technology effectively in class for the development of media-pedagogical competencies *(ibid)*. Second, to develop these complex mental models, higher-order learning and collaborative, constructive learning processes are required, leading to TPACK as meta-conceptual awareness of how to implement emerging technologies in class. Therefore, in the following, peer group supervision as a potential concept to promote collaborative,

constructive learning processes will be outlined.

Peer group supervision

In teacher education, peer group supervision refers mainly to peer coaching approaches for in-service teacher's professional development. Since the early 1980s, more and more peer learning approaches have been implemented in preservice teacher education. Only a few studies however consider peer group supervision as a pedagogical approach in initial teacher education (Tietze, 2021). In this paper, the term peer group supervision follows the German concept of *kollegialer Fallberatung* of Tietze (2010) and related concepts (Richard & Rodway, 1992). It describes the process of people with the same profession share, reflect, and discuss problems or questions related to their profession. Although one could assume based on the word "supervision" that there is a hierarchical order, the participants are neither subordinate nor superior to each other regarding power structures. The peer group

-603-

supervision's main goal is that the participants conclude future actions for their profession through peer feedback and self-reflection (Tietze, 2010).

To distinguish peer group supervision from other similar concepts, Tietze names four critical features of peer group supervision:

(1) the concept must take place in a *group*. The author suggests building groups of five to ten participants depending on the existing conditions.

(2) *questions and cases* addressed are related to a shared profession. Cases should refer to an experienced professional role conflict, dilemma, or problematic interaction. Beyond this, they should also be of personal significance. The participant contributing to the case should be personally involved in it and have a personal interest in new perspectives (Tietze, 2010; 2021).

(3) according to Richard and Rodway (1992) most *consultation processes* follow a basic form of a fourphase structure: (a) The peer group supervision process starts with a participant's request for help, (b) the person presenting the case is exposing more information about it, (c) the group reacts to the question and the case focused on, giving room for further inquiries on the case and deepening the understanding of the presented information, (d) a decision or reflection on further possible actions is taken. The phases help participants as guidelines for communication processes, and thus also support "systematic problem-solving, such as a clear separation of problem description and solution development" (Tietze, 2021). Throughout the peer group supervision process, multiple perspectives play a particularly important role. By working on problems, questions, and cases from different angles, they can be viewed and reflected upon more closely and perceived in their complexity (Hesse & Lütgert, 2020).

(4) all *participants' roles* must be reversible. During the phases, participants fulfill reversible roles. Either participant takes the role of an advice seeker, presenting his/her question/case/problem to others, or he/she can support other advice-seekers as an advisor. Also, he/she can moderate the case consultation process in the peer group.

According to current research peer learning approaches based on constructive learning theories offer the potential to promote competencies needed for the teaching profession (Krauskopf et al., 2012; Krauskopf et al., 2018). Potential benefits of peer learning linked to observational learning based on socio-cognitive learning theory (Bandura, 1979; Tietze, 2010, 2021) are the development of confidence, self-esteem, collaborative skills, critical inquiry, and reflection. The authors also add to the benefits the communication and articulation of knowledge, understanding and skills, managing learning, and how to assess oneself and others (Boud et al., 2001)

Against this background, it is assumed, that with the integration of peer group supervision in remote initial teacher education, higher-order and constructive thinking processes are promoted and are leading to the development of TPACK as meta-conceptual awareness.

Web-conferencing systems and Social VR's affordances for peer group supervision

Digital media communication tools are used to replace or complement face-to-face communication. As during COVID-19, a high number of people were forced to use web-conferencing tools such as Zoom and Microsoft Teams, Hacker et al. (2020) investigated the affordances and constraints of web-conference systems for its users. To only name some of the affordances, the study's results lead to the conclusion that the use of web-conference systems supported the social co-presence of people and thus created "a social technology that led to a new virtual togetherness." (*ibid.*) Garrison et al. (2013) identified in their COI framework social presence as one core element of a collaborative constructivist learning environment required to create and sustain a purposeful learning community in online learning environments. Other works, however, show that computer-supported learning environments have limitations regarding synchronousness, non-verbal cues, physical proximity, spatial cohesiveness (Abfalter et al., 2012), and processing (Ferran & Watts, 2008) that might influence the feeling of "virtual togetherness" and thus limit the positive effects they assumingly have on collaborative constructive learning processes.

A medium that also favors mediated social interactions is social VR. Fully immersive VR as a communication and collaboration medium is widely applied and studied in a wide range of areas (Billingsley et al., 2019; Slater & Sanchez-Vives, 2016) Based on its main aspects such as immersion, presence, place illusion, plausibility illusion, and coherence (Bailenson et al., 2008; Latoschik & Wienrich, 2021; Skarbez et al., 2020; Slater & Steed, 2000), social VR offers the possibility of experiencing communication, collaboration, and interactions in VR close to the "real world" sensations. As in previous works outlined (Latoschik et al., 2019; Ripka, Grafe, &

Latoschik, 2020; Ripka, Tiede et al., 2020) fully immersive social VR's characteristics enable the planning and design of collaborative and constructive virtual teaching and learning processes. Yet, as with any medium, the sheer integration of VR in teaching and learning does not guarantee an additional value or improved learning success. A growing number of research works confirm its affordances and suggest additional values when it is included in educational settings reasonably. This however requires from designers of immersive teaching and learning scenarios the proper identification of social VR's appropriate implementation in line with learning objectives in lessons designs without overwhelming its learners (Johnson-Glenberg, 2018).

As communication and collaboration are two vital aspects of performing peer group supervision successfully regarding social learning processes, this work investigates how students perceive Zoom's and social VR's usage performing the peer group supervision cycles and its implications for learning processes.

3. Research Methodology

Against this background, built upon results of previous studies (Ripka, Grafe, & Latoschik, 2020) this paper investigates the benefits and challenges of peer group supervision in social VR in ITE to promote mediapedagogical competencies, focusing on the following research questions:

1) How should a pedagogical concept for remote initial teacher education be designed to promote the metacognitive learning processes of preservice teachers?

2) How do preservice teachers perceive these learning processes in video-based communication and social VR?

Study Design

Pedagogical Concept

The pedagogical concept was designed based on action- and development-oriented didactics (Tulodziecki et al., 2017; Tulodziecki et al., 2017; Tulodziecki et al., 2017; Tulodziecki et al., 2019) using complex tasks and a structured learning process combined with flipped classroom principles. Its primary pedagogical objective was the students' constructive and iterative design development of a technology-integrated instructional design in teaching and learning scenarios. According to flipped classroom principles, students prepared the learning content in advance asynchronously to perform design tasks throughout the seminar sessions synchronously. Course units comprised a combination of synchronous and asynchronous teaching and learning scenarios supported by digital media platforms such as LMS, Zoom, Miro, Flipgrid, and social VR. The course concept followed a sequenced four-stage structure with its primary focus on stages two and three, in that the peer group supervision cycles took place.

Stage I set the ground for the implementation of peer group supervision. Course sessions one to three covered a basic introduction to the seminar and media education, media competencies, and media design. In preparation for the first peer group supervision in session four, preservice teachers had to perform asynchronously a design task. Central to this task was the development and critical reflection of a technology-integrated instructional design in teaching and learning scenarios

Stage II starts with the first cycle of peer group supervision. The process was structured according to Tietze's peer group supervision's features (2010):

- (1) The group size was limited to 3-4 participants. As one seminar session lasted 90 minutes, each
- participant should have the chance to present a case or a question.
- (2) Students had to prepare a complex design task.
- (3) The consultation process comprised three main phases \pm 10 minutes:
 - a. The advice seeker presents his/her case. The others listen and do not interfere.
 - b. The advice-givers ask questions to clarify the stated case and information
 - c. The advisers offer ideas, information, or concepts that might help the advice seeker. A discussion or joint reflection can take place.
- (4) Three roles were assigned: the advice seeker, the advice-giver, and the moderator who also visualized the consultation process results on the online collaboration board. The teacher educator is not present but has the task of being a facilitator that monitors processes and intervenes when necessary.

Following the first peer group supervision, to promote further reflective processes, students uploaded a reflective video statement on Flipgrid, a web-based application that offers a platform for classes to upload engaging media content such as personal video clips that motivate students to interact with each other. Students or teachers can feedback on uploaded content. In addition to individual feedback given by the teacher educator, all participants reflected together on the process of their peer consultation.

For the second peer group supervision cycle, based on the previous session's content on designing and planning technology integration into the classroom, preservice teachers got a second task to design an instructional design of their choice. To investigate how video-based communication and social VR's affordances favor or hinder peer group supervision, students were allowed to choose either form (Zoom or social VR) of communication and collaboration tool.

Stage III started with the second peer group supervision cycle in Zoom and social VR. Again, following the peer group supervision cycle, preservice teachers uploaded their guided reflective video statement that the teacher educator commented on. The following session served as a joint reflection and opportunity for feedback.

At the end of the semester, in stage IV, students presented their final version of their technology-integrated instructional designs based on sound reasoning and their pedagogical, technological, and content choices.

Social VR prototype

The social VR prototype, based on Unity 2019.4 and optimized for Microsoft Windows 10, offered a fully immersive seminar room. The Oculus Rift S, a Head Mounted Display, and a Laptop served as VR hardware. To avoid unnecessary distractions for the students, the virtual seminar room is kept clean and simple, offering room for collaboration and interaction.



Virtual instructions guide the user to set up the avatar and one's virtual representation. For avatar representation, a comic-alike abstract avatar is available. Outer appearances like the color of the upper body, gender, and name are customizable for the users. After entering the preferred avatar choice, representations' names appear over the avatar's head, visible for all participants. Before joining the group room, students can see their representation in a virtual mirror and train how to use the controller elements. For the facilitation of communication, a bright pulsating dot next to the speaker's name signals the speaker's turns. As the peer group supervision follows a fixed-timed structure, students can set a virtual stopwatch that runs for all participants visibly next to the presentation wall. On the wall, participants can see the peer group supervision procedure as a guideline.

Methodology and data collection

The convenience sample consisted of 17 preservice teachers (12 female and 5 male). The students had the opportunity to sign up voluntarily for the intervention. Data was collected using qualitative methods at three points of time:

- After peer group, supervision I and II, preservice teachers had to upload guided reflective video statements on the online platform Flipgrid (n=17)
- After peer group supervision II, qualitative half-structured group interviews were conducted. Groups were divided according to the medium they participated in, Zoom or social VR. (Zoom: n =5; social VR: n = 12)

Due to COVID-19 restrictions at the university, web-based social VR's participation required a stable internet connection and living no more than 10 km away from the university's location. That is why only a limited number of students could participate, as not all students met the requirements.

Regarding reflective video statements on Flipgrid, students watched an introductory video statement of the teacher educator who gave two main reflective prompts. The prompts follow Schön's (1987) understanding of "reflection on action" and were presented after each peer group supervision as follows:

- a) Describe in short how you perceived peer group supervision.
- b) Peer group supervision is supposed to offer the possibility of receiving new impulses and
- perspectives based on the exchange with your fellow students.
 - Which aspects of the peer group supervision did you perceive as goal-oriented or, more minor goal-oriented?
 - Which aspects of the video conferencing software ZOOM did you perceive as supportive or obstructive in the three phases (case presentation, question round, discussion/reflection round?

Besides students' self-reflection, group interviews were conducted to consider also group reflection with the goal that group dynamics lead to a multidimensional understanding of group processes in Zoom and social VR.

The interview questions for the qualitative group interviews were derived from Tietze's peer group supervision, aiming to understand the nature of performing the consultation process twice in remote teacher education:

- a) Compared to peer group supervision I in Zoom, how did you perceive the communication processes with your peers throughout the first phase, the presentation phase?
- b) Compared to peer group supervision I in Zoom, how did you perceive the communication processes with your peers throughout the second phase, throughout question phase?
- c) Compared to peer group supervision I in Zoom, how did you perceive the communication processes with your peers throughout the third phase, the discussion/reflection phase?
- d) Compared peer group supervision I, what technology's characteristics (Zoom and/ or social VR) did you, as a group, find supportive or obstructive throughout the consultation process?

The interviews were recorded, transcribed, and analyzed through qualitative content analysis (Mayring, 2015). They were coded using MAXQDA (Rädiker & Kuckartz, 2019). The following categories were determined deductively following the approach of Mayring (2015) and the four main features of peer group supervision according to Tietze (2010, 2013, 2018, 2021):

- (1) Group size
- (2) Design task
- (3) Peer group supervision cycle
- (4) Communication and collaboration with peers in reversible roles

4. Results

The results of the guided reflective video statements and qualitative group interviews conducted with preservice teachers will be presented systematically following the two cycles of peer group supervision (PGS 1 and 2), in stages two and three, and structured according to selected categories derived from Tietze's peer group supervision features (2010). The categories were adapted and extended according to research interest.

Categories	Students' perceptions	Examples
PGS I (Zoom)	Reflective video statements on Flipgrid	
(1) Group size	Students perceived group sizes of three and four persons as interactive and helpful.	"Um, I can only speak from experience: the smaller the group, the more sense it makes. The more, um, effective you are in the discussion, I think." (Student K_PGSI)
	 Limited timeframe of 90 minutes for each group did not allow more group participants. 	"Good group of three. Good preparation and different approaches were insightful." (Student K_PGSI_Video statements)

		"I find it quite good in small groups. ()I imagine the time limit to be a bit difficult, if one carries out this with colleagues at school in this way and then has only 10 minutes to look for possible solutions and so on
		because I think this can be very difficult, especially when selecting the right approaches []." (Student J_PGSI)
(2) Design Task I	 Design task was complex and not easy to approach which led to some insecurities performing the task. Task's complexity permitted individual approaches of how to perform the task. 	"First of all, I found it reassuring that my two group members didn't present a sample solution, we were struggling with how to approach the task." (Student H_PGSI) "There were three of us and we really had three completely different - that is, lesson designs and approaches - and that was kind of cool to see." (Student F_PGSI)
(3) Peer group supervision cycle I	 All groups had difficulties sticking to the given structure and PGS's phases were mixed up. Students asked questions at the presentation's exact point without waiting for the next phase to start. Students that loosely followed a given structure perceived the questioning round as the most valuable part of peer group supervision, recognizing the gaps in their approaches to a solution. Some students felt overwhelmed by peer group supervision. In general, independently of the cycle's structure, most students perceived and helpful. 	"Um, the only thing is that the question round had included the discussion round. If someone had questions about their own case, then it was actually immediately also - um - discussed in the discussion round []." (Student M_PGSI) "Well, I must admit that I didn't perceive the peer group supervision well at first, and to be honest, I was still a bit overwhelmed at the beginning and didn't know what to expect. And that's why I had a lot of questions and - but once we started, it got easier and easier. Or rather, you feel, um - a little more confident." (Studem N_PGSI) "[] the feedback from my three fellow students, in the group, was really very purposeful and very beneficial, too." (Student L PGSI)
(4) Communica tion and collaboratio n with peers in reversible roles (Zoom)	 Most students perceived communication as interactive, helpful, and supportive. While most students thought their peers 'feedback helped the design's development, some felt the need for more expert feedback. Regarding reversible roles, students recognized the roles' function as a facilitator for work processes. For a successful peer group supervision, students realized that they must be open to criticism and that the team has to work together. 	"So, I think, mmm, the yes, it's like always - the higher the expertise in some area, the better you can also um, help others or maybe express yourself." (Student J_PGS1) "Aspects that I found purposeful: above all, the joint agreement, and the distribution of roles. Just that you had this moderator and case 1, 2, 3, and 4. Um, took a lot of organizational work off at the beginning and, um, everyone knew roughly in which role he was and what he had to do." (Student M_PGS1) "Openness to criticism must be present. A

EdMedia + Innovate Learning 2021 - Online, United States, July 6-8, 202	21
---	----

	Regarding Zoom's affordances:	functioning team is crucial for this." (Student
	 The screen sharing option was numerously named as good support for collaborating throughout the process. When questions came up that students could not answer ad hoc, they shared their screens to share material or information to follow or help the group to find the information needed. Using the breakout rooms without any supervision of teachers gave students the feeling of speaking openly and without pressure. 	"Yes, regarding Zoom, what I actually thought was helpful about the Zoom application that we always had the working material visible, and we could easily switch from our presentation to the Internet using the screen sharing option. There, we had the syllabus displayed, which was actually super useful, because then we could always show directly what we referred to or what we had in mind." (Student P_PGSI) "I thought it was great in the way we did it: So completely unevaluated, just us students among ourselves. So that you could really exchange ideas and also - yes, try to help the others without being judged or feeling like you were being watched. I thought that was good - it was a very pleasant setting." (Student H_PGSI)
PGS II (Zoom	Reflective video statements on Flipgrid and	
(1) Group size	 Students favored groups of three to four for productive group work. The students of a group of only two participants stated that this had no negative consequences for the process, as they had more time to talk about their two design concepts, but they would have wished for more perspectives and feedbacks. 	There were only two of us. On the one hand, that had advantages because we were able to have a good conversation and exchanged ideasand helped each other. Um, this time, like last time, there were cool new ideas that helped me. Well, because we were only two people, it was just, yes Impulses or the perspective of two people." (Student F_PGS2_video statement)
(2) Design Task II	• Students gained more confidence in creating their instructional design based on the seminar's theoretical basis.	"I thought there was definitely clear learning progress. Both, um, with my concept on which I had continued to work, as well as with my two fellow students, with whom I was in the VR session. Um. Actually, the concept was more advanced and, of course, also more goal-oriented. Because we had all been through peer group supervision and reflection before, and there we still had had questions in our heads. I went into the peer group supervision and didn't really know what I was actually doing. That has definitely improved a lot. Of course, also through the session that we had again with you. Then in general I would like to say that I perceived the session as very pleasant." (Student H_PGS2_video statement)
(3) Peer group supervision cycle II	 a. In Zoom Most students perceived the second PGS II as more structured. 	"In the first one, I had little idea what I was doing. In the second, I felt quite confident,

-609-

		TT1 1 1 1.1	and the Later Lange to a Later Call
	•	They described the process as more coherent and goal-oriented as structure was followed more strictly. Groups seemed to be more focused. Students who took part via Zoom underlined the positive supervision's development regarding structure and workflow.	and uh. I also have to admit that my fellow students they have had a deeper understanding, uh, of the topic. Because this time we stuck to the concept. [] (Student N_PGS2_video statement) "I also think that since more and more theoretical knowledge was added over time, you also build up a different structure for yourself when you introduce things, so you just give more reasons, which is not the case the first time, [] but I honestly didn't understand it deeply: How to connect this and
			that? The first time it was just like I write something down and the second time it was already very well-founded and somehow also, yes for me already more structured from preparation on. (Student M_PGS2_interview)
	•	b. In social VR In contrast to participants in Zoom, students in social VR had no online platform to collaborate and had no chance to use written notes. Many preservice teachers that joined PGS II in social VR reported that the first phase, in contrast to PGS I in Zoom, was strictly performed as a presentation phase, and no one asked any questions. Reasons given for this were mainly the restricted communication cues in social VR. As students could not interpret speakers' turns because of missing gestures or facial expressions, they waited for the person presenting to end the talk before asking questions. Following this, boundaries of the subsequent two phases, question round, and discussion/reflection blurred and blended. Some students in each group reminded others of sticking to the given structure In some groups, it was difficult to maintain an orderly process as several participants felt the need to take one or more breaks due to feeling exhausted wearing the VR	"This also shifted the distribution of roles a bit - we didn't do it in such a way that everyone presented their topics first, but rather we went straight to the questions afterward - we more or less confronted the presentation with the questions. Otherwise, it would have been lost, and you had the feeling: When can I finally refer back to what she said? That's why after the first presentation we said: Ok, now the questions and discussion, and then the next person presents first. Before we all present, then do the big round of questions, and then the discussion." (Student I_PGS2_video statement) "I didn't even notice that I needed a break, but afterward I was completely exhausted for half an hour and couldn't do anything anymore. Although throughout the process I actually felt totally fine. Such a mandatory break I think would be really useful!" (Student G_PG2_interview)
		very late that they needed a break as they focused intensely on the group work.	
(4) Communica		a. In Zoom	"I think everything worked without any
uon and collaboratio	•	many students agreed on Zoom as a practical communication and collaboration	talk, with the chance to interrupt and engage
		ration communication and conductation	,

n with peers in reversible roles (Zoom and social VR)	 tool. One student emphasized its advantages compared to f2f seminars at university. They agreed on the usefulness of the screen sharing and breakout session functions. The positive effects of heterogeneous groups were that diverse perspectives were added to question and discussion rounds, pointing at unclear content. However, one student felt it hard to put herself in the others' perspective and the age of their future students, and thus, it was challenging to give helpful advice. 	in speech, or, yes I think one understands more when everyone can freely speak and see each other." (Student N_PGS2_interview) "In general, in Zoom, I think the breakout rooms are super helpful because when I imagine how this works at Uni, this always takes time until one set up the workspace, this, however, is super relaxed, two klicks and you are in the group, and you can start right away to collaborate." (Student_D_PGS2_ interview) "And we also all had different, uh uh types of schools. I actually thought it was interesting that we were able to see the designs of the other types of schools and also what challenges this meant for the others. I thought it was good that we didn't just saw one school-type design." (Student B_PGS2_video statement) "Although I would say that () I find it hard enough sometimes to put myself in my pupils' perspective, i.e. the ones I will be teaching in the defined framework of the 1st-4th grade, and it is quite good to exchange ideas with people who are in the same grade and not to exchange ideas with someone else of other grades." (Student F_PGS2_interview)
	 b. In Social VR Participants perceived PGS II in social VR 	"Mhm and also really one has the feeling that one is really sitting next to each other and I personally thought that was super, super cool. Um, also the consultation is in my opinion, compared to the last time, much, yes almost more intensive, somehow." (Student L_PGS2_video statement)
	 as intensive and close to reality. Regarding communication and collaboration, many students had the impression that the interaction with others to be more "real" because of their avatar's representation. Also, they felt a sense of community. One said he had a "first person effect" and the mathematication for the term for the term in "form. 	"You somehow feel a bit closer, because you have a virtual person standing in front of you. Exactly, um, feels somehow a bit more real (laughter) than via ZOOM or similar, even though (in Zoom) you can also actually see each other." (Student E_PG2_video statement)
	 Interestingly, he had the feeling that in social VR his stuttering was minor, and he gained more stability because of the given feedback. The reduced non-verbal cues in social VR led to positive and negative effects on the participants' communication. The preservice teachers perceived that everyone 	"I think about Zoom you always have the problem that you see yourself and because it's first person in social VR, um, you don't have this problem and you can just talk much more freely with the other people. And um, also that you can manage it much better - no to stutter and everything. Um, in any case, VR has helped a lot as far as speaking freely is

EdMedia + Innovate	Learning 2021 -	Online,	United States,	July 6-8,	2021
--------------------	-----------------	---------	----------------	-----------	------

in social VR complied strictly with	concerned." (Student C_PG2_video statement)
conversation rules because of missing	"In the presentation place it was new
this most of students focused more on the	in the presentation phase, it was very
snoken word and the presented content	which is not a had thing At least I didn't dara
However, because of this participants did	to ask questions back because you don't sae
not dore to ask questions at the	any assturas or facial appractions and you
corresponding parts. Without the	don't notice whether the person is fully
opportunity of taking notes in social VR	involved in his or her presentation and
consequently, they forgot the questions	shouldn't be disturbed And so one
before having asked them. Thus, the	concentrated very strongly on listening and
question and discussion round were shorter	always thought. I'll keen the question in mind
than in PGS L and some felt they missed	for now But I guess a few questions were lost
meaningful opportunities of reflecting on	that we actually would have had for the
their design	second nart " (Student I PGS2 interview)
• In one group observed seated positions of	
participants were incorrectly calibrated and	"Well, it's uncomfortable for me to talk to you
thus one student was a lot smaller virtually	because vou're so above me." (Student
represented than the other two. The three	K_PGS2_interview)
stated that this was irritating for	
conversation. The lower-positioned student	"That made me a little sad, because I didn't
had the feeling that everyone was looking	know what they looked like, but they knew
down on her while speaking. She felt	what they looked like. But otherwise - I also
uncomfortable and tried to stand up on her	paid a lot of attention to the nodding of the
chair to become taller.	head or the shaking, or the hands, actually
Some students remarked that they did not	() I think that I was perhaps a bit more
know their social VR peers and would like	confident in my presentation because I didn't
to have had a picture of the real person in	look around all the time to see how people are
mind. One added, though, that her peers'	looking or reacting. And I think I wouldn't
anonymity caused her feeling more secure	necessarily have asked a lot of questions
in presenting her case and her questions, as	vectouse 1 inougni io myself al ine momeni,
she had not the feeling to say something	mayoe inui's a siupia question - bui I alan't
stupid.	see now people were looking at the moment
	anyway, so I was more likely to ask them and
	statement)

5. Discussion and Implications

The results summarized above are subject to certain limitations. Regarding the samples of this exploratory study, it is essential to note that a convenience sample was used. Hence, against the background of the qualitative research approach and sampling method, the interviewees are not representative of their respective groups. Thus, the results may not apply to other preservice teachers in the same way. Furthermore, in this first exploratory study, the focus was on the perceived learning processes but not on the effects of the instructional design on the advancement of TPACK as meta-conceptual knowledge. This focus was chosen to better understand the effective implementation of video-based communication and a fully immersive learning environment for instructional design.

Concerning the further development of a pedagogical concept for initial teacher education, the results and the participatory observation of the study are leading to the following conclusions and pedagogical implications for an instructional design that promotes metacognitive learning processes in remote teaching and learning, taking into account preservice teachers' perceptions of the peer group supervision cycles in Zoom and social VR:

(1) Group size

To adapt the concept of peer group supervision to the seminar's conditions, the seminar duration was set to 90 minutes to give every participant the same amount of time to present a case or question. Consequently, no more than three to four participants (each 30 minutes supervision cycle) could form a group.

→ Small size groups of up to a maximum of four people were perceived as productive and helpful. However, less than three participants would lead to less input and limited exchange of perspectives and opinions.

(2) Design task

At the beginning of introducing the design tasks and the new concept of peer group supervision, students faced multiple insecurities such as how to perform the complex task that allows multiple approaches and how to conduct peer group supervision. Moreover, students did not know each other and what to expect from their peers. However, this led to the initiation of questioning, reasoning, and reflective processes necessary to start creating their design. Students realized that there were several approaches of how to conduct the design task, observing their peers. For the students' support throughout this process, timely teacher feedback is essential. This way, insecurities related to their approaches can be reduced, and students become more confident to follow their design. As observed, the second design task was perceived as more manageable and clearer. Also, students seemed to be more focused and goal-oriented in working on their designs. After the second peer group supervision, reflection statements led to the assumption that students gained more confidence in presenting their design drafts and showed fewer insecurities. Based on the video statements and the interviews, peer and teacher feedback related to the designing process played a vital role.

→ The design task and PGS cycles need a thorough introduction and a test run.

→ Teacher's feedback should be placed after each PGS to clarify uncertainties.

(3) Peer group supervision cycle

As already mentioned, one can assume that preservice teachers gained more confidence throughout the design process from PGS I to PGS II and reported a more effective workflow in groups.

Participants who took part in Zoom in PGS I followed the PGS cycle's structure more consistently than in cycle one. Assumingly, students became more acquainted with the task, the cycle, their peers, and their design process. As a consequence, they felt more comfortable with interrupting their peers and risking the cycle's structure, but at the same time, this led to a lively exchange.

Preservice teachers that took part in the new social VR environment in PGS II had some difficulties in following the structure as intended. First, it took longer for them to start with the PGS cycle as they were distracted by social VR's surroundings and avatar representations. After starting the process, missing non-verbal communication cues led to uncertainties when to speak without interrupting peers. As a result, students listened more closely to each other's presentations and paid attention to gestures and body movements to interpret speakers' turns. On the one side, this might promote the cycle's consistency, as fewer interruptions will occur. On the other side, prompt questions and peer feedback might also be reduced and, thus, also its value for the learning process.

- → For future PGS in social VR, it is recommendable to have more social VR sessions, so that students get used to the social VR surroundings, preventing too much time spent with VR's distractions.
- (4) Communication and collaboration with peers in reversible roles

Concerning the implementation of digital collaboration media supported communication and collaboration throughout the PGS cycles, from students' perspective Zoom was beneficial for the group work in the PGS cycles. The screen sharing option and the breakout rooms resembled the peers' communication and collaboration style. Described as "realistic" and like "f2f" discussions, communication in Zoom was not perceived as disruptive. However, Zoom and the other collaboration platform (Miro) were the main communication tools throughout the semester and thus were frequently used. Through this repetitive media usage of the teacher educator as well as in group works, students internalized how to integrate the platforms in their PGS. In seminar sessions, when PGS did not take place student tutors and the teacher educator accompanied closely breakout sessions to support media usage and to give prompt feedback.

Communication and collaboration in social VR were not perceived as disruptive. However, communication processes were hindered or restricted due to missing non-verbal communication cues. Although the pulsating dot signaled the speaker's turn, it cost more concentration and cognitive load to interpret how group communication processes work. Within 90 minutes, the groups in social VR had to manage more cognitive load on top of managing
EdMedia + Innovate Learning 2021 - Online, United States, July 6-8, 2021

the PGS cycle's structure. This could be facilitated with more support measures, adapting it to social VR's conditions. Such measures could be:

- \rightarrow more breaks throughout the PGS cycles signaled in VR by a watch
- \rightarrow implementation of features that allow note-taking, f.e. a virtual chat
- → regular social VR sessions so that students get acquainted with VR hard-and software and thus communication and collaboration processes
- → more than two PGS cycles to facilitate structure to be followed
- \rightarrow strengthening the sense of community with group tasks at the beginning of the semester
- ➔ flexibility in teacher presence and absence so that students have someone to turn to with questions but still mainly collaborate with peers

Simultaneously, the notions of the consequences of social VR's anonymity based on avatar representation might help students to be more self-confident, not fearing to be judged by others, or reducing stuttering and this way offering pedagogical potentials like promoting self-regulation and self-efficacy. This effect could be used, when groups are acquainted with each other and with the process, groups could be mixed up with changing avatar representations to create the anonymity effect and to promote change of perspectives that favors the reflecting process.

From a teacher educator's perspective, the implementation of peer group supervision in social VR requires thorough planning and designing of seminar sessions. The difficulty lies within finding the right balance of knowledge transfer, technology integration, and giving enough time for students to construct knowledge on their own without overwhelming them. Most importantly is the close teacher's support for students throughout the process. As the constructive learning process and the technology are perceived as new and connected to a sense of insecurity, students tend to struggle with the detachment of teacher-centered seminar sessions.

For future works, it will be necessary to investigate how preservice teachers' TPACK development takes place and how video-based communication and social VR might influence it.

The findings from this exploratory study are currently incorporated into the further development of pedagogical concepts using social VR in teacher education. The provision of the pedagogical approaches and the developed materials, as well as the social virtual environment as open-source, will contribute to the dissemination of social VR scenarios in different educational contexts.

References

- Abfalter, D., Zaglia, M. E., & Mueller, J. (2012). Sense of virtual community: A follow up on its measurement. Computers in Human Behavior, 28(2), 400–404. https://doi.org/10.1016/j.chb.2011.10.010
- Bailenson, J. N., Yee, N., Blascovich, J., Beall, A. C., Lundblad, N., & Jin, M. (2008). The Use of Immersive Virtual Reality in the Learning Sciences: Digital Transformations of Teachers, Students, and Social Context. *Journal of the Learning Sciences*, 17(1), 102–141. https://doi.org/10.1080/10508400701793141
- Bandura, A. (1979). Self-referent mechanisms in social learning theory. American Psychologist, 34(5), 439–441. https://doi.org/10.1037/0003-066X.34.5.439.b
- Billingsley, G., Smith, S., Smith, S., & Meritt, J. (2019). A Systematic Literature Review of Using Immersive Virtual Reality Technology in Teacher Education. *Journal of Interactive Learning Research*, 30(1), 65–90. https://www.learntechlib.org/primary/p/176261/paper_176261.pdf
- Blömeke, S. (2017). Erwerb medienpädagogischer Kompetenz in der Lehrerausbildung. Modell der Zielqualifikation, Lernvoraussetzungen der Studierenden und Folgerungen für Struktur und Inhalte des medienpädagogischen Lehramtsstudiums. MedienPädagogik: Zeitschrift Für Theorie Und Praxis Der Medienbildung, 3(Jahrbuch Medienpädagogik), 231–244. https://doi.org/10.21240/mpaed/retro/2017.07.13.X
- Boud, D., Cohen, R., & Sampson, J. (2001). Peer Learning in higher Education: Learning from & with each other. Kogan Page.
- Ferran, C., & Watts, S. (2008). Videoconferencing in the Field: A Heuristic Processing Model. Management Science, 54(9), 1565–1578. https://doi.org/10.1287/mnsc.1080.0879
- Foulger Teresa S., Graziano, K., Schmidt-Crawford, D. A., & Slykhuis David (2017). Teacher Educator Technology Competencies. Journal of Technology and Teacher Education, 25(4), 413–448.

EdMedia + Innovate Learning 2021 - Online, United States, July 6-8, 2021

- Garrison, D. R., & Akyol, Z. (2013). Toward the development of a metacognition construct for communities of inquiry. *The Internet and Higher Education*, 17, 84–89. https://doi.org/10.1016/j.iheduc.2012.11.005
- Hacker, J., vom Brocke, J., Handali, J., Otto, M., & Schneider, J. (2020). Virtually in this together how webconferencing systems enabled a new virtual togetherness during the COVID-19 crisis. *European Journal of Information Systems*, 29(5), 563–584. https://doi.org/10.1080/0960085X.2020.1814680
- Hesse, F., & Lütgert, W. (Eds.). (2020). Auf die Lernbegleitung kommt es an! Konzepte und Befunde zu Praxisphasen in der Lehrerbildung. Klinkhardt, Julius.
- Johnson-Glenberg, M. C. (2018). Immersive VR and Education: Embodied Design Principles That Include Gesture and Hand Controls. Frontiers in Robotics and AI, 5, 81. https://doi.org/10.3389/frobt.2018.00081
- Kramarski, B., & Michalsky, T. (2010). Preparing preservice teachers for self-regulated learning in the context of technological pedagogical content knowledge. *Learning and Instruction*, 20(5), 434–447. https://doi.org/10.1016/j.learninstruc.2009.05.003
- Krauskopf, K., Foulger, T. S., & Williams, M. K. (2018). Prompting teachers' reflection of their professional knowledge: A proof-of-concept study of the Graphic Assessment of TPACK Instrument. Postprints der Universität Potsdam : Humanwissenschaftliche Reihe: Vol. 466. Universität Potsdam.
- Krauskopf, K., Zahn, C., & Hesse, F. W. (2012). Leveraging the affordances of Youtube: The role of pedagogical knowledge and mental models of technology functions for lesson planning with technology. *Computers & Education*, 58(4), 1194–1206. https://doi.org/10.1016/j.compedu.2011.12.010
- Latoschik, M. E., Kern, F., Stauffert, J.-P., Bartl, A., Botsch, M., & Lugrin, J.-L. (2019). Not Alone Here?! Scalability and User Experience of Embodied Ambient Crowds in Distributed Social Virtual Reality. *IEEE Transactions on Visualization and Computer Graphics*, 25(5), 2134–2144. https://doi.org/10.1109/TVCG.2019.2899250
- Latoschik, M. E., & Wienrich, C. (2021, April 10). Coherence and Plausibility, not Presence?! Pivotal Conditions for XR Experiences and Effects, a Novel Model. http://arxiv.org/pdf/2104.04846v2
- Mayring, P. (2015). Qualitative Content Analysis: Theoretical Background and Procedures. In A. Bikner-Ahsbahs, C. Knipping, & N. Presmeg (Eds.), Advances in Mathematics Education. Approaches to Qualitative Research in Mathematics Education (Vol. 23, pp. 365–380). Springer Netherlands. https://doi.org/10.1007/978-94-017-9181-6 13
- MISHRA, P., & Koehler, M. (2006). Technological Pedagogical Content Knowledge: A Framework for Teacher Knowledge. *Teachers College Record*, 108(6), 1017–1054. https://doi.org/10.1111/j.1467-9620.2006.00684.x
- Rädiker, S., & Kuckartz, U. (2019). Analyse qualitativer Daten mit MAXQDA. Springer.
- Richard, R., & Rodway, M. R. (1992). The peer consultation group: A problem-solving perspective. The Clinical Supervisor, 10(1), 83–100.
- Ripka, G., Grafe, S., & Latoschik, M. E. Preservice Teachers' encounter with Social VR–Exploring Virtual Teaching and Learning Processes in Initial Teacher Education. In *SITE Interactive Conference* (pp. 549–562). Association for the Advancement of Computing in Education (AACE).
- Ripka, G., Tiede, J., Grafe, S., & Latoschik, M. Teaching and Learning Processes in Immersive VR–Comparing Expectations of Preservice Teachers and Teacher Educators. In *Society for information technology & teacher education international conference* (pp. 1836–1844). Association for the Advancement of Computing in Education (AACE).
- Ripka, G., Tiede, J., Grafe, S., & Latoschik, M. Teaching and Learning Processes in Immersive VR–Comparing Expectations of Preservice Teachers and Teacher Educators. In *Society for information technology & teacher education international conference* (pp. 1836–1844). Association for the Advancement of Computing in Education (AACE).
- Schmid, M., Brianza, E., & Petko, D. (2020). Developing a short assessment instrument for Technological Pedagogical Content Knowledge (TPACK.xs) and comparing the factor structure of an integrative and a transformative model. *Computers & Education*, 157, 103967. https://doi.org/10.1016/j.compedu.2020.103967
- Schön, D. A. (1987). Educating the reflective practitioner: Toward a new design for teaching and learning in the professions. Jossey-Bass.
- Skarbez, R., Brooks, F., & Whitton, M. (2020). Immersion and Coherence: Research Agenda and Early Results. IEEE Transactions on Visualization and Computer Graphics, PP. https://doi.org/10.1109/TVCG.2020.2983701

EdMedia + Innovate Learning 2021 - Online, United States, July 6-8, 2021

- Slater, M., & Sanchez-Vives, M. V. (2016). Enhancing Our Lives with Immersive Virtual Reality. Frontiers in Robotics and AI, 3. https://doi.org/10.3389/frobt.2016.00074
- Slater, M., & Steed, A. (2000). A Virtual Presence Counter. Presence: Teleoperators and Virtual Environments, 9(5), 413–434. https://doi.org/10.1162/105474600566925
- Tietze, K.-O. (2010). Wirkprozesse und personenbezogene Wirkungen von kollegialer Beratung: Theoretische Entwürfe und empirische Forschung. VS Verlag für Sozialwissenschaften / GWV Fachverlage GmbH, Wiesbaden.
- Tietze, K.-O. (2013). Kollegiale Beratung: Problemlösungen gemeinsam entwickeln (6. Aufl., Orig.-Ausg). Miteinander reden: Vol. 61544. Rowohlt-Taschenbuch-Verl. http://www.socialnet.de/rezensionen/isbn.php? isbn=978-3-499-61544-3
- Tietze, K.-O. (2018). Kollegiale Beratung: Problemlösungen gemeinsam entwickeln (9. Auflage). Miteinander reden: Vol. 61544. Rowohlt Taschenbuch Verlag.
- Tietze, K.-O. (2021). Kollegiale Beratung für und von Schulberaterinnen. In (pp. 1-27).
- Tseng, J.-J., Chai, C. S., Tan, L., & Park, M. (2020). A critical review of research on technological pedagogical and content knowledge (TPACK) in language teaching. *Computer Assisted Language Learning*, 1–24. https://doi.org/10.1080/09588221.2020.1868531
- Tulodziecki, G., Herzig, B., & Blömeke, S. (2017). Gestaltung von Unterricht: Eine Einführung in die Didaktik (3., überarbeitete und erweiterte Auflage). UTB Erziehungswissenschaft, Schulpädagogik, Allgemeine Didaktik: Vol. 3311. Verlag Julius Klinkhardt. http://www.utb-studi-e-book.de/9783838547947
- Tulodziecki, G., Herzig, B., & Grafe, S. (2019). Medienbildung in Schule und Unterricht: Grundlagen und Beispiele. UTB: Vol. 3414. Klinkhardt.
- Zohar, A., & Schwartzer, N. (2005). Assessing Teachers' Pedagogical Knowledge in the Context of Teaching Higher-order Thinking. *International Journal of Science Education*, 27(13), 1595–1620. https://doi.org/10.1080/09500690500186592

5.3 Conclusion

The aim of the third study was to further refine the pedagogical concept to promote the metacognitive learning processes of pre-service teachers and analyze how such teachers perceive their learning processes in social VR compared with video-based communication. The results showed that, although the refined pedagogical concept could be implemented successfully in general using a video conference system and social VR, some design principles should be further refined for the final study (see Section 6.2). Notably, students needed more breaks and more time to become accustomed to the communication processes in social VR than in video communication. However, the following advantages of using social VR were identified in the third study: In comparison with peer group supervision in Zoom, students reported that they felt closer to each other in social VR, and mainly because of social distancing regulations, they enjoyed being together (see Section 5.2). To harness these positive effects for knowledge co-construction (Garrison & Akyol, 2013) in social VR, group work and collaboration processes can be supported by integrating group tasks that strengthen the feeling of community (see Section 5.2). As the results also showed, repeating the peer group supervision cycle allowed students to become accustomed to the underlying communication and collaboration processes, and their lesson designs profited from more runs (see Section 5.2). Moreover, the three-part structure of peer group supervision allowed the flexibility for more breaks to be taken if needed and contributed to guiding students through the reflection and scaffolding processes (see Section 5.2). Although peer group supervision is highly learner-centered, the teacher plays a crucial role in giving support and feedback. The results showed that teacher feedback should be an integral part of each peer group supervision cycle to ensure the provision of expert knowledge (see Section 5.2).

Regarding group sizes, students agreed on no more than three to four participants in one peer group supervision (see Section 5.2). Further, essential insights were gained about the nature of the design task. Its complexity is crucial in ensuring constructive exchange between students and thus a successful condition for the systematic construction of their lesson design (see Section 5.2).

Although still open to further development at this stage of the research and development cycle, the pedagogical concept seemed to be robust for the final evaluation. Therefore, as a next step,

the effect of the pedagogical concept on pre-service teachers' TPACK development was investigated and evaluated (see Section 6.2).

6. Mapping Pre-service Teachers' TPACK Development Using a Social Virtual Reality and Video Conferencing System6.1 Aims of Study 4

In the first two studies carried out for the current research, contextual factors for a pedagogical concept to foster the TPACK of pre-service teachers in social VR were analyzed and explored by implementing a prototypical pedagogical concept to promote such TPACK using social VR on campus (Sections 3.2 and 4.2). In the third study, the pedagogical concept was refined by integrating further design principles and shifting from the lab setting to distributed teaching and learning processes to compare the use of social VR and a video-conferencing system (see Section 5.2). The aim of Study 4 was to evaluate how the refined pedagogical concept can foster pre-service teachers' TPACK as metaconceptual awareness using social VR. Based on the insights gained from previous studies (Sections 3.2 to 5.2), the design of the pedagogical concept to promote TPACK as a metaconceptual awareness was refined and took the following design principles into account:

- Presentation of a complex and authentic design task
- Learner-centered learning activities
- Social learning processes
- Flipped classroom principles
- Action orientation
- Leaving space for support and breaks
- Scaffolding, reflection, and problem solving
- Content alignment to TPACK
- Teacher's feedback

As in Study 3, the implementation of the pedagogical concept in social VR was compared with use of a video-conferencing system. Thus, Study 4 addressed the following research question for the evaluation of the achievement of the pedagogical concept's objective:

• How did pre-service teachers' TPACK develop when they used a social VR learning environment prototype compared with a video-conferencing platform throughout the semester?

Accordingly, Study 4 aimed to enable consolidation of the pedagogical concept and its design principles in a summative evaluation with regard to Euler's (2014) design cycle.

6.2 Study 4

Innovate Learning Summit Online 2021 - , United States, November 9-11, 2021

Mapping pre-service teachers' TPACK development using a social virtual reality and a video-conferencing system

Gabriela Ripka Chair of School Pedagogy University of Wuerzburg Germany Gabriela.Ripka@uni-wuerzburg.de Si

Silke Grafe Chair of School Pedagogy University of Wuerzburg Germany Silke.Grafe@uni-wuerzburg.de Marc Latoschik Chair of HCI University of Wuerzburg Germany Marc.Latoschik@uni-wuerzburg.de

Abstract: Social VR's characteristics, by offering authentic learning environments that enable interaction remotely and synchronously and permit learning experiences that affect learners in a multi-sensory way, offer great potential for teaching and learning processes. However, concerning its use to promote pre-service teachers' TPACK in initial teacher education, there remains a research desideratum. In this context, this exploratory study addressed the following research question: How did pre-service teachers' TPACK develop using a social VR learning environment prototype in comparison to a video-conferencing platform throughout a semester? Following a design-based research approach, an action-oriented pedagogical concept for teaching and learning in social VR was designed and implemented for initial teacher education at a German university with a convenience sample of 14 participants. The lesson plans were collected and analyzed with the help of Epistemic Network Analysis (Shaffer, 2017) at three points of time during the semester and the GATI reflection process (Krauskopf et al., 2018). Further, 14 GATI diagrams gave insights into pre-service teachers' self-estimated TPACK. As the results indicate, pre-service students constructed more complex mental models of TPACK in social VR compared to the video-conferencing platform, indicating that more interrelations between knowledge domains could be constructed by planning and designing VR-integrated lesson plans.

Introduction

The sound integration of technology into classrooms involves multiple decisions from pre-service teachers. They need to 1) plan and design appropriate learning activities and scenarios for teaching with technology; 2) choose digital media and content accordingly; 3) embed them in the pedagogical concept that supports their choice; and 4) decide when and how to use it (Kramarski & Michalsky, 2010). In initial teacher education, it is thus vital to promote pre-service teachers' media pedagogical competencies, which include the competent use of the quickly developing social VR platforms, whose classroom use differs from that of other digital media in terms of resources and implementation. Therefore, to prepare students to meet these requirements, pedagogical concepts need to be developed, which include learning prerequisites, learning objectives, teaching and learning activities, technology, content, and social forms (Tulodziecki et al., 2021). There is, however, still a research desideratum on how to promote pre-service teachers' Technological Pedagogical Content Knowledge (TPACK) using social VR platforms in initial teacher education. In this context, this exploratory research study focused on promoting pre-service teachers' TPACK by aiming to answer the following research question:

RQ1: How did pre-service teachers' TPACK develop using a social VR learning environment prototype compared to using a video-conferencing system?

Literature Review

TPACK: Development and Measurement

A variety of models exist to describe the media-related competencies and knowledge of pre-service teachers (for an overview, see Tiede (2020)). Based on Shulman's (1986) research, the conceptual framework TPACK (Mishra & Koehler, 2006) is internationally well known for addressing the multifaceted implications of the effective integration of technology into learning scenarios. In addition to the three core knowledge domains—pedagogical, technological, and content knowledge (PK, TK, CK)—the framework comprises their intersectional components: technological pedagogical knowledge (TPK), technological content knowledge (TCK), pedagogical content knowledge (PCK), and technological pedagogical content knowledge (TPACK) (Koehler & Mishra, 2008).

Different definitions of the individual knowledge domains exist (13 definitions for TCK and 89 for the central construct TPACK) (Cox, 2008). However, the interrelation or the distinction of the domains and their overlaps or their relation to external context aspects remains unclear (Mishra, 2019). From 2009, a discussion of the two views concerning the quality and interplay of the seven knowledge domains evolved. According to the integrative view, "high levels of TPCK will be constituted by high levels of TPK, TCK, PCK, TK, PK, and CK." In contrast, in the transformative view, "TPCK cannot simply be accounted for by summing all other TPACK components, but rather it is a distinct form of knowledge which transforms beyond the components at its base" (Schmid et al., 2020). In other words, according to the integrative view, TPACK originates from the seven knowledge domains and is a part of them, whereas, according to the transformative view, TPACK forms a whole new unique body of knowledge.

Krauskopf et al. (2012, 2018) critically approached the transformative view with their coherent theory of TPACK as a meta-conceptual awareness. Following the transformative view, the theory assumes that the TPACK framework also comprises metacognitive learning processes involving higher-order thinking. Thus, according to the authors, the TPACK model is a multi-level structure, assuming a two-level transformation to achieve TPACK. They characterize the first level as the construction of mental models that occurs by the conversion of knowledge of the basic knowledge domains (TK, PK, CK) into knowledge of the overlapping subdomains (PCK, TPK, TCK). Further, on the second level, the transformed knowledge domain of TPCK represents the meta-conceptual awareness of the demands of designing a lesson with technology (Krauskopf et al., 2018). This however leaves unresolved how the knowledge subdomains support the construction of mental models and how they are transformed from level one to level two.

In addition to the different interpretations and the rise of numerous extended versions of the TPACK framework, Chai et al. (2010) highlighted the difficulty of measuring the TPACK of inexperienced pre-service teachers: doing so might lead to biased results based on their differing perceptions of the seven knowledge domains. Moreover, as further developments have also demonstrated, more variables, such as demographic data, teacher beliefs, self-perception, confidence, and self-efficacy, should be measured by TPACK's instruments (Chai et al., 2010). This led to the assumption that different types of TPACK profiles of pre-and in-service teachers should be considered when assessing TPACK (cf. Koh & Chai, 2014; Koh, Wooh, & Lim, 2013; Shih & Chuang, 2012).

Furthermore, research has focused on the TPACK construct's validity and reliability, its implementation in learning and teaching scenarios, or its impact on learning processes and teaching practices (cf. Abbitt, 2011).

According to Su and Foulger (2020), there has been "a shift over time, from quantitatively examining teachers' professed TPACK to qualitatively investigating teachers' enacted TPACK." In addition, to posit a more holistic view of TPACK, researchers moved away from using only one measurement method, turning instead to mixed methods approaches or combining different measuring instruments to balance efficiency, reliability, validity, or subjectivity (cf. Chai, Koh, & Tsai, 2011). Wang et al. (2019) identified mainly five methods to measure TPACK development in pre-service teacher education: 1) self-report measures (i.e., Likert scale, cf. Schmidt (2009), Koh, & Tsai (2011)); 2) open-ended questionnaires (i.e., written responses to questionnaire questions); 3) performance assessments (i.e., rubrics, performance tasks, created artifacts, lesson plan, content analysis, and reflections); 4) interviews (i.e., oral responses), and 5) observations (i.e., taking field notes, video-recording lessons).

The Graphical Assessment of TPACK instrument (GATI) was tested by Krauskopf et al. (2018), following their transformative perspective of TPACK, in their proof-of-concept study to visualize teachers' professional knowledge and prompt self-reflection through meta-conceptual awareness. Moreover, they considered the study a test of GATI's applicability as a scaffold for teachers' professional learning concerning technological integration.

In a three-part process, two in-service teachers participated in the GATI process. First, they completed a self-report questionnaire adapted from Schmidt et al. (2009) using a five-point Likert scale. After, with the help of a template with a set of six circles sized for each of the three knowledge areas of TPACK, they were asked to reflect on their current professional knowledge in each of the three domains. Further, they were asked to rearrange the circles in the way they perceived how the domains were interrelated, constructing a Venn diagram. They had to explain how they decided the size of each circle and the amount of overlap. After, they had to create and explain a second Venn diagram, representing the professional knowledge they wanted to achieve. Concluding that the GATI reflection process supports teachers in reflecting on their professional knowledge, the authors acknowledge GATI's potential as a measurement instrument (Krauskopf et al., 2018).

In contrast to a self-assessment survey, during the GATI process, the participants could state their understanding of what comprises TPACK and whether and how knowledge domains overlap. However, due to the participants' subjective view on TPACK, comparability of results is difficult. On the measurement of TPACK's development to explore the validity of the different domain sizes and overlaps and combine them with quantitative research methods, the authors noted the need for further research.

Another possible and valuable technique to map and trace TPACK development, because it can model the relationships between the seven knowledge domains and how pre-service teachers' knowledge domains transform over time, is Epistemic Network Analysis (ENA) (Shaffer, 2012, 2016, 2017). As the structure of

connections in the data is considered most important in the analysis, the following is assumed: (1) that it is possible to systematically identify a set of meaningful features in the data (codes); (2) that the data has local structure (conversations); and (3) that an essential feature of the data is how the codes are connected within conversations (Shaffer, 2017; Shaffer, Collier, & Ruis, 2016; Shaffer & Ruis, 2017). ENA models the connections between codes by quantifying the co-occurrence of codes within conversations or texts, producing a weighted network of co-occurrences along with associated visualizations for each unit of analysis in the data. ENA analyses all the networks simultaneously, producing a set of networks that can be compared visually and statistically.

Multimedia Learning Environments for Situated Learning and Teaching

During the COVID-19 pandemic, remote teaching and learning processes relied on digital media, allowing remote lessons to be taught as close to "real" face-to-face classroom scenarios as possible. For teacher educators in higher education, this meant pursuing learning objectives by effectively combining the positives of f2f teaching and learning with online learning scenarios.

Learning environments should embed the learning content into a situation-specific context (Dawley et al., 2014) and allow learners to establish connections to their prior experiences and their previous knowledge (Tulodziecki et al. 2021). Based on the supposition that the situation and context of the learning process determine the conditions in which knowledge is later applied (Jonassen, 1992), learning activities should comprise authentic application examples that leave enough room for self-determined learning processes (Reigeluth, 2015).

Moreover, based on the assumption that the situation and context of the learning process determine the conditions in which the knowledge is later applied (Jonassen, 1992), learning activities should comprise authentic application examples that leave enough room for the learner to have maximum self-control throughout the learning process.

Furthermore, since knowledge construction occurs through a cooperative exchange with others through co-construction (Reusser, 2001), teaching-learning activities should take social processes into account and provide space for them. However, when teaching and learning shifts from f2f to remote learning environments and the social presence of participants must be sustained by digital media, this becomes especially challenging. Social presence, as Garrison et al. (2013) emphasized, is paramount for collaborative and constructive learning environments.

During the pandemic, educational institutions had to rely on well-established web conferencing tools such as ZOOM, Big Blue Button, Microsoft Teams, or Cisco Webex. Features such as 2D-video exchange, screen sharing, and breakout rooms for group activities support teaching and learning processes. Web conferencing tools, Hacker et al. (2020) concluded, make participants feel that they are virtually together with others and thus create a social co-presence.

However, video-conferencing is also known for specific deficits in terms of social interactions regarding synchronousness, non-verbal cues, physical proximity, spatial cohesiveness (Abfalter et al., 2012), and processing (Ferran & Watts, 2008). For example, consumer-grade systems lack proper support for well-working eye contact; they reduce the shared spatial and hence referential space to little 2D images of the participants. Overall, they do not directly support the necessary social signals, such as eye contact, joint attention, or grouping (Roth et al., 2018).

In contrast, virtual, augmented, and mixed reality (VR, AR, MR; XR for short) can convey such signals, increasing user embodiment, co-presence, and the possibility of interaction (Latoschik et al., 2019). On this basis, studies have elaborated VR's suitability for situated and constructive learning approaches. Johnson-Glenberg (2020), for example, based on Woolfolk (2007), highlighted VR's versatility: it allows "complex, realistic and relevant learning environments," the "support of multiple perspectives and representations of content" as well as agency and ownership of one's learning (pp. 7-8). In particular, the opportunity to create "a "complex, realistic and relevant learning environment" that in real-life learning scenarios would be too expensive, unethical, dangerous, or not possible harbors the high potential of creating learning scenarios according to situated learning (cf. Hellriegel & Čubela, 2018). Even complex and abstract concepts and facts that are difficult to visualize in the conventional sense can be concretely illustrated using VR (Hellriegel & Čubela, 2018). This way, knowledge acquirement will take place in the context or environment to which the knowledge will be applied. However, studies that have assessed and evaluated VR's effects on situated learning are rare.

Having implemented a social VR learning environment prototype, Ripka et al. (2021) reported its supporting but also hindering effects on situated and cooperative learning processes in initial teacher education. For example, social VR's anonymity based on avatar representation favored students' self-confidence, allowing them to be not fearful of being judged by others. The participants' insecurities and stuttering were also reduced, which led to pedagogical opportunities for promoting self-regulation and self-efficacy Moreover, social VR's realistic communication and interactions enabled a sense of community, which supported the social learning processes (Ripka et al., 2021).

However, the authors highlighted the difficulties concerning social VR's integration into seminar concepts. For example, missing non-verbal cues, visual stimuli, and disruptive communications may result in cognitive overload, influencing the success of learning processes in social VR. Makransky et al. (2019), for example, assessed the influence of immersive technologies on learning outcomes and concluded that students were overloaded during learning processes when they learned in immersive VR. For instructional designers of VR-integrated lessons, this means finding the right balance in offering a variety of design principles to promote the respective competencies in a situated learning scenario and avoiding cognitive overloading students.

Research Methodology

This study aimed to explore how pre-service teachers develop their TPACK using an action-oriented pedagogical concept using a prototypical, fully immersive social VR learning environment compared to using a video-conferencing software. To this end, we used a design-based research methodology (Tulodziecki et al., 2013).

Study Design

Pedagogical Concept

The pedagogical concept is to promote pre-service teachers' TPACK development as a meta-cognitive awareness (Krauskopf et al., 2018). Regarding previous research (Ripka et al., 2021), the following design principles were considered goal-oriented:

- Following the principles of action-oriented teaching and learning by design (Tulodziecki et al., 2017) and a learning-by-design approach (Koehler & Mishra, 2005), the pre-service teachers get a complex task to evolve a lesson design on their own throughout the teaching and learning process. The design task is to develop a VR-integrated lesson design for teaching and learning scenarios according to primary, secondary, or special education curriculums in a constructive and iterative design process.
- The progression of the pedagogical concept follows a consecutive structure that comprises learning and teaching scenarios based on action- and development-oriented teaching following an ideal-typical structure of eight phases (Tulodziecki et al., 2017). These involved introducing and discussing the scenarios, agreeing on learning goals and their meaningfulness, planning activities, learning about important theoretical and empirical foundations for VR-integrated lesson designs, designing, presenting and discussing own lesson designs, and reflecting on the learning content and process.
- To promote the iterative development process and reflective practice, the pre-service teachers participate in peer-group supervisions following a pre-structured consulting cycle (Tietze 2013, 2018; Ripka et al., 2021). In groups of three, the students present their lesson design or a related question, a problem, or a case. They take three roles throughout the consultation cycle to coordinate and facilitate the communication: the advice seeker, the advice-giver, and the moderator. The teacher educator is not present, but is a facilitator, who monitors the processes and intervenes when necessary. The consultation process comprises three main phases of approximately 10 minutes:
 - a. The advice seekers present their case. The others listen and do not interfere.
 - b. The advice-givers pose questions to clarify the stated case and information.

c. The advisers offer ideas, information, or concepts that might help. Finally, a discussion or joint reflection takes place.

After each peer group supervision, the teacher educator offers support and inquiries to tackle unanswered questions or prevent misunderstandings.

To enhance reflection and metacognitive awareness (Krauskopf et al., 2018; Tietze 2013, 2018), the
pre-service teachers document and illustrate their lesson designs' development and reflections in a
portfolio. With the insights gained from each seminar session, the students revise their portfolios
continuously and integrate the consultation advice they get during peer group supervision into their
lesson designs. They hand in revised versions at five points throughout the semester.

Implementation

The implementation of the pedagogical concept took place in two groups of 12 students, each in an advanced media pedagogical seminar for pre-service teachers in initial teacher education at a German university during the summer semester from April until July 2021. Following and integrating the design principles explained above, the following structure was derived:



Innovate Learning Summit Online 2021 - , United States, November 9-11, 2021

Figure 1 The Derived Structure of the Pedagogical Concept

Social VR Learning Environment and Group Video-Conferencing System

Based on Unity 2019.4 and optimized for Microsoft Windows 10, the social VR prototype offered a fully immersive seminar room. The students and the teacher educator used the Oculus Rift S, a head-mounted display, and a laptop as VR hardware. The virtual seminar room was clean and simple to avoid unnecessary distractions for students and offered a space for collaboration and interaction (Ripka et al., 2021).



Figure 2 Social VR Learning Environment

The VR system allows users to personalize their own stylized avatars and select characteristics such as skin color, gender, and torso color. After a short tutorial and introduction to the software's main features, the participants can see their avatars' representation in a virtual mirror. When participants speak, a pulsating dot appears next to their names regarding communication facilitation. Further integrated features, such as a virtual stopwatch, support learning activities. All the participants can activate a control board with one click that offers the possibility of changing presentation mode, muting microphones, and changing position assignments for the avatars in the room. A voice memo function allows the recording of short notes during seminar sessions. In addition, students can take screenshots of the presented screen at the time of recording so that they can revise at home their notes set into the proper context. Compared to the students, teacher educators have additional options. They can switch between the different group rooms (A–D) and activate or deactivate 3D-based group tasks in and outside group rooms.

For its video-conferencing system, this study used ZOOM Version 5.7.6 (1320). Due to COVID-19 regulations, remote teaching at university took place via ZOOM, and the students are acquainted with its primary functions, including screen sharing, collaborating on the virtual whiteboard, and moving between breakout rooms.

Methodology, Sample, and Data Collection

The study was conducted during the summer semester of 2021 at a German university, from April until July 2021. A total of 14 pre-service teachers participated voluntarily in all rounds of data collection. The

convenience sample of the group using ViLeArn consisted of ($n_1=7$; 6 females, 1 male), and the group using ZOOM consisted of ($n_2=7$, 5 females, 2 males).

The seminars' content did not differ. The students of the VR group received a one-to-one introduction about the VR hard- and software in compliance with social distancing rules. The VR hardware was cleaned and disinfected before and after the seminar. However, due to COVID 19 restrictions, the ZOOM group could not test their VR hard-and software in a face-to-face workshop, but they received a 30-minute introduction to VR hardware via ZOOM.

Data were collected qualitatively in a pre-and post-test study design. The pre-service teachers' lesson designs were coded and evaluated with ENA (Shaffer, 2006, 2007, 2012) at three stages throughout the semester. For a detailed interpretation of the ENA results, the students were asked to describe and estimate their current state of TPACK with the help of the GATI reflection procedure according to an adapted version of Krauskopf et al. (2018) at the beginning and the end of the semester. The data collection is described in more detail in the following section.

a.) Portfolios and ENA

To make TPACK development visible as learning artifacts, the students revised and redesigned their lesson plans iteratively in a portfolio. Thus, at the beginning of the semester, the pre-service teachers got a situated complex task:

You are in teacher training at school, and your supervising teacher asks you to design a lesson with VR for a grade of your choice. Your final version of the lesson design should answer the following guiding questions and be able to justify your respective decision:

- For which target group is the concept intended?
- Which teaching-learning theoretical approaches does your concept refer to?
- To which curricular goals are the teaching and learning processes aligned?
- Which media competencies are promoted by your concept?
- How do you use the VR medium in the classroom, and what significance does it play in the implementation of the teaching and learning processes?
- Which features of the VR medium support teaching-learning activities?
- What opportunities and challenges did you face while planning and designing your teaching concept through VR? (own translation).

As described above, the students integrated the insights into the seminar content and the consultation advice from peer group supervision into their lesson designs. The versions at three points of time of each student were collected and used for ENA: (1) the beginning of the semester, (2) after the second peer group supervision, and (3) the final version.

To analyze TPACK development, we applied ENA (Shaffer, 2017; Shaffer, Collier, & Ruis, 2016; Shaffer & Ruis, 2017) to our data using the ENA1.7.0 (Marquart, Swiecki, et al., 2018) Web Tool (version 1.7.0) (Marquart et al., 2018).

The earlier test runs with ENA using the seven TPACK knowledge domains as codes showed that the demarcations between the individual knowledge domains are not sufficiently clear-cut for ENA, which is why a statement about the development and quality of the (meta-) cognitive processes and thus also about the respective references would not have been reliable and valid. Given that cognitive processes and developments are to be mapped with ENA, the TPACK codes set must also take into account not just knowledge states but also processes. Hence, we combined the seven knowledge dimensions (Wiesner & Schreiner, 2020) with process dimensions (Anderson et al., 2001). According to Anderson et al. (2001), the four knowledge dimensions differentiate between declarative (factual), conceptual, procedural, and metacognitive knowledge, and they build on each other in the sense of a continuum from the concrete to the abstract. Since we assumed that TPACK as meta-cognitive awareness can be achieved through two levels of transformations of the knowledge domains by building mental models, we included the following codes:

Table 1

Overview Code Categories for Coding Process of Portfolios

Knowledge Dimension	Level of transformation	Knowledge Domains	Example Code Category PK
Declarative Knowledge	1. Level transformation	TK1	
	(TK, PK, CK)	PK1	Pre-service teachers can summarize
		CK1	and select the basic teaching-learning
	2 Level transformation	TPK1	theories according to self-selected
	(TPK PCK)	PCK1	criteria. (PK1)
	(111,101)	TCK1	
Conceptual Knowledge	1 I aval transformation	TK2	
	(TK, PK, CK)	PK2	Pre-service teachers can recognize and
		CK2	categorize the basic teaching-learning
	2 Level transformation	TPK2	theories according to self-selected criteria. (PK2)
	(TPK PCK)	PCK2]
	(1111, 1 011)	TCK2	
	1 Laval transformation	TK3	
	TK PK CK)	PK3	Pre-service teachers can explain and
Procedural	$(\Pi \mathbf{K}, \Pi \mathbf{K}, \bigcirc \mathbf{K})$	CK3	evaluate the basic teaching-learning
Knowledge		TPK3	theories according to self-selected
	2. Level transformation	PCK 3	criteria. (PK3)
	(TPK, PCK)	TCK3	
	1 Level transformation	TK4	
Metacognitive Knowledge	(TK PK CK)	PK4	Pre-service teachers can identify and
	(111, 111, 011)	CK4	reflect on the basic teaching-learning
	2. Level transformation	TPK4	theories according to self-selected
	(TPK, PCK)	PCK4	criteria. (PK4)
	· · · /	TCK4	

We defined the units of analysis as all lines of data associated with the comparison plots between group VR and group ZOOM subset by the group type (for data comparison: VR 01, ZOOM 02), point of time (01, 02, 03), number of documents (f.e 2030303), and the associated peer group of the supervision cycle (f.e 203). In addition, we defined the references made in the portfolios as all lines of data associated with a single value of the semantical relation to all TPACK codes.

The ENA algorithm uses a moving window to construct a network model for each line in the data, showing how codes in the current line are connected to codes that occur within the current temporal context, defined as four lines (each line plus the three previous lines) within a given reference to TPACK codes. The resulting networks are aggregated for all lines for each unit of analysis in the model. In this model, we aggregated the networks using a binary summation in which the networks for a given line reflect the presence or absence of the co-occurrence of each pair of codes.

The ENA model normalized the networks for all units of analysis before they were subjected to a dimensional reduction, which accounts that different units of analysis may have different amounts of coded lines in the data. For the dimensional reduction, we used a singular value decomposition, which produces orthogonal dimensions that maximize the variance explained by each dimension (Shaffer et al., 2016). For a more detailed explanation of mathematics, see Sullivan et al. (2017).

The networks were visualized using network graphs, where nodes correspond to the codes and edges reflect the relative frequency of co-occurrence, or connection, between two codes. This results in two coordinated representations for each unit of analysis: (1) a plotted point, representing the location of that unit's network in the low-dimensional projected space, and (2) a weighted network graph. The positions of the network graph nodes are fixed, and those positions are determined by an optimization routine that minimizes the difference between the plotted points and their corresponding network centroids.

Data were analyzed and coded by the first and second coders. The first coder set the codes according to their appearance in the portfolios. The codes were set in order of sense units. The second coder then did the same, and after a final discussion between the first and second coders, the references of the individual codes to each other were set.

To go beyond the interrelations drawn between the knowledge domains, we considered the occurrence of codes in the respective group.

b.) TPACK GATI Diagram

To visualize pre-service teachers' estimation of their TPACK development, students were asked to reflect on their individual GATI diagram in their portfolio at the beginning and the end of the seminar. We implemented an adaptation of the GATI procedure of Krauskopf et al. (2018) and used the reflective prompts for the "current status" of TPACK and the GATI template to select the size of the knowledge bases (size 1–6). The task required that the students 1) describe the individual areas of knowledge, assuming that the smallest circles represent the knowledge of a non-expert and the largest that of an expert; and 2) put them in relation to each other and reflect on their Venn diagrams in writing.

Results

The results of the portfolios' ENA and the GATI diagrams are presented systematically following the assumed two-level transformation processes to achieve TPACK as a meta-conceptual awareness through constructing mental models (1st and 2nd level of transformation). To answer the research question of how preservice teachers' TPACK developed throughout the semester, this study focused on three main aspects:

- The development of each group's ENA network considering the connections' and nodes' weights at the three time points
- The isolated development of the references made between the three points of time
- The frequency of codes of each group without considering the references made between the knowledge domains

First-level Transformation of the Basic Knowledge Domains TK, PK, CK into Mental Models (TPK, PCK, TCK)

a.) ENA Networks

For the interpretation of the results, the ENA networks' nodes and connections with the minimal node's edge weight of 0.15 and scale edge weight of 0.5 were considered. This means that, even though the codes were set and counted for ENA, only those would be displayed that reoccur often enough to create a node and related connections in the network with the respective weight. The smaller the set edge weight, the more connections are involved, even if their reoccurrence is not meaningful enough for interpretation. That is why the edge weights were adapted to extract higher reoccurrences of the knowledge domain's connections. Therefore, the presentation of the results is based on the graphs of the ENA networks.

In Figure 1, the ENA network shows the positions of both groups in correlation to each other. The dots of each group (group VR: blue (1), ZOOM: red (0)) indicate the projected point of each student (1–7), and the squares represent the average of the confidence interval along both dimensions.



Figure 3: The ENA network of both groups; point of time 03 (blue = VR; red = ZOOM)

The ENA networks of both groups show an increase in connections and, thus, a thicker compound in color between the knowledge domains. As figures 3, 4abc and 5abc indicate, the connection between PK1 and PK2 is in both groups a reoccurring and strengthening connection across all points in time. Whereas group VR's network indicates stronger connections drawn between PK2 and TK2 at all time points, ZOOM's network only shows at the first and second point of time (Figure 4a, b) the inclusion of the knowledge domain TK3 connected to PK2. In general, the ZOOM group's network displays the repetitive connections between PK1 and PK2 and

PK3 knowledge domains that increases from 01 to 03. A further slight connection is implied to CK1 at point of time 02. Moreover, the network indicates the central starting point PK1 for connections made (PK1-PK2, PK1-PK3-PK1-PK4), creating a star-like ENA network representation. In comparison, at point of time 02, the VR group established additional links between TK1-CK1-PK3 and PK3-PK2, resulting in more interrelations between knowledge domains at point of time 02 and 03 (figures 4bc and 5bc).



Figure 4a: Group ZOOM, time point 01 Figure 4b: Group ZOOM, time point 02



Figure 5a: Group VR, time point 01 Figure 5b: Group VR, time point 02

Figure 5c: Group VR, time point 03

Figure 4c: Group ZOOM, time point 03

In a further step, to back up the interpretation of the ENA networks, including all codes of all portfolios, the codes that were added from the point of time 01 to 02 and from 02 to 03 were extracted and analyzed, so that only the newly created connections would be displayed:







● TK4 PK2

Figure 7c: Group VR, time point 03

Figure 7a: Group VR, time point 01

Figure 7b: Group VR, time point 02

In addition, the frequency of the reoccurrence of codes was counted and compared between the groups:

Codes	Point of Time	TK1	TK2	TK3	TK4	PK1	PK2	РКЗ	PK4		CK1	CK2	СКЗ	CK4
VR	1	14	17	6	2	27	22	11	6		9	7	1	0
	2	21	25	14	8	37	38	39	20	Ī	25	13	9	2
	3	34	48	34	14	66	58	73	27		34	18	9	0
		69	90	54	24	130	118	123	53		68	38	19	2
Zoom	1	8	4	4	2	25	16	8	1	Ī	7	1	2	0
	2	20	7	7	5	46	40	50	16	1	17	5	1	1
	3	29	17	17	16	64	57	77	30		12	7	5	2
		57	28	25	23	135	113	135	47		36	13	8	3

The summary of all the three sets of data indicate the following developments of codes and their interrelations:

In both groups, there was nearly no difference between the knowledge domains' development of PK1, PK2, PK3, and PK4. However, the ZOOM group made the most progress between time points 01 and 02, whereas the VR group made more progress in the PK areas after the second time point. The further divergent developments are as follows:

1.) Group ZOOM:

- The strongest connection was built between PK1-PK2. Studying the data of the ZOOM group, it was noticeable that the references were mainly formed in the PK areas. Additionally, all the references originated in PK1: PK1-PK2, PK1-CK1, PK1-PK3, and PK1-PK4.
 - The highest number of codes represented in the ZOOM group was limited to the PK areas PK1, PK2, PK3, and PK4.

2.) Group VR:

- The strongest connection was between PK2-TK2. However, the ENA networks indicate the development of further connections between PK1-CK1 at point of time 01. Furthermore, at the last observed time 03, students made additional connections between the knowledge domains PK3, resulting in an ENA network displaying the interrelations between TK1-PK1-CK1 and PK1-PK2-TK2 PK1-PK3, PK3-CK1, and CK1-PK1.
- The sum of codes within the TK knowledge domains suggests that TK2 and TK3 were applied more often than in the other group. Further, CK1 and CK2 appeared more frequently.

b.) GATI Diagrams

In their GATI diagrams, the students of both groups described gains in their respective knowledge areas. All of them reported an increase of technical knowledge in the form of at least one bigger circle size (cf. figures 8 and 9).



The groups differed, however, in the focus of their remarks. The VR group mainly described the growth and connections of TK and CK knowledge, while the ZOOM group mainly referred to pedagogical knowledge.



Second-Level Transformation of Mental Models TPK, PCK, TCK to TPACK

a.) ENA Networks

VR, time point 01 and 03

To represent the existing mental models, their development, and relations to each other in students' lessons plans, we repeated the same procedure for the knowledge domains TPK, PCK, and TCK. As TCK was not coded, the result representation was focused on TPK and TCK.



Figure 12a: Group ZOOM, time point 01 Figure 12b: Group ZOOM, time point 02 Figure 12c: Group ZOOM, time point 03





Figure 13a: Group VR, time point 01

Figure 13b: Group VR, time point 02

Figure 13c: Group VR, time point 03

Figure 11: GATI Diagram, ZOOM,

Student E, time point 01 and 03

Codes	Point	TPK1	TPK2	TPK3	TPK4	PCK1	PCK2	PCK3	PCK4
	of								
	Time								
VR	1	10	10	4	2	4	1	1	0
	2	17	13	14	6	9	5	4	2
	3	26	37	17	15	15	12	9	0
		53	60	35	23	28	18	14	2
ZOOM	1	8	7	11	1	6	0	1	0
	2	19	18	29	5	9	3	4	1
	3	18	24	28	6	8	6	9	0
		45	49	68	12	23	9	14	1

There was nearly no difference in the number of codes regarding the knowledge domains TPK and PCK. Nevertheless, the two groups differed in the distribution and development of the respective connections. The main differences can be seen between the knowledge domains TPK2, TPK3, and TPK4. While the VR group from the start elaborated on the connections between TPK1-TPK2-PCK1 and TP1-TPK3-PCK3, the ZOOM group focused more on TPK1 and TPK2 and TPK3 connections. In point of time 02, both groups' ENA

network showed an approximation of both groups to each other, and they seemed to be similar. However, the numbers reported a rise in TPK3 codes for the ZOOM group and PCK2 for the VR group.

b.) GATI Diagrams

Most of the students pictured their development with circles but did not comment or reflect on it explicitly. Only a few students stated how they estimated the difference in overlapping knowledge domains. However, it was noticeable that, instead of using the circle for PK, two participants of the ZOOM group chose PCK as one basic knowledge domain and described its development. Both stated that, since they study in the field of special education, and since PK and CK are always quite closely related to each other, PCK, in their opinion, is a unique body of knowledge.



Figure 14: GATI diagram, Student B, ZOOM, time point 01 and 03

Discussion and Implications

The results summarized above are subject to certain limitations. It is essential to note that a convenience sample was used in this exploratory study. Hence, considering the background of the qualitative research approach and sampling method, the participants are not representative of their respective groups. The results thus may not similarly apply to other pre-service teachers. In addition, the ENA coding process comes with some vulnerabilities, such as subjectivity.

This study aimed to develop a better understanding of how pre-service teachers' TPACK developed over a semester of social VR and ZOOM use. Further, it explored what connections were drawn between the individual knowledge domains that indicated the progress toward a meta-cognitive awareness of how to plan and design a digital media-integrated lesson.

Due to the nature of TPACK, we turned to ENA as a research method and the GATI reflection process, which allowed us to map the emerging connections between the knowledge domains of the respective groups. Particularly, the inclusion of the knowledge levels (Wiesner & Schreiner, 2020) permitted us to analyze the progress students made throughout the lesson design process. Furthermore, the abstraction of the different knowledge domains into factual, conceptual, procedural, and meta-cognitive knowledge offered us the chance to abstract pre-service teachers' learning process. Although this study does not make any claims to holistically display cognitive processes and their nature, from a pedagogical view, the results of mapping pre-service teachers' TPACK allow concluding the implementation of social VR in initial teacher education, the further improvement of the seminar concept and for further TPACK research with ENA.

Overall, the data collected on pre-service teachers demonstrate considerable homogeneity in some instances regarding their development of the basic pedagogical knowledge domains PK1, PK2, PK3, and PK4 and their first transformations. By studying the seminar process and content and the portfolio assessment criteria for passing the course closer, the pedagogical focus becomes apparent and explains the high frequencies of the PK (1–4) connections.

Regarding the quality of the basic knowledge domains, both groups progressed from summarizing and reproducing pedagogical knowledge as factual knowledge to making connections not only to procedural knowledge but also (a few) to meta-cognitive knowledge. However, this homogenous development was only observed regarding the overall consideration. A closer look at the points of time reveals that the VR group established more TK2 (conceptual knowledge) connections in the first period and progressed with PK and CK connections in the second period. Nevertheless, aligned to the seminar concept, after the second point of time it was expected that there would be a similar rise and development of knowledge domains as students of both groups had participated in peer-group supervision and received individual teacher educator's feedback before handing in the portfolio for 02.

The deviations in the development of both groups allow the assumption that the VR group, based on their application experience of VR hard-and software, included the TK2 (conceptual knowledge) components at an earlier stage and, thus, drew connections between PK1-TK2 for their lesson designs more frequently. They rarely used TK1; instead, they drew connections to the conceptual knowledge (TK2). On the contrary, the ZOOM group had no experience using VR in the learning contexts and could assume solely from the seminar's content how its application would influence the planning and designing of lessons. Another aspect that might

support this assumption is the development of the content knowledge's progress in the VR group. Although content knowledge was not promoted directly by the seminar concept, both groups made connections to CK. While the VR group developed successively more content knowledge on the conceptual level from the point of time 02 onwards, the ZOOM group participants' frequency of established CK codes decreased from the point of time 02 to 03. According to the situated learning approach, it can be assumed that through the application of VR hardware and software, the participants in the VR group, considering the conceptual implications of integrating VR into their respective school subjects, constructed the knowledge built on their experiences. To do so, they had to delve deeper into the content knowledge to conceptually decide based on their factual knowledge and even more on their experiences about whether their learning objectives could be achieved with VR. The ENA networks showed more complex mental models with more connections between the components than the ZOOM group. This is supported by the second transformation's results, which show more complex TPK4 and PCK3 references made in the VR group. However, the ZOOM group elaborated their portfolios based on their pedagogical knowledge and drew mainly links originating in the pedagogical knowledge domains, leading to more TPK3 references in the second transformation and fewer additional PCK connections. All in all, the results of both groups illustrate a shift toward activating more procedural and meta-cognitive knowledge domains and respective interrelations.

Following the assumption that pre-service teachers in social VR formed more complex mental models in planning and designing VR-integrated lesson plans based on their VR experiences, the importance of integrating VR in seminar concepts in initial teacher education becomes evident. However, whether this development applies across diverse groups should be confirmed by future studies with larger sample sizes. From a pedagogical and empirical point of view the results contribute to a design-based research approach in initial teacher education to derive theory-based design principles for the development of seminars using social VR to promote TPACK of pre-service teachers in initial teacher education.

The findings of this study are currently incorporated into the further development of pedagogical concepts using social VR in teacher education within the ViLeArn project's framework. Moreover, the provision of the pedagogical approaches and the developed materials, as well as the social virtual environment as open source, will contribute to the dissemination of social VR scenarios in different educational contexts.

References

- Abfalter, D., Zaglia, M. E., & Mueller, J. (2012). Sense of virtual community: A follow up on its measurement. Computers in Human Behavior, 28(2), 400–404. https://doi.org/10.1016/j.chb.2011.10.010
- Abbitt, J. T. (2011). An Investigation of the Relationship between Self-Efficacy Beliefs about Technology Integration and Technological Pedagogical Content Knowledge (TPACK) among Preservice Teachers. Journal of Digital Learning in Teacher Education, 27(4), 134–143. https://doi.org/10.1080/21532974.2011.10784670
- Anderson, L. W., Krathwohl, D. R., Airasian, P. W., Cruikshank, K. A., Mayer, R. E., Pintrich, P. R., ... Wittrock, M. C. (Hrsg.). (2001). A taxonomy for learning, teaching, and assessing: A revision of Bloom's taxonomy of educational objectives (Complete ed). New York: Longman.
- Arastoopour, G., Swiecki, Z., Chesler, N. C., & Shaffer, D. W. (2015). Epistemic Network Analysis as a tool for engineering design assessment. Presented at the American Society for Engineering Education, Seattle, WA
- Andrist, S., Collier, W., Gleicher, M., Mutlu, B., & Shaffer, D. (2015). Look together: Analyzing gaze coordination with epistemic network analysis. Frontiers in Psychology, 6(1016).
- Bandura, A. (1979). Self-referent mechanisms in social learning theory. American Psychologist, 34(5), 439–441. https://doi.org/10.1037/0003-066X.34.5.439.b
- Billingsley, G., Smith, S., Smith, S., & Meritt, J. (2019). A Systematic Literature Review of Using Immersive Virtual Reality Technology in Teacher Education. *Journal of Interactive Learning Research*, 30(1), 65–90. https://www.learntechlib.org/primary/p/176261/paper_176261.pdf
- Blömeke, S. (2017). Erwerb medienpädagogischer Kompetenz in der Lehrerausbildung. Modell der Zielqualifikation, Lernvoraussetzungen der Studierenden und Folgerungen für Struktur und Inhalte des medienpädagogischen Lehramtsstudiums. MedienPädagogik: Zeitschrift Für Theorie Und Praxis Der Medienbildung, 3(Jahrbuch Medienpädagogik), 231–244. https://doi.org/10.21240/mpaed/retro/2017.07.13.X
- Boud, D., Cohen, R., & Sampson, J. (2001). Peer Learning in higher Education: Learning from & with each other. Kogan Page.
- Chai, C. S., Koh, J. H. L., & Tsai, C.C. (2010). Facilitating preservice teachers' development of technological, pedagogical, and content knowledge (TPACK). Journal of Educational Technology & Society, 13(4), 63-73.
- Cox, S. (2008). A conceptual analysis of technological pedagogical content knowledge. Unpublished doctoral dissertation. Provo, UT: Brigham Young University
- Dawley, L., & Dede, C. (2014). Situated Learning in Virtual Worlds and Immersive Simulations. In J. M. Spector, M. D. Merrill, J. Elen, & M. Bishop, Handbook of Research on Educational Communications and Technology (S. 723-734). New York: Springer.
- DiSessa, A. A. (1988). Knowledge in pieces. In G. Forman & P. Pufall (Eds.), Constructivism in the computer age (pp. 47– 70). Hillsdale, NJ: Erlbaum.

Eagan, B., & Hamilton, E. (2018). Epistemic Network Analysis fo an International Digital Makerspace in Africa, Europe, and the US. Presented at the Annual Meeting of the American Education Research Association, New York, NY.

- Ferran, C., & Watts, S. (2008). Videoconferencing in the Field: A Heuristic Processing Model. Management Science, 54(9), 1565–1578. https://doi.org/10.1287/mnsc.1080.0879
- Foulger Teresa S., Graziano, K., Schmidt-Crawford, D. A., & Slykhuis David (2017). Teacher Educator Technology Competencies. Journal of Technology and Teacher Education, 25(4), 413–448.
- Garrison, D. R., & Akyol, Z. (2013). Toward the development of a metacognition construct for communities of inquiry. The Internet and Higher Education, 17, 84–89. https://doi.org/10.1016/j.iheduc.2012.11.005
- Hacker, J., vom Brocke, J., Handali, J., Otto, M., & Schneider, J. (2020). Virtually in this together how web-conferencing systems enabled a new virtual togetherness during the COVID-19 crisis. *European Journal of Information Systems*, 29(5), 563–584. https://doi.org/10.1080/0960085X.2020.1814680
- Hellriegel, J. & Čubela, D. (2018). Das Potenzial von Virtual Reality für den schulischen Unterricht Eine konstruktivistische Sicht. MedienPädagogik: Zeitschrift für Theorie und Praxis der Medienbildung, 58–80. https://doi.org/10.21240/mpaed/00/2018.12.11.x
- Hesse, F., & Lütgert, W. (Eds.). (2020). Auf die Lernbegleitung kommt es an! Konzepte und Befunde zu Praxisphasen in der Lehrerbildung. Klinkhardt, Julius.
- Johnson-Glenberg, M. C. (2018). Immersive VR and Education: Embodied Design Principles That Include Gesture and Hand Controls. Frontiers in Robotics and AI, 5, 81. https://doi.org/10.3389/frobt.2018.00081
- Jonassen, D. H. (1992). What are cognitive tools?. In *Cognitive tools for learning* (pp. 1-6). Springer, Berlin, Heidelberg. Koehler, M.J., & Mishra, P. (2008). Introducing TPCK. In AACTE Committee on Innovation and Technology (Ed.), The
- handbook of technological pedagogical content knowledge (TPCK) for educators (pp. 3-29). New York, NY: Routledge. Koh, J. H. L., Woo, H. L. & Lim, W. Y. (2013). Understanding the relationship between Singapore preservice teachers' ICT course experiences and technological pedagogical content knowledge (TPACK) through ICT course evaluation.
- Educational Assessment, Evaluation and Accountability, 25(4), 321–339. https://doi.org/10.1007/s11092-013-9165-y
 Koh, J. H. L. & Chai, C. S. (2014). Teacher clusters and their perceptions of technological pedagogical content knowledge (TPACK) development through ICT lesson design. Computers & Education, 70, 222–232.
 - https://doi.org/10.1016/j.compedu.2013.08.017
- Kramarski, B., & Michalsky, T. (2010). Preparing pre-service teachers for self-regulated learning in the context of technological pedagogical content knowledge. *Learning and Instruction*, 20(5), 434–447. https://doi.org/10.1016/j.learninstruc.2009.05.003
- Krauskopf, K., Foulger, T. S., & Williams, M. K. (2018). Prompting teachers' reflection of their professional knowledge: A proof-of-concept study of the Graphic Assessment of TPACK Instrument. Postprints der Universität Potsdam : Humanwissenschaftliche Reihe: Vol. 466. Universität Potsdam.
- Krauskopf, K., Zahn, C., & Hesse, F. W. (2012). Leveraging the affordances of Youtube: The role of pedagogical knowledge and mental models of technology functions for lesson planning with technology. *Computers & Education*, 58(4), 1194– 1206. https://doi.org/10.1016/j.compedu.2011.12.010
- Latoschik, M. E., Kern, F., Stauffert, J. P., Bartl, A., Botsch, M., & Lugrin, J. L. (2019). Not alone here?! scalability and user experience of embodied ambient crowds in distributed social virtual reality. *IEEE transactions on visualization and computer graphics*, 25(5), 2134-2144.
- Linn, M. C., Eylon, B.-S., & Davis, E. A. (2004). The knowledge integration perspective on learning. In M. C. Linn, E. A. Davis, & P. Bell (Eds.), Internet environments for science education (pp. 29–46). Mahwah, NJ: Lawrence Erlbaum Associates.
- Makransky, G., Terkildsen, T. S. & Mayer, R. E. (2019). Adding immersive virtual reality to a science lab simulation causes more presence but less learning. *Learning and Instruction*, 60, 225–236. https://doi.org/10.1016/j.learninstruc.2017.12.007
- Marquart, C. L., Hinojosa, C., Swiecki, Z., Eagan, B., & Shaffer, D. W. (2018). Epistemic Network Analysis (Version 1.7.0) [Software]. Available from http://app.epistemicnetwork.org
- Mishra, P., & Koehler, M. (2006). Technological Pedagogical Content Knowledge: A Framework for Teacher Knowledge. *Teachers College Record*, 108(6), 1017–1054. https://doi.org/10.1111/j.1467-9620.2006.00684.x.
- Mishra, P. (2019). Considering Contextual Knowledge: The TPACK Diagram Gets an Upgrade. Journal of Digital Learning in Teacher Education, 35(2), 76–78. https://doi.org/10.1080/21532974.2019.1588611
- Reigeluth, C. M. (2015). Instructional Theory and Technology for the New Pardigm of Education. In B. Khan, & M. Ally, *International Handbook of E-learning, Volume 1* (S. 79-92). New York, London: Routledge.Richard, R., & Rodway, M. R. (1992). The peer consultation group: A problem-solving perspective. *The Clinical Supervisor*, 10(1), 83–100.
- Ripka, G., Grafe, S., & Latoschik, M. E. (2021, July). Peer group supervision in Zoom and social VR-Preparing preservice teachers for planning and designing digital media integrated classes. In *EdMedia+ Innovate Learning* (pp. 602-616). Association for the Advancement of Computing in Education (AACE).
- Roth, D., Kleinbeck, C., Feigl, T., Mutschler, C., & Latoschik, M. E. (2018, March). Beyond replication: Augmenting social behaviors in multi-user virtual realities. In 2018 IEEE Conference on Virtual Reality and 3D User Interfaces (VR) (pp. 215-222). IEEE.

- Ruis, A. R. (2018). "Trois Empreintes d'un Même Cachet": Toward a Historical Definition of Nutrition. In E. T. Ewing & K. Randall (eds.), Viral networks: Connecting digital humanities and medical history (pp. 179–212). Blacksburg: VT Publishing.
- Ruis, A. R., Rosser, A. A., Quandt-Walle, C., Nathwani, J. N., Shaffer, D. W., & Pugh, C. M. (2018). The hands and head of a surgeon: Modeling operative competency with multimodal epistemic network analysis. American Journal of Surgery, 216(5), 835-840.
- Shaffer, D. W. (2012). Models of situated action: Computer games and the problem of transfer. In C. Steinkuehler, K. D. Squire, & S. A. Barab (Eds.), Games, learning, and society: Learning and meaning in the digital age (pp. 403–431). Cambridge, UK: Cambridge University Press.
- Shaffer, D. W. (2017). Quantitative ethnography. Madison, WI: Cathcart Press.
- Shaffer, D. W., Collier, W., & Ruis, A. R. (2016). A tutorial on epistemic network analysis: Analyzing the structure of connections in cognitive, social, and interaction data. Journal of Learning Analytics, 3(3), 9–45.
- Shaffer, D. W., Hatfield, D. L., Svarovsky, G. N., Nash, P., Nulty, A., Bagley, E. A., ... Frank, K. (2009). Epistemic network analysis: A prototype for 21st century assessment of learning. International Journal of Learning and Media, 1(1), 1–21.
- Shaffer, D. W., & Ruis, A. R. (2017). Epistemic network analysis: A worked example of theory-based learning analytics. In C. Lang, G. Siemens, A. F. Wise, & D. Gasevic (Eds.), Handbook of learning analytics (pp. 175–187). Society for Learning Analytics Research.
- Schön, D. A. (1987). Educating the reflective practitioner: Toward a new design for teaching and learning in the professions. Jossey-Bass.
- Shulman, L. S. (1986). Those who understand: Knowledge growth in teaching. Educational researcher, 15(2), 4-14.
- Shih, C. L., & Chuang, H. H. (2012). The development and validation of an instrument for assessing college students' perceptions of faculty knowledge in technology-supported class environments. Computers & Education, 63, 109-118.
- Skarbez, R., Brooks, F., & Whitton, M. (2020). Immersion and Coherence: Research Agenda and Early Results. IEEE Transactions on Visualization and Computer Graphics, PP. https://doi.org/10.1109/TVCG.2020.2983701
- Schmidt, D. A., Baran, E., Thompson, A. D., Mishra, P., Koehler, M. J., & Shin, T. S. (2009). Technological pedagogical content knowledge (TPACK) the development and validation of an assessment instrument for preservice teachers. *Journal of research on Technology in Education*, 42(2), 123-149.
- Su, M., & Foulger, T. (2019, March). We aren't there yet: A progression of literature on TPACK measures to assess technology integration. In *Society for Information Technology & Teacher Education International Conference* (pp. 2534-2542). Association for the Advancement of Computing in Education (AACE).
- Sullivan, S. A., Warner-Hillard, C., Eagan, B. R., Thompson, R., Ruis, A. R., Haines, K., ... Jung, H. S. (2017). Using epistemic network analysis to identify targets for educational interventions in trauma team communication. Surgery, in press
- Tiede, J., & Grafe, S. (2020). International Perspectives on the Media-related Educational Competencies of Teacher Educators. A Systematic Category-Based Comparison of Competency Frameworks. In D. Schmidt-Crawford (Ed.), Proceedings of Society for Information Technology & Teacher Education International Conference (pp. 1251– 1260). Online: Association for the Advancement of Computing in Education (AACE).
- Tietze, K.-O. (2010). Wirkprozesse und personenbezogene Wirkungen von kollegialer Beratung: Theoretische Entwürfe und empirische Forschung. VS Verlag für Sozialwissenschaften / GWV Fachverlage GmbH, Wiesbaden.
- Tietze, K.-O. (2013). Kollegiale Beratung: Problemlösungen gemeinsam entwickeln (6. Aufl., Orig.-Ausg). Miteinander reden: Vol. 61544. Rowohlt-Taschenbuch-Verl. http://www.socialnet.de/rezensionen/isbn.php?isbn=978-3-499-61544-3
- Tietze, K.-O. (2018). Kollegiale Beratung: Problemlösungen gemeinsam entwickeln (9. Auflage). Miteinander reden: Vol. 61544. Rowohlt Taschenbuch Verlag.
- Tseng, J.-J., Chai, C. S., Tan, L., & Park, M. (2020). A critical review of research on technological pedagogical and content knowledge (TPACK) in language teaching. *Computer Assisted Language Learning*, 1–24. https://doi.org/10.1080/09588221.2020.1868531
- Tulodziecki, G., Herzig, B., & Grafe, S. (2013). Gestaltungsorientierte Bildungsforschung und Didaktik. Theorie Empirie Praxis. Klinkhardt.
- Tulodziecki, G., Herzig, B., & Blömeke, S. (2017). Gestaltung von Unterricht: Eine Einführung in die Didaktik (3., überarbeitete und erweiterte Auflage). UTB Erziehungswissenschaft, Schulpädagogik, Allgemeine Didaktik: Vol. 3311. Verlag Julius Klinkhardt. http://www.utb-studi-e-book.de/9783838547947
- Tulodziecki, G., Herzig, B. & Grafe, S. (2021). Medienbildung in Schule und Unterricht: Grundlagen und Beispiele (3. durchges. u. aktual. Aufl.). UTB GmbH.
- Wenger, E. (1999). Communities of practice: Learning, meaning, and identity. Cambridge, UK: Cambridge University Press. Retrieved from http://books.google.com/books?id=heBZpgYUKdAC&pgis=1
- Wiesner, C. & Schreiner, C. (2020). Digitale Kompetenzen: Computational Thinking als Basis eines Kompetenzmodells. Bildung und Digitalisierung, 29–50. https://doi.org/10.5771/9783748906247-29.
- Wooldridge, A. R., Carayon, P., Eagan, B. R., & Shaffer, D. W. (2018). Quantifying the qualitative with epistemic network analysis: A human factors case study of task-allocation communication in a primary care team. IIE Transactions on Healthcare Systems Engineering, in press.

6.3 Conclusion

The aim of Study 4 was to evaluate how the refined pedagogical concept can promote preservice teachers' TPACK as a metaconceptual awareness using social VR. The analysis of the GATI diagrams and the pre-service teachers' epistemic network analysis (ENA) confirmed the successful implementation of the pedagogical concept in practice. The generated design principles led to the development of pre-service teachers' TPACK to successfully implement social VR in class (see Section 6.2). The results of the GATI diagrams demonstrated that preservice teachers participating in social VR constructed more complex mental models of TPACK than their peers who used Zoom (see Section 6.2). The data from the portfolios revealed that students in the social VR group could anticipate the challenges they would face when using social VR in the classroom (see Section 6.2). Drawing connections between their TK, PK, and CK and the subdomains, they considered more contextual factors when designing their lesson plans. Students also increasingly involved CK in the subject matter, going beyond the knowledge domains fostered by the pedagogical concept (see Section 6.2). One possible implication here is that the pre-service teachers had hands-on experience, whereas those participating in Zoom did not (see Section 6.2).

The development of the pedagogical concept in this work has two implications for the future integration of social VR into ITE. First, the implementation of social VR depends highly on the interplay of pedagogical and technical components and their contextual factors. Through the iterative design and development processes, it was possible to create and implement a pedagogical concept for the promotion of pre-service teachers' TPACK as metaconceptual awareness using social VR that considers important contextual pedagogical and technical aspects.

As VR hardware and software is developing quickly, and the choice of hardware and software influences the nature of teaching and learning activities, the pedagogical concept will need further refinement in the future: A close adaptation of pedagogical and technological design elements is indispensable because these elements are interdependent.

7. Conclusion and Outlook

Against the background of the need to provide future teachers with the knowledge to plan and design school lessons using innovative technologies in a practice- and action-oriented way, the aim of this dissertation was to iteratively develop a pedagogical concept that promotes pre-service teachers' TPACK while considering potential social VR affordances. In four studies, the following main research questions were addressed:

- What do student teachers and teacher educators expect of a successful virtual reality application in Initial Teacher Education (ITE?
- How do pre-service teachers perceive teaching and learning activities in social VR?
- How should a pedagogical concept for remote initial teacher education be designed to promote the metacognitive learning processes of pre-service teachers?
- How do pre-service teachers perceive these learning processes in video-based communication and social VR? How did pre-service teachers' TPACK develop using a social VR learning environment prototype compared to a video-conferencing system?

The design and development process of the pedagogical concept to promote pre-service teachers' TPACK as metacognitive awareness in social VR took place through the close interaction of research and practice in a DBR process.

Figure 3



Adaptation of Euler's (2014) DBR structure model

The results of the research process are summarized in this chapter, with the four studies assigned to the phases of Euler's (2014) design cycle (see Figure 3). Furthermore, the limitations of the studies are summarized. Finally, possible consequences and implications are derived for research and teaching practice with special consideration of the two following frames of reference, which are central to this thesis: (a) TPACK as a metaconceptual awareness (Section 2.1) and (b) social VR's affordances for teaching and learning processes (Section 2.2).

7.1 Summary of the Development Cycles and Their Research Results

The design and development processes of the pedagogical concept started by identifying the problem of practice to promote the TPACK of pre-service teachers using social VR. Against the background of a literature review of existing pedagogical and technical requirements (Section 3.2) in Study 1, a needs analysis was carried out to explore what student teachers and teacher educators expect of a successful VR application in ITE. To consider important contextual factors from the perspective of pre-service teachers and teacher educators, 12 pre-service teachers and 10 teacher educators from a German university participated in guideline-based interviews (see Section 3.2). From the results of a content analysis (Mayring, 2015), it was possible to derive key design features of the pedagogical concept to foster pre-service teachers' TPACK in social VR based on existing literature and the in-practice experiences of the target groups (Sections 3.2 and 3.3). Important design features included close guidance and support for using VR hardware and software; accessible and user-friendly materials; elements of constructivist learning theory; and disruption-free communication, collaboration, and interaction in a social VR environment, keeping design elements minor and clean (Section 3.2).

The aim of Study 2 was to analyze how pre-service teachers perceive teaching and learning activities in fully immersive social VR and to derive consequences for further refinement of the pedagogical concept to promote the TPACK of pre-service teachers (Section 4.2). Thus, the results were used to further analyze and explore the design assumptions for teaching and learning processes in social VR. The second study represented the first development and refinement step of the first iterative cycle, which involved the testing and further refinement of design principles (see Euler, 2014). Accordingly, a teacher-centered and learner-centered seminar unit using social VR was designed according to the requirements identified in Study 1. Design principles comprised action-oriented, constructivist, and situated-learning activities and changing social formats.

Further, pedagogical and technical support were integrated. To prevent potential technical obstacles to pre-service teachers' use of social VR hardware and software, the study took place in a lab (see Section 4.2). The seminar unit was implemented with a convenience sample of three groups, each of five students, in the ITE program at a German university. The results of a content analysis (Mayring, 2015) of the three group interviews showed that throughout the seminar units, particularly in the teacher-centered scenario, content representation on virtual boards led to fatigue and decreased concentration among students (Section 4.2). Visual stimuli in the social VR seminar room were kept to a minimum; this strategy was received positively as pre-service teachers felt more present and less distracted than in similar face-to-face settings (Section 4.2). Because of missing verbal and nonverbal abstract avatar representations, students needed more time to carry out their tasks in the group work, substituting missing communications signals with their avatar's body movements or anticipating speakers' turns (Section 4.2). At the same time, positive effects were noted, such as more concentration on the spoken word and being able to state one's mind without fearing criticism (Section 4.2). Target groups mainly imagined possible teaching and learning scenarios in social VR, known from face-to-face teaching and learning scenarios (Section 4.2). In line with the assumption that the feeling of social co-presence positively influences social learning processes and thus benefits collaborative tasks (Garrison & Akyol, 2013) in social VR, students showed excitement, motivation, and engagement throughout the group task (Section 4.2).

The aim of Study 3 was to further refine the pedagogical concept to promote the metacognitive learning processes of pre-service teachers and analyze how such teachers perceive their learning processes in social VR compared with video-based communication. The results of Study 2 were used to adapt the design principles in Study 3 (see Section 5.2). With the key objective of promoting TPACK in social VR, the primary approach of the pedagogical concept not only had to consider social VR's affordances and challenges but also had to enable teaching and learning processes leading to the metaconceptual knowledge of how to implement social VR successfully in class (Section 5.2). Hence, guiding the second iterative cycle (see Figure 3; Euler, 2014) of a pedagogical concept promoting TPACK as metaconceptual awareness in social VR in a remote teaching and learning setting were the following design principles (Section 5.2):

- Active learner engagement with the subject matter.
- Group tasks for small groups.
- Flipped classroom principles and action orientation.

- Clear and structured content representation to prevent students' cognitive overload.
- Use of a potential feeling of togetherness in social VR.
- Close support from a pedagogical and technical perspective.

To address metaconceptual learning processes in particular, a peer group supervision approach was implemented (Tietze, 2010) with joint reflecting, scaffolding, or problem solving as design principles (Section 5.2). The pedagogical concept was implemented with a convenience sample of 17 pre-service teachers at a German university. The students participated in two iterative cycles of peer group supervision, performing design tasks in groups in social VR (n = 12) or using a video-conferencing system (n = 5). Data were collected via guided reflective video statements of students and semi-structured group interviews. Qualitative content analysis (Mayring, 2015) of the group interviews and reflective video statements was deductively carried out according to the four main features of peer group supervision delineated by Tietze (2010): group size, the design task, the structure of peer group supervision, and the communication and interaction processes involved in social VR and Zoom (Section 5.2). The results showed that students perceived the design task as complex and not easy to carry out. Nevertheless, it led to reflective and problemsolving processes, allowing for the consideration of multiple perspectives on and potential solutions to the problem at hand (Section 5.2). At first, the structure of the peer group supervision was hard for the students to follow, but from the second repetition on, they became used to it (Section 5.2). Furthermore, the steps of the peer group supervision process served some students as facilitators of communication processes (Section 5.2). The results emphasized the recurring importance of integrating multiple breaks to relieve cognitive load (see Chandler & Sweller, 1991; Parong & Mayer, 2021; Section 5.2).

Students positively perceived the interaction with peers in social VR and compared it favorably to face-to-face communication and interaction in reality. Notably, this sense of community had significant value for most students because they felt isolated during the pandemic (Section 5.2). Moreover, compared with Study 2 (Section 4.2), the missing social cues represented an advantage because students tried to follow the peer group supervision cycle's structure strictly and used it to initiate and stop speaker turns clearly (Section 5.2). With regard to structuring the teaching and learning process, the study results suggested the importance of enough time being available; this would allow breaks and repetitions to facilitate cognitive and collaboration processes in social VR. It also became apparent that there was still a need for more supportive measures to facilitate

social VR's hardware and software use (Section 5.2). Finally, the findings underlined that the right balance between confronting students with pedagogical and technical stimuli and not overwhelming them plays a crucial role in the success of teaching and learning processes in social VR (Section 5.2).

The aim of Study 4 was to evaluate how the refined pedagogical concept can foster pre-service teachers' TPACK as metaconceptual awareness using social VR. After adapting the design principles according to the newly gained insights, the pedagogical concept seemed robust enough to be implemented throughout a semester, covering the full complexity of its future in-practice use for remote teaching and learning processes (Section 6.2). Therefore, investigation was undertaken of how pre-service teachers' TPACK develops when they use a social VR learning environment prototype compared to when they use a video-conferencing system (Section 6.2). The pedagogical concept to foster pre-service teachers' TPACK as a metaconceptual awareness was implemented with a convenience sample of 14 pre-service teachers at a German university.

Following the principles of action-oriented teaching and learning by design (Koehler & Mishra, 2009; Tulodziecki et al., 2017) and peer group supervision (Tietze, 2010), pre-service teachers designed VR-integrated lesson designs in an iterative peer feedback process using social VR (n = 7) or a video-conferencing system (n = 7) and documented their results and further reflections in a portfolio. After each peer group supervision cycle and teacher feedback, the student teachers handed in a new version of their lesson design (Section 6.2).

With regard to data collection, the pre-service teachers' GATI diagrams (Krauskopf et al., 2018) illustrated how they estimated their TPACK before and after the seminar (Section 6.2). The portfolios document the pre-service teachers' design and development process of successfully constructing a lesson plan integrating social VR into the curriculum's framework (Section 6.2). GATI diagrams and portfolio data were coded and qualitatively evaluated with ENA (Shaffer et al., 2016). The results showed that the ENA networks of both groups developed in the PK knowledge domains (Section 6.2). The pedagogical concept and assessment criteria show a pedagogical focus that explains the high frequency of PK-related connections (Section 6.2). The lesson designs of both groups progressed over the semester from the first dimension (factual knowledge) to the third dimension (procedural knowledge; Section 6.2). Few drew connections in the fourth dimension of metacognitive knowledge (Section 6.2).

Overall, both groups developed quite homogenously (Section 6.2). However, a closer look revealed that the lesson designs of the social VR group showed more complex connections involving TK of the second dimension at an earlier stage of the design progress and increased inclusion of CK (see Section 6.2). The results led to the inference that the design principles of the pedagogical concept favored the development of TPACK in social VR (Section 6.2). Based on the overall impression and results, it can be concluded that students who worked with social VR regularly were able to derive implications for planning and designing lessons based on their in-practice experience (Section 6.2).

Experiencing the effort connected to implementing social VR and testing social VR hardware and software, pre-service teachers observed firsthand insecurities, joy, and frustration when using social VR (Section 6.2). Transferring this new understanding to the perspective of their future students, the pre-service teachers gave reasonable justification in their lesson plans for their desire to use a specific VR software for a particular learning objective (Section 6.2). To do so, they had to consider the curriculum of the subject matter and include their CK to weigh the added value of social VR for the individual lesson plan (Section 6.2). Moreover, they were aware of the benefits and challenges of the chosen social VR hardware for school implementation (Section 6.2).

7.2. Limitations of the Studies

The findings presented in this dissertation are subject to certain limitations, as discussed below. Concerning the data, the findings were based on convenience samples of pre-service teachers in Germany. The study sample consisted of student teachers in their first phase of teacher education. Therefore, the findings cannot be transferred directly to pre-service teachers in the second phase of teacher education in Germany or to pre-service teachers in other countries with different models of ITE. In addition, the prerequisites of the participating students varied according to their study progress and focus. Furthermore, a positive sample bias can be assumed because students could choose the course from a variety of other courses covering various topics. Because of the small sample sizes, the results of the exploratory case studies are not representative; however, the precise description of learning prerequisites, targets, teaching, and learning activities, social forms, and digital media allows for a replication or an adaptation of the study for other target groups and aims. Accordingly, the findings of this study provide helpful insights for future research and

practice with regard to implementing social VR in ITE and evaluating the promotion of pre-service teachers' TPACK development.

There are also limitations concerning the instruments used to map pre-service teachers' TPACK development in Study 4. Students developed a GATI diagram for the self-reflection process at the beginning and end of the semester in which Study 4 was conducted, showing their TPACK progress over the semester (see Section 6.2). However, the GATI diagram is more of a tool for reflection than a test instrument for measurement. Krauskopf et al. (2018) saw the potential of the GATI procedure as a measurement tool, but to confirm this assumption, more research is needed, including studies with larger sample sizes (see Krauskopf et al., 2018).

The ENA used as the main method to map the development of TPACK comes with some challenges. The coding guide must be highly detailed, and raters must be intensively trained. Because the construction of ENA networks depends significantly on the references made between the codes, coders must set references according to the respective context given by sense units (see Shaffer et al., 2016). To ensure that coders interpret sense units as similarly as possible, several repetitive alignment processes are necessary. While this approach enables the mapping of dynamic cognitive processes, it also carries the risk of open interpretive spaces. To keep such spaces as low in number as possible, in this study, reoccurrences of codes were counted to back up assumptions made about the created mental model networks (Section 6.2).

7.3 Implications and Outlook

As a result of the iterative design and research process, the created pedagogical concept led to the successful promotion of pre-service teachers' metaconceptual awareness of TPACK using social VR. The main underlying concepts of the research questions are as follows: (a) TPACK as metaconceptual awareness and (b) social VR's affordances for teaching and learning processes in ITE. Against the background of these theoretical starting points, the implications for research and practice from the findings of this work are outlined below. The next section identifies fields of action for ITE that require further investigation with regard to fostering TPACK using social VR in ITE.

7.3.1 Modeling Technological Pedagogical Content Knowledge as Metaconceptual Awareness

In the scientific discourse, systematic overviews show the variety of TPACK research studies, but they rarely refer to TPACK as metacognitive awareness (Krauskopf et al., 2012; Krauskopf et al., 2015; Krauskopf et al., 2018). Consideration of metacognitive learning processes could provide beneficial insights into the promotion of the TPACK of pre-service teachers in future research and practice.

This study's findings confirm that pre-service teachers developed TPACK as metaconceptual awareness with the help of metacognitive learning activities in social VR (see Chapter 6.2). TPACK is a framework for teacher knowledge (Mishra & Koehler, 2006); accordingly, further efforts must be made to model and validate the relationship between TPACK and media-related educational competencies. Regarding the goal that pre-service teachers should be able not only to use social VR successfully but also prepare their future students for the self-determined and competent use of social VR (Chapter 1), this requirement implies shifting the main focus of TPACK from the use of educational technology to integrating aspects of media literacy education.

This work focused mainly on the assumed underlying cognitive processes connected to TPACK's development (see Section 6.2). Parong and Mayer (2021) found that emotional arousal based on VR experiences influences learning outcomes and is thus a vital aspect to consider when designing pedagogical concepts for social VR. Accordingly, exploring the affective effects of using social VR on TPACK's development would allow helpful assumptions about how to implement and design social VR software in ITE.

Concerning implications for the in-practice use of social VR to promote TPACK, it would be illuminating to implement student teachers' lesson designs (see Section 6.2) in practice at schools. Using this approach, long-term studies could investigate whether student teachers can transfer their knowledge from theory to practice and apply the metaconceptual strategies acquired in ITE. Based on these findings, further implications for the design, development, and evaluation of pedagogical concepts supporting the promotion of TPACK as metaconceptual awareness could be derived.

7.3.2 Measuring Technological Pedagogical Content Knowledge as Metaconceptual Awareness

Previous research has identified instruments to measure pre-service teachers' TPACK (Section 2.1). TPACK as metaconceptual awareness comprises dynamic metacognitive processes that take place within a two-level transformation process of constructing mental models (first transformation) and TPACK awareness (second transformation) (see Krauskopf et al., 2012; Krauskopf et al., 2015; Krauskopf et al., 2018). This work used the GATI diagram (Krauskopf et al., 2018) and ENA (Shaffer et al., 2016) as two possible tools to map the transformation processes of pre-service teachers' TPACK. Although both instruments proved suitable in Study 4 (see Section 6.2), further research on the topic is necessary. Concerning the GATI process (Krauskopf et al., 2018), in Study 4, student teachers with a study focus on special education indicated that they saw PCK as one of the leading knowledge components and not as a subdomain (see Section 6.2). This perception would have consequences for the interpretation of the mental models and the design of the pedagogical concept. Accordingly, the repetition of the evaluation process with larger groups divided according to study focus and learning prerequisites would allow for more assumptions about the specifications for each TPACK domain. The ENA (Shaffer et al., 2016) provided a way to map TPACK as metaconceptual awareness (see Section 6.2). Since coding and mapping are extensive and require many resources, a possible approach to facilitate the procedure for future in-practice implementations in ITE is to investigate the involvement of AI automation processes.

7.3.3 Teaching and Research with Social Virtual Reality: Implications for Ethics and Sustainability

When using social VR in class, teacher educators and pre-service teachers must decide on appropriate social VR hardware and software for their learners and be aware of possible risks and dangers to which their learners are exposed. Moreover, ethical questions arise with regard to the use of social VR hardware and software because VR's characteristics can affect the user's emotional, mental, or physical state (Chapter 1). Adams et al. (2018) categorized VR's risks into the following: (a) manipulation and violation of immersive experiences, (b) physical harm, and (c) data collection and inferences. Exposing users to a multisensory world in which the stimuli resemble real ones (Gonzalez-Franco & Lanier, 2017) turns VR into a powerful medium.

Harzenmoser et al. (2019), for example, demonstrated that VR could cause psychological trauma and a high level of emotional response, leading to physical suffering. In line with this finding, Slater et al. (2020) listed numerous novel possible harms and risks associated with experiencing "superrealism" (p. 3) created by mixed reality technologies. Accordingly, future research should carefully weigh the expected positive effects and possible risks when using social VR for teaching and learning processes.

Along with social VR's effects on the state of mind and body of users, data protection is an important issue for further research and practice. The provider retrieves biometrics-related information and the IP address, as well as data about the environment, the location's dimensions, and users' motion. Moreover, when using realistic 3D avatars, the question of ownership of such avatars arises. With regard to the rapid further development of technology, these aspects are not only important for future research but also relate to the data literacy of pre-service teachers.

Another dilemma for future research and practice is the sustainable choice of VR hardware and software. Since new VR hardware is developed quickly, IT support may end soon after the headset is bought or the data regulations may change; this may lead to a short period of use and unnecessary e-waste if the hardware soon becomes unusable.

For the future implementation of social VR in ITE, implications for ethics and sustainability are not only important for research but also for TPACK acquisition and curriculum development.

7.3.4 Social Virtual Reality as an Authentic Learning Environment in Initial Teacher Education

As the results of Study 4 showed, using social VR in ITE supports pre-service teachers' construction of complex mental models (see Section 6.2). Social VR's affordances offer many possibilities to increase the authenticity of learning environments for ITE (see Wang et al., 2021). The design elements of the social VR platform applied in this work focused on real-world scenarios. Accordingly, seminar rooms and furnishings were based on standard design features from the real world (see Sections 4.2, 5.2, and 6.2). Future development and research on the applied social VR prototype could extend design features to allow a variety of learning scenarios and settings so that pre-service teachers could interact and engage with objects, travel back in time, or teleport to other countries. Furthermore, implementing pedagogical agents in social VR may positively influence teaching and learning processes (see Clarebout et al., 2002, for an

overview). Such agents could give immediate feedback cues, comment on the difficulties or benefits of learning activities, point out design elements supporting or hindering learning processes, and act as mentors or peers. Whereas some see pedagogical agents only as a further unnecessary cognitive load for learners (Clark & Choi, 2007), others have proved their potential for positively influencing learning outcomes (Moreno, 2005). In a study conducted during the COVID-19 pandemic, Petersen et al. (2021) found that although the pedagogical agent contributed—as expected-to a higher cognitive load, realistic agents could support the acquisition of conceptual information. Furthermore, they found that realistic agents increase the feeling of social presence (Petersen et al., 2021), which is assumed to influence the co-construction of knowledge and collaborative processes in VR (Garrison & Akyol, 2013). In particular, in such social learning processes as peer group supervision, the use of an agent could either increase the feeling of co-presence or give important feedback to pre-service teachers.

7.3.5 Creating Collaborative Cultures in Social Virtual Reality and Beyond

Previous studies have confirmed the overall positive effect of social relations on teachers' wellbeing, supporting resilience and positive emotions throughout the pandemic (see McCallum, 2021). With the increasing workload and complexity of the teaching profession, mental health and teacher wellbeing have gained in importance (see Hascher & Waber, 2021). Throughout the research and development cycles of this work, positive side effects appeared throughout the collaboration in peer group supervision cycles. Apart from the formal learning setting, students started to form informal study groups outside the university to consult each other regularly on problems, including not only how to organize their study but also pandemic struggles (see Section 6.2). With the help of social media, students coordinated and shared their workloads. It is assumed that peer group supervision contributed not only to the acquisition of metaconceptual strategies for implementing social VR (see Section 6.2) but also to the development of collaboration strategies for coping with workload, the shift to remote studies, and the isolation experienced because of the pandemic. Serving as an informal learning setting outside the seminar framework, peer collaboration was a significant incentive offered by the pedagogical concept (see Section 6.2). Looking forward to possible 21st-century challenges, the curriculum in ITE could contribute to teachers' preparation by offering incentives for the acquisition of coping strategies embracing a collaborative learning culture.

Overall, the present work has demonstrated that in an iterative design and development process, a pedagogical concept was designed that promoted pre-service teachers' TPACK as metaconceptual awareness in a social VR learning environment. TPACK as metaconceptual awareness allows pre-service teachers to acquire metacognitive strategies to successfully plan and design target-oriented social VR-integrated lessons for each grade. As a crucial incentive for learning activities allowing metacognitive processes, such as reflecting, scaffolding, and problem solving, peer group supervision represented an efficient way to combine the benefits of social VR with the pedagogical principles of constructive, situated, and action-oriented teaching and learning. Thus, this work contributes to the current debate in educational policy, educational research, and educational practice concerning pre-service teachers' requirements for the sound and successful implementation of educational technology in their future teaching practice. It is hoped that it will contribute to transforming and opening teaching and learning processes in ITE toward a learner-centered, collaborative, and sustainable learning culture that will help prepare pre-service teachers for future teaching practice and upcoming 21st-century challenge.

8. References

- Abbitt, J. T. (2011). An investigation of the relationship between self-efficacy beliefs about integration and technological pedagogical content knowledge (TPACK) among preservice teachers. *Journal of Digital Learning in Teacher Education*, 27(4), 134–143. https://doi.org/10.1080/21532974.2011.10784670
- Abfalter, D., Zaglia, M. E., & Mueller, J. (2012). Sense of virtual community: A follow up on its measurement. *Computers in Human Behavior*, 28(2), 400–404. https://doi.org/10.1016/j.chb.2011.10.010
- Adams, B. S., Cummins, M., Davis, A., Freeman, A., Hall Giesinger, C., & Anantharayanan,
 V. (2017). NMC Horizon Report: 2017 Higher Education Edition. Austin, TX: The New
 Media Consortium.
- Adams, D., Bah, A., Barwulor, C., Musaby, N., Pitkin, K., & Redmiles, E. M. (2018). Ethics emerging: the story of privacy and security perceptions in virtual reality. In *Fourteenth Symposium on Usable Privacy and Security (SOUPS 2018)* (pp. 427-442).
- Altrichter, H., & Posch, P. (2007). Lehrerinnen und Lehrer erforschen ihren Unterricht (4th ed.). Klinkhardt.
- Anderson, L. W., Krathwohl, D. R., Airasian, P. W., Cruikshank, K. A., Mayer, R. E., Pintrich,
 P. R., Raths, J., &Wittrock, M. C. (Eds.). (2001). A taxonomy for learning, teaching, and assessing: A revision of Bloom's taxonomy of educational objectives (Complete ed.). Longman.
- Anderson, T., & Shattuck, J. (2012). Design-based research: A decade of progress in educational research? *Educational Researcher*, 41(1), 16–25.
- Andrist, S., Collier, W., Gleicher, M., Mutlu, B., & Shaffer, D. (2015). Look together: Analyzing gaze coordination with epistemic network analysis. *Frontiers in Psychology*, 6(1016).
- Angeli, C., & Valanides, N. (2013). Technology mapping: An approach for developing technological pedagogical content knowledge. *Journal of Educational Computing Research*, 48(2), 199-221.
- Arastoopour, G., Chesler, N., Shaffer, D., & Swiecki, Z. (2015). Epistemic network analysis as a tool for engineering design assessment. ASEE Annual Conference and Exposition, Conference Proceedings, 122. https://doi.org/ 10.18260/p.24016

- Archambault, L. M., & Barnett, J. H. (2010). Revisiting technological pedagogical content knowledge: Exploring the TPACK framework. *Computers & Education*, 55(4), 1656– 1662. https://doi.org/10.1016/j.compedu.2010.07.009
- Arnold, R. (2012). Wie man lehrt, ohne zu belehren: 29 Regeln für eine kluge Lehre: das LENA-Modell. Systemische Pädagogik. Auer.
- Bailenson, J. N. (2021). Nonverbal overload: A theoretical argument for the causes of Zoom fatigue. *Technology, Mind, and Behavior*, 2(1). <u>https://doi.org/10.1037/tmb0000030</u>
- Bailenson, J. N., Yee, N., Blascovich, J., Beall, A. C., Lundblad, N., & Jin, M. (2008). The use of immersive virtual reality in the learning sciences: Digital transformations of teachers, students, and social context. *Journal of the Learning Sciences*, 17(1), 102–141. https://doi.org/10.1080/10508400701793141
- Bandura, A. (1979). Self-referent mechanisms in social learning theory. *American Psychologist*, 34(5), 439–441. https://doi.org/10.1037/0003-066X.34.5.439.b
- Bannan-Ritland, B. (2003). The role of design in research: the integrative learning design framework. *Educational Researcher*, 32(1), 21–24.
- Baran, E., & Uygun, E. (2016). Putting technological, pedagogical, and content knowledge (TPACK) in action: An integrated TPACK-design-based learning (DBL) approach.
 Australasian Journal of Educational Technology, 32(2).
 https://doi.org/10.14742/ajet.2551
- Billingsley, G., Smith, S., Smith, S., & Meritt, J. (2019). A systematic literature review of using immersive virtual reality technology in teacher education. *Journal of Interactive Learning Research*, 30(1), 65–90. https://www.learntechlib.org/primary/p/176261/paper 176261.pdf
- Blömeke, S. (2017). Erwerb medienpädagogischer Kompetenz in der Lehrerausbildung: Modell der Zielqualifikation, Lernvoraussetzungen der Studierenden und Folgerungen für Struktur und Inhalte des medienpädagogischen Lehramtsstudiums. *MedienPädagogik: Zeitschrift Für Theorie Und Praxis Der Medienbildung*, 3 (Jahrbuch Medienpädagogik), 231–244. https://doi.org/10.21240/mpaed/retro/2017.07.13.X
- Borthwick, A. C., & Hansen, R. (2017). Digital literacy in teacher education: Are teacher educators competent? *Journal of Digital Learning in Teacher Education*, 33(2), 46–48. https://doi.org/10.1080/21532974.2017. 1291249
- Boud, D., Cohen, R., & Sampson, J. (2001). Peer Learning in higher Education: Learning from & with each other. Kogan Page.
- Bower, M. (2008). Affordance analysis-matching learning tasks with learning technologies. *Educational Media International*, 45(1), 3-15.
- Bower, M., & Sturman, D. (2015). What are the educational affordances of wearable technologies? *Computers & Education*, 88, 343-353.
- Brown, A. L. (1992). Design experiments: Theoretical and methodological challenges in creating complex interventions in classroom settings. *Journal of the Learning Science*, 2, 141-178.
- Browne, N., & Cooper, E. (2000). Pedagogically appropriate integration of informational technology in an elementary preservice teacher education program. *Journal of Technology and Teacher Education*, 8(3), 219–229.
- Bucy, E. P. (2017). Nonverbal Cues. *The International Encyclopedia of Media Effects*, 1–11. https://doi.org/10.1002/9781118783764.wbieme0199
- Burdea, G. C., & Coiffet, P. (2003). Virtual reality technology (2nd ed.). John Wiley & Sons Inc.
- Chai, C. S. (2018). Teacher professional development for science, technology, engineering and mathematics (STEM) education: A review from the perspectives of technological pedagogical content (TPACK). *The Asia-Pacific Education Researcher*, 28(1), 5–13. <u>https://doi.org/10.1007/s40299-018-0400-7</u>
- Chai, C. S., Koh, J. H. L., & Tsai, C.C. (2010). Facilitating preservice teachers' development of technological, pedagogical, and content knowledge (TPACK). *Journal of Educational Technology & Society*, 13(4), 63-73.
 - Chandler, P., & Sweller, J. (1991). Cognitive load theory and the format of instruction. *Cognition and Instruction*, 8(4), 293–332. https://doi.org/10.1207/s1532690xci0804_2
- Clarebout, G., Elen, J., Johnson, W.L., & Shaw, E. (2002). Animated Pedagogical Agents: An Opportunity to be Grasped? *Journal of Educational Multimedia and Hypermedia*, 11(3), 267-286. Association for the Advancement of Computing in Education (AACE). https://www.learntechlib.org/primary/p/9270/
- Clark, R. E., & Choi, S. (2007). The questionable benefits of pedagogical agents: Response to veletsianos. *Journal of Educational Computing Research*, 36(4), 379–381. https://doi.org/10.2190/2781-3471-67mg-5033
- Cleveland-Innes, M. F., & Garrison, R. D. (2020). *An Introduction to distance education: understanding teaching and learning in a new era* (2nd ed.). Routledge.

- Concannon, B. J., Esmail, S., & Roduta Roberts, M. (2019). Head-mounted display virtual reality in post-secondary education and skill training. *Frontiers in Education*, 4(80). https://doi.org/10.3389/feduc.2019.00080
- Conole G., & Dyke M. (2004). What are the affordances of information and communication technologies? *Research in Learning Technology*, *12*(2). https://doi.org/10.3402/rlt.v12i2.11246
- Cox, S. M. (2008). A conceptual analysis of technological pedagogical content knowledge (Publication No. 1482) [Doctoral dissertation, Brigham Young University]. BYU ScholarsArchive.
- Cress, U., Diethelm, I., Eickelmann, B., Köller, O., Nickolaus, R., Pant, H. A., & Reiss, K.
 (2018). Schule in der digitalen Transformation: Perspektiven der Bildungswissenschaften. acatech DISKUSSION. acatech.
- Dalgarno, B., & Lee, M. J. W. (2010). What are the learning affordances of 3-D virtual environments? *British Journal of Educational Technology*, 41(1), 10–32. <u>https://doi.org/10.1111/j.1467-8535.2009.01038.x</u>
- Dawley, L., & Dede, C. (2014). Situated learning in virtual worlds and immersive simulations.
 In *Handbook of research on educational communications and technology* (pp. 723-734). Springer, New York, NY.
- Dede, C. (2009). Immersive interfaces for engagement and learning. *Science*, *323*(5910), 66–69. https://doi.org/10.1126/science.1167311
- Design-Based Research Collective. (2003). Design-based research: An emerging paradigm for educational inquiry. *Educational researcher*, *32*(1), 5-8.
- Deutsches Institut für Vertrauen und Sicherheit im Internet. (2018). DIVSI. https://www.divsi.de
- DiSessa, A. A. (1988). Knowledge in pieces. In G. Forman & P. Pufall (Eds.), *Constructivism in the computer age* (pp. 47–70). Erlbaum.
- Dorairaj, S., Noble, J. & Malik, P. (2012). Understanding lack of trust in distributed agile teams:
 a grounded theory study. In 16th International Conference on Evaluation & Assessment
 in Software Engineering (EASE 2012) (pp. 81–90). IET.
 https://doi.org/10.1049/ic.2012.0011
- Dzardanova, E., Kasapakis, V., & Gavalas, D. (2018). Social Virtual Reality. *Encyclopedia of Computer Graphics and Games*, 1–3. <u>https://doi.org/10.1007/978-3-319-08234-9_204-</u> <u>1</u>

- Dzardanova, E., Kasapakis, V., Gavalas, D., & Sylaiou, S. (2019, June). Exploring aspects of obedience in VR-mediated communication. In 2019 Eleventh International Conference on Quality of Multimedia Experience (QoMEX) (pp. 1-3). IEEE.
- Eagan, B., & Hamilton, E. (2018). Epistemic network analysis of an international digital makerspace in Africa, Europe, and the US. In *annual meeting of the American education research association*.
- Eichelberger, A., & Leong, P. (2019). Using TPACK as a framework to study the influence of college faculty's beliefs on online teaching. *Educational Media International*, 56(2), 116–133. https://doi.org/10.1080/09523987.2019.1614246
- Eickelmann, B., Bos, W., Gerick, J., Goldhammer, F., Schaumburg, H., Schwippert, K., Senkbeil, M., & Vahrenhold, J. (Eds.). (2019). *ICILS 2018 #Deutschland: Computerund informationsbezogene Kompetenzen von Schülerinnen und Schülern im zweiten internationalen Vergleich und Kompetenzen im Bereich Computational Thinking* (1st ed.). Waxmann.
- Einsiedler, W. (2010). Didaktische Entwicklungsforschung als Transferförderung. Zeitschrift für Erziehungswissenschaft, 13(1), 59–81. https://doi.org/10.1007/s11618-010-0106-y
- Ertmer, P. A., & Ottenbreit-Leftwich, A. T. (2010). Teacher technology change: How knowledge, confidence, beliefs, and culture intersect. *Journal of research on Technology in Education*, 42(3), 255-284.
- Euler, D. (2011). Wirkungs- vs. Gestaltungsforschung eine feindliche Koexistenz? Zeitschrift für Berufs- und Wirtschaftspädagogik, 107(4), 520–542.
- Euler, D. (2014). Design-Research a paradigm under development. In D. Euler & P.F.E. Sloane (Eds.). Design-Based Research (15-44). Stuttgart: Franz Steiner Verlag.
- Fauville, G., Luo, M., Queiroz, A. C. M., Bailenson, J. N., & Hancock, J. (2021). Zoom Exhaustion & Fatigue Scale. SSRN Electronic Journal. https://doi.org/10.2139/ssrn.3786329
- Feierabend, S., Rathgeb, T., Kheredmand, H., & Glöckler, S. (2020). JIM-Studie 2020: Jugend, Information, Medien: Basisuntersuchung zum Medienumgang 12-bis 19-Jähriger.
 [JIM-Study 2020. Youth, Information, Media. A baseline study on media use among 12- to 19-year-olds] Medienpädagogischer Forschungsverbund Südwest.
- Feng, Z., González, V. A., Amor, R., Lovreglio, R., & Cabrera-Guerrero, G. (2018). Immersive virtual reality serious games for evacuation training and research: A systematic literature review. *Computers & Education*, 127, 252–266.

- Ferdig, R. E., Baumgartner, E., Hartshorne, R., Kaplan-Rakowski, R., & Mouza, C. (2020). *Teaching, technology, and teacher education during the COVID-19 pandemic: Stories from the field*. Association for the Advancement of Computing in Education (AACE). www.learntechlib.org/p/216903/
- Ferran, C., & Watts, S. (2008). Videoconferencing in the field: A heuristic processing model. *Management Science*, 54(9), 1565–1578. https://doi.org/10.1287/mnsc.1080.0879
- Foulger, T. S., Graziano, K. J., Schmidt-Crawford, D., & Slykhuis, D. A. (2017). Teacher Educator Technology Competencies. *Journal of Technology and Teacher Education*, 25(4), 413–448.
- Foulger, T. S., Wetzel, K., & Buss, R. R. (2019). Moving toward a technology infusion approach: Considerations for teacher preparation programs. *Journal of Digital Learning in Teacher Education*, 35(2), 79–91. <u>https://doi.org/10.1080/21532974.2019.1568325</u>
- Fowler, C. (2014). Virtual reality and learning: Where is the pedagogy? *British Journal of Educational Technology*, 46(2), 412–422. <u>https://doi.org/10.1111/bjet.12135</u>
- Garrison, D. R., & Akyol, Z. (2013). Toward the development of a metacognition construct for communities of inquiry. *The Internet and Higher Education*, 17, 84–89. <u>https://doi.org/10.1016/j.iheduc.2012.11.005</u>
- Garrison, D. R., & Akyol, Z. (2015). Thinking collaboratively in educational environments: shared metacognition and co-regulation in communities of inquiry. In J. Lock, P. Redmond & P. A. Danaher (Eds.), *Educational Developments, Practices and Effectiveness* (pp.39–52). Palgrave Macmillan UK. https://doi.org/10.1057/9781137469939_3
- Gibson, J. J. (1979). *The Ecological Approach to Visual Perception*. Houghton, Mifflin and Company.
- Goktas, Y., Yildirim, Z., & Yildirim, S. (2009). Investigation of K-12 teachers' ICT competencies and the contributing factors in acquiring these competencies. *The New Educational Review*, 17(1), 276–294.
- Gonzalez-Franco, M., & Lanier, J. (2017). Model of illusions and virtual reality. *Frontiers in Psychology*, 8. https://doi.org/10.3389/fpsyg.2017.01125
- Graham, C., Borup, J., & Smith, N. (2012). Using TPACK as a framework to understand teacher candidates' technology integration decisions. *Journal of Computer Assisted Learning*, 28(6), 530–546. <u>https://doi.org/10.1111/j.1365-2729.2011.00472.x</u>
- Grenier, S., Forget, H., Bouchard, S., Isere, S., Belleville, S., Potvin, O., & Talbot, M. (2015). Using virtual reality to improve the efficacy of cognitive-behavioral therapy (CBT) in

the treatment of late-life anxiety: preliminary recommendations for future research. *Int. Psychogeriatr.*, *27*, 1217–1225. https://doi.org/10.1017/S1041610214002300

Guski, R. (1996). Wahrnehmen – ein Lehrbuch. Kohlhammer.

- Guzey, S. S., & Roehrig, G. H. (2009). Teaching science with technology: Case studies of science teachers' development of technology, pedagogy, and content knowledge. *Contemporary Issues in Technology and Teacher Education*, 9(1), 25-45.
- Hacker, J. V., Johnson, M., Saunders, C., & Thayer, A. L. (2019). Trust in virtual teams: A multidisciplinary review and integration. *Australasian Journal of Information Systems*, 23. <u>https://doi.org/10.3127/ajis.v23i0.1757</u>
- Hacker, J. V., vom Brocke, J., Handali, J., Otto, M., & Schneider, J. (2020). Virtually in this together how web-conferencing systems enabled a new virtual togetherness during the COVID-19 crisis. *European Journal of Information Systems*, 29(5), 563–584. https://doi.org/10.1080/0960085X.2020.1814680
- Hambley, L. A., O'Neill, T. A., & Kline, T. J. (2007). Virtual team leadership: Perspectives from the field. *International Journal of e-Collaboration (IJeC)*, 3(1), 40-64. <u>http://doi.org/10.4018/jec.2007010103</u>
- Hamilton, D., McKechnie, J., Edgerton, E., & Wilson, C. (2020). Immersive virtual reality as a pedagogical tool in education: a systematic literature review of quantitative learning outcomes and experimental design. *Journal of Computers in Education*, 8(1), 1–32. <u>https://doi.org/10.1007/s40692-020-00169-2</u>
- Harris, J., & Gallagher, E. (Eds.). (2021, March 31). *TPACK newsletter issue #44: March 2021* [Electronic mailing list message]. http://bit.ly/TPACKNewslettersArchive
- Harris, J., & Hofer, M. (2009). Instructional planning activity types as vehicles for curriculum based TPACK development. In C. D., Maddux (Ed.), *Research highlights in technology* and teacher education 2009 (pp. 99–108). Society for Information Technology & Teacher Education.
- Harvey, D. M., & Caro, R. (2017). Building TPACK in preservice teachers through explicit course design. *TechTrends*, 61(2), 106–114. <u>https://doi.org/10.1007/s11528-016-0120-</u>
- Harzenmoser, D., Norton, J., & Peterson, B. (2019). Preventing VR-induced psychogenic death. 10.13140/RG.2.2.22657.71529

- Hascher, T., & Waber, J. (2021). Teacher well-being: A systematic review of the research literature from the year 2000–2019. *Educational Research Review*, 34, 100411. https://doi.org/10.1016/j.edurev.2021.100411
- Hellriegel, J., & Čubela, D. (2018). Das Potenzial von Virtual Reality für den schulischen Unterricht - Eine konstruktivistische Sicht. MedienPädagogik: Zeitschrift für Theorie und Praxis der Medienbildung, 58–80. <u>https://doi.org/10.21240/mpaed/00/2018.12.11.x</u>
- Herring, M., Thomas, T., & Redmond, P. (2014). Special Editorial: Technology leadership for preparing tomorrow's teachers to use technology. *Journal of Digital Learning in Teacher Education*, 30(3), 76–80. https://doi.org/10.1080/21532974.2014.891875
- Herzig, B., Martin, A., Schaper, N., & Ossenschmidt, D. (2015). Modellierung und Messung medienpädagogischer Kompetenz Grundlagen und erste Ergebnisse. In B. Koch-Priewe, A. Köker, J. Seifried, & E. Wuttke (Eds.), Kompetenzerwerb an Hochschulen: Modellierung und Messung. Zur Professionalisierung angehender Lehrerinnen und Lehrer sowie frühpädagogischer Fachkräfte (pp. 153–176). Klinkhardt.
- Herzig, B., Schaper, N., Martin, A., & Ossenschmidt, D. (2016). Schlussbericht zum BMBF Verbundprojekt M³K – Modellierung und Messung medienpädagogischer Kompetenz, Teilprojekt Medienerzieherische und mediendidaktische Facetten und handlungsleitende Einstellungen. Universität Paderborn.
- Hesse, F., & Lütgert, W. (Eds.). (2020). Auf die Lernbegleitung kommt es an! Konzepte und Befunde zu Praxisphasen in der Lehrerbildung. Klinkhardt, Julius.
- Huang, H. M., & Liaw, S. S. (2018). An Analysis of Learners' Intentions Toward Virtual Reality Learning Based on Constructivist and Technology Acceptance Approaches. *The International Review of Research in Open and Distributed Learning*, 19(1). <u>https://doi.org/10.19173/irrodl.v19i1.2503</u>
- Huang, H.-M., Rauch, U., & Liaw, S.-S. (2010). Investigating learners' attitudes toward virtual reality learning environments: Based on a constructivist approach. *Computers & Education*, 55(3), 1171–1182. <u>https://doi.org/10.1016/j.compedu.2010.05.014</u>
- Irwanto, I. (2021). Research trends in technological pedagogical content knowledge (TPACK): A systematic literature review from 2010 to 2021. European Journal of Educational Research, 10(4), 2045–2054. <u>https://doi.org/10.12973/eu-jer.10.4.2045</u>
- Johnson-Glenberg, M. C. (2018). Immersive VR and education: Embodied design principles that include gesture and hand controls. *Frontiers in Robotics and AI*, 5, 81. https://doi.org/10.3389/frobt.2018.00081

- Jonassen, D. H. (1992). What are cognitive tools? In P. A. M. Kommers, D. H. Jonassen, J. T. Mayes & A. Ferreira (Eds.), *Cognitive Tools for Learning* (pp.1–6). Springer-Verlag. https://doi.org/10.1007/978-3-642-77222-1_1
- jugendschutz.net. (2021, November). 2020 2021 Bericht Rechtsextremismus im Netz. https://www.jugendschutz.net/fileadmin/daten/publikationen/lageberichte/bericht_202 0_2021_rechtsextremismus_im_netz.pdf
- Kay, R. H. (2006). Evaluating strategies used to incorporate technology into pre-service education: A review of the literature. *Journal of Research on Technology in Education* and Information Technologies, 38(4), 383–408.
- Kilteni, K., Bergstrom, I., & Slater, M. (2013). Drumming in immersive virtual reality: the body shapes the way we play. *IEEE Trans. Visual. Comput. Graphics*, 19(4), 597–605. https://doi.org/10.1109/VR.2013.6549442
- Kilteni, K., Groten, R., & Slater, M. (2012). The Sense of Embodiment in Virtual Reality. *Presence: Teleoperators and Virtual Environments*, 21(4), 373–387. <u>https://doi.org/10.1162/pres_a_00124</u>
- Koehler, M. J., & Mishra, P. (2005). Teachers learning technology by design. *Journal of computing in teacher education*, 21(3), 94-102.
- Koehler, M. J., Mishra, P., & Yahya, K. (2007). Tracing the development of teacher knowledge in a design seminar: Integrating content, pedagogy and technology. *Computers & Education*, 49(3), 740-762.
- Koehler, M., & Mishra, P. (2009). What is technological pedagogical content knowledge (TPACK)? *Contemporary issues in technology and teacher education*, 9(1), 60-70.
- Koehler, M. J., & Mishra, P. (2008). Introducing TPCK. In AACTE Committee on Innovation and Technology (Ed.). *The handbook of technological pedagogical content knowledge* (*TPCK*) for educators (pp. 3-29). Routledge.
- Koh, J. H. L., & Chai, C. S. (2014). Teacher clusters and their perceptions of technological pedagogical content knowledge (TPACK) development through ICT lesson design. *Computers & Education*, 70, 222–232. https://doi.org/10.1016/j.compedu.2013.08.017
- Koh, J. H. L., Woo, H. L., & Lim, W. Y. (2013). Understanding the relationship between Singapore preservice teachers' ICT course experiences and technological pedagogical content knowledge (TPACK) through ICT course evaluation. *Educational Assessment, Evaluation and Accountability*, 25(4), 321–339. https://doi.org/10.1007/s11092-013-9165-y

- Kramarski, B., & Michalsky, T. (2010). Preparing pre-service teachers for self-regulated learning in the context of technological pedagogical content knowledge. *Learning and Instruction*, 20(5), 434–447. https://doi.org/10.1016/j.learninstruc.2009.05.003
- Krauskopf, K., Foulger, T. S., & Williams, M. K. (2018). Prompting teachers' reflection of their professional knowledge. A proof-of-concept study of the Graphic Assessment of TPACK Instrument. *Teacher Development*, 22(2), 153-174. https://doi.org/10.1080/13664530.2017.1367717
- Krauskopf, K., Zahn, C., & Hesse, F. W. (2012). Leveraging the affordances of Youtube: The role of pedagogical knowledge and mental models of technology functions for lesson planning with technology. *Computers & Education*, 58(4), 1194–1206. <u>https://doi.org/10.1016/j.compedu.2011.12.010</u>
- Krauskopf, K., Zahn, C., & Hesse, F. W. (2015). Cognitive processes underlying TPCK: Mental models, cognitive transformation, and meta-conceptual awareness. In N. Valanides & C. Angeli (Eds.), *Technological Pedagogical Content Knowledge: Exploring, Developing, and Assessing TPCK* (pp. 41–61). Springer. https://doi.org/10.1007/978-1-4899-8080-9 3
- Krueger, R. A., & Casey, M. A. (2015). Focus Groups. A practical guide for applied research (5th ed.). Sage.
- Krumsvik, R. J. (2014). Teacher educators' digital competence. *Scandinavian Journal of Educational Research*, 58(3), 269–280. <u>https://doi.org/10.1080/00313831.2012.726273</u>
- Lamb, R., Antonenko, P., Etopio, E., & Seccia, A. (2018). Comparison of virtual reality and hands on activities in science education via functional near infrared spectroscopy. *Computers & Education*, 124, 14–26. https://doi.org/10.1016/j.compedu.2018.05.014
- Lamb, R., Etopio, E., Hand, B., & Yoon S. (2019). Virtual reality simulation: Effects on academic performance within two domains of writing in science. *Journal of Science Education and Technology*, 1-11.
- Latoschik, M. E., & Wienrich, C. (2021, April 10). Coherence and plausibility, not presence?! pivotal conditions for XR experiences and effects, a novel model. http://arxiv.org/pdf/2104.04846v2
- Latoschik, M. E., Kern, F., Stauffert, J.-P., Bartl, A., Botsch, M., & Lugrin, J.-L. (2019). Not Alone Here?! Scalability and User Experience of Embodied Ambient Crowds in Distributed Social Virtual Reality. *IEEE Transactions on Visualization and Computer Graphics*, 25(5), 2134–2144. <u>https://doi.org/10.1109/tvcg.2019.2899250</u>

- Latoschik, M. E., Roth, D., Gall, D., Achenbach, J., Waltemate, T., & Botsch, M. (2017). The effect of avatar realism in immersive social virtual realities. *Proceedings of the 23rd* ACM Symposium on Virtual Reality Software and Technology - VRST '17. https://doi.org/10.1145/3139131.3139156
- Lee, K. M. (2004). Presence, explicated. *Communication Theory*, *14*(1), 27–50. <u>https://doi.org/10.1111/j.1468-2885.2004.tb00302.x</u>
- Lombard, M., Reich, R. D., Grabe, M. E., Bracken, C., & Ditton, T. B. (2000). Presence and television: The role of screen size. *Human Communication Research*, 26, 75–98.
- Lin, T. C., Tsai, C. C., Chai, C. S., & Lee, M. H. (2013). Identifying science teachers' perceptions of technological pedagogical and content knowledge (TPACK). *Journal of Science Education and Technology*, 22(3), 325–336. <u>https://doi.org/10.1007/s10956-012-9396-6</u>
- Linn, M. C., Eylon, B.-S., & Davis, E. A. (2004). The knowledge integration perspective on learning. In M. C. Linn, E. A. Davis, & P. Bell (Eds.), *Internet environments for science education* (pp. 29–46). Lawrence Erlbaum Associates.
- Liou, W.-K. & Chang, C.-Y. (2018). Virtual reality classroom applied to science education. In IT: 2018 23rd International Scientific-Professional Conference on Information Technology: 19-24 February 2018, Zabljak, Montenegro (pp. 1–4). Institute of Electrical and Electronics Engineers. https://doi.org/10.1109/SPIT.2018.8350861
- Lugrin, J.-L., Latoschik, M. E., Habel, M., Roth, D., Seufert, C., & Grafe, S. (2016). Breaking Bad Behaviors: A new tool for learning classroom management using virtual reality. *Frontiers in ICT*, 3, Article 26. https://doi.org/10.3389/fict.2016.00026
- Luo, H., Li, G., Feng, Q., Yang, Y., & Zuo, M. (2021). Virtual reality in K-12 and higher education: A systematic review of the literature from 2000 to 2019. *Journal of Computer Assisted Learning*. <u>https://doi.org/10.1111/jcal.12538</u>
- Makransky, G., Terkildsen, T. S., & Mayer, R. E. (2019). Adding immersive virtual reality to a science lab simulation causes more presence but less learning. *Learning and Instruction*, 60, 225–236. https://doi.org/10.1016/j.learninstruc.2017.12.007
- Marquart, C. L., Hinojosa, C., Swiecki, Z., Eagan, B., & Shaffer, D. W. (2018). *Epistemic Network Analysis* (Version 1.7.0) [Software]. Available from http://app.epistemicnetwork.org
- Massner, C. K. (2021). The Use of Videoconferencing in Higher Education. In *Communication Management*. IntechOpen. https://doi.org/10.5772/intechopen.99308

- Mayring, P. (2015). Qualitative Content Analysis: Theoretical background and procedures. In
 A. Bikner-Ahsbahs, C. Knipping & N. Presmeg (Eds.), Advances in mathematics education. Approaches to qualitative research in mathematics education: Examples of methodology and methods (pp. 365–380). Springer Netherlands.
- McCallum, F. (2021). Teachers' wellbeing during times of change and disruption. In M. A.
 White & F. McCallum (Eds.), *Wellbeing and resilience education: Covid-19 and its impact on education systems* (pp. 183–208). Routledge.
- McGarr, O., & McDonagh, A. (2019). Digital competence in teacher education: Output 1 of the Erasmus+ funded Developing Student Teachers' Digital Competence (DICTE) project. https://dicte.oslomet.no/
- McGrenere, J., & Ho, W. (2000). Affordances: Clarifying and evolving a concept. In *Proceedings of the Graphics Interface 2000 Conference* (pp. 179-186), Montreal, Canada.
- McKenney, S., & Reeves, T. C. (2018). Conducting educational design research. Routledge.
- McKenney, S., & Reeves, T. (2013). Systematic review of design-based research progress: Is a little knowledge a dangerous thing? *Educational Researcher*, 42(2), 97-100. <u>https://doi.org/10.3102/0013189X12463781</u>
- Meeus, W., Van Ouytsel, J., Driesen, A., & T'Sas, J. (2014). Media Didactica: A media literacy reference framework for learners, teachers and teacher educators. *Merz medien* + *erziehung*, 58(6), 41–49.
- Merchant, Z., Goetz, E. T., Cifuentes, L., Keeney-Kennicutt, W., & Davis, T. J. (2014).
 Effectiveness of virtual reality-based instruction on students' learning outcomes in K-12 and higher education: A meta-analysis. *Computers & Education*, 70, 29–40. http://dx.doi.org/10.1016/j.compedu.2013.07.033
- Mishra, P., & Koehler, M. J. (2006). Technological pedagogical content knowledge: A framework for teacher knowledge. *Teachers College Record*, 108(6), 1017–1054. <u>https://doi.org/10.1111/j.1467-9620.2006.00684.x</u>
- Mishra, P. (2019). Considering contextual knowledge: The TPACK diagram gets an upgrade. *Journal of Digital Learning in Teacher Education*, 35(2), 76–78. https://doi.org/10.1080/21532974.2019.1588611
- Moreno, R. (2005). Multimedia learning with animated pedagogical agents. *The Cambridge Handbook of Multimedia Learning*, 507–524. https://doi.org/10.1017/cbo9780511816819.032

- Moro, C., Štromberga, Z., Raikos, A., & Stirling, A. (2017). The effectiveness of virtual and augmented reality in health sciences and medical anatomy. *Anat. Sci. Educ.* 10, 549– 559. doi: 10.1002/ase.1696
- Newman, D. (1990). Opportunities for research on the organizational impact of school computers. *Educational Researcher*. 19(3), 8–13.
- Niess, M. L. (2005). Preparing teachers to teach science and mathematics with technology: Developing a technology pedagogical content knowledge. *Teaching and teacher education*, 21(5), 509-523.
- Norman, D. A. (1988). The Psychology of everyday things, New York, Basic Book. *paperback* as the Design of Everyday Things, 1990.
- Parong, J., & Mayer, R. E. (2021). Learning about history in immersive virtual reality: does immersion facilitate learning? *Educational Technology Research and Development*, 69(3), 1433–1451. https://doi.org/10.1007/s11423-021-09999-y
- Parong, J., & Mayer, R. E. (2018). Learning science in immersive virtual reality. Journal of Educational Psychology, 110(6), 785–797. https://doi.org/10.1037/edu0000241
- Pellas, N., Mystakidis, S., & Kazanidis, I. (2021). Immersive Virtual Reality in K-12 and Higher Education: A systematic review of the last decade scientific literature. *Virtual Reality*, 25(3), 835–861. <u>https://doi.org/10.1007/s10055-020-00489-9</u>
- Pelletier, K., Brown, M., Brooks, D.C., McCormack, M., Reeves, J., Arbino, N., Bozkurt, A., Crawford, S., Czerniewicz, L., Gibson, R., Linder, K., Mason, J., & Mondelli, V. (2021). 2021 EDUCAUSE Horizon Report Teaching and Learning Edition. Boulder, CO: EDU. <u>https://www.learntechlib.org/p/219489/</u>.
- Perry, T. S. (2016). Virtual reality goes social. *IEEE Spectrum*, 53(1), 56–57. https://doi.org/10.1109/MSPEC.2016.7367470
- Perkins Coie. (2020). 2020 Augmented and Virtual Reality Survey Report: Industry Insights into the Future of Immersive Technology. https://www.perkinscoie.com/images/content/2/3/v4/231654/2020-AR-VR-Surveyv3.pdf
- Petersen, G. B., Mottelson, A., & Makransky, G. (2021). Pedagogical agents in educational VR: An in the wild study. Proceedings of the 2021 CHI Conference on Human Factors in Computing Systems. <u>https://doi.org/10.1145/3411764.3445760</u>
- Pierson, M. E. (2001). Technology integration practice as a function of pedagogical expertise. Journal of Research on Computing in Education, 33(4), 413–429.

- Plomp, T. (2010). Educational design research: An introduction. In T. Plomp & N. Nieveen (Ed.), An introduction to Educational Design Research (3. Aufl., S. 9–36).
- Porras-Hernández, L. H., & Salinas-Amescua, B. (2013). Strengthening Tpack: A broader notion of context and the use of teacher's narratives to reveal knowledge construction. *Journal of Educational Computing Research*, 48(2), 223–244. https://doi.org/10.2190/ec.48.2.f
- Przyborski, A., & Wohlrab-Sahr, M. (2013). *Qualitative Sozialforschung: Ein Arbeitsbuch*. München, Germany: Oldenbourg.
- Raatz, S. (2016). II. Design-based Research–Innovation für Bildungswissenschaft und-praxis.
 In *Entwicklung von Einstellungen gegenüber verantwortungsvoller Führung* (pp. 37-61). Springer VS, Wiesbaden. https://doi.org/10.1007/978-3-658-11980-5_2
- Radianti, J., Majchrzak, T. A., Fromm, J., & Wohlgenannt, I. (2020). A systematic review of immersive virtual reality applications for higher education: Design elements, lessons learned, and research agenda. *Computers & Education*, 147, 103778. https://doi.org/10.1016/j.compedu.2019.103778
- Rädiker, S. & Kuckartz, U. (2019). *Analyse qualitativer Daten mit MAXQDA: Text, Audio und Video*. Wiesbaden, Germany: Springer Fachmedien Wiesbaden.
- Rädiker, S., & Kuckartz, U. (2018). *Analyse qualitativer Daten mit MAXQDA: Text, Audio und Video*. Wiesbaden, Germany: Springer Fachmedien Wiesbaden.
- Redecker, C. (2017). European Framework for the Digital Competence of Educators: DigCompEdu. Luxembourg. Publications Office of the European Union. <u>https://doi.org/10.2760/159770</u>
- Reeves, T. (2006). Design research from a technology perspective. In J. Van Den Akker,
 K.Gravemeijer, S. McKenney, & N. Nieveen (Eds.), *Educational design research* (52–66). London: Routledge.
- Reigeluth, C. M. (2015). Instructional Theory and Technology for the New Pardigm of Education. In B. Khan, & M. Ally, *International Handbook of E-learning, Volume 1* (S. 79-92). New York, London: Routledge.
- Reinmann, G., & Sesink, W. (2011). Entwicklungsorientierte Bildungsforschung (Diskussionspapier). Online verfügbar unter: <u>http://gabi-reinmann.de/wpcontent/uploads/2011/11/SesinkReinmann_Entwicklungsforschung_v05_20_11_2011.</u> <u>pdf</u>.
- Richard, R., & Rodway, M. R. (1992). The peer consultation group: A problem-solving perspective. *The Clinical Supervisor*, *10*(1), 83–100.

- Riegler, A., Riener, A., & Holzmann, C. (2021). A systematic review of virtual reality Applications for Automated Driving: 2009–2020. *Frontiers in Human Dynamics*, *3*. https://doi.org/10.3389/fhumd.2021.689856
- Ripka, G., Tiede, J., Grafe, S., & Latoschik, E. (2020a). Teaching and learning processes in immersive VR Comparing expectations of preservice teachers and teacher educators.
 In D. Schmidt-Crawford (Ed.), *Proceedings of Society for Information Technology & Teacher Education International Conference* (pp. 1836–1844)
- Ripka, G., Grafe, S., & Latoschik, M. E. (2020b). Preservice teachers' encounter with social VR–Exploring virtual teaching and learning processes in initial teacher education. In *SITE Interactive Conference* (pp. 549–562). Association for the Advancement of Computing in Education (AACE).
- Ripka, G., Grafe, S., & Latoschik, M. E. (2021a). Peer group supervision in Zoom and social VR-Preparing preservice teachers for planning and designing digital media integrated classes. In *EdMedia+ Innovate Learning* (pp. 602-616). Association for the Advancement of Computing in Education (AACE).
- Ripka, G., Grafe, S. & Latoschik, M.E. (2021b). Mapping pre-service teachers' TPACK development using a social virtual reality and a video-conferencing system. In T. Bastiaens (Ed.), *Proceedings of Innovate Learning Summit 2021* (pp. 145-159). Online, United States: Association for the Advancement of Computing in Education (AACE). <u>https://www.learntechlib.org/primary/p/220280/</u>.
- Roth, D., & Latoschik, M. E. (2020). Construction of the virtual embodiment questionnaire (VEQ). *IEEE Transactions on Visualization and Computer Graphics*, 1. https://doi.org/10.1109/tvcg.2020.3023603
- Roth, D., Kleinbeck, C., Feigl, T., Mutschler, C., & Latoschik, M. E. (2018, March). Beyond replication: Augmenting social behaviors in multi-user virtual realities. In 2018 IEEE Conference on Virtual Reality and 3D User Interfaces (VR) (pp. 215-222). IEEE
- Roth, D., Latoschik, M. E., Bloch, C., & Bente, G. (2017). When some things are missing: The quality of interpersonal communication in social virtual reality. Presentation at the 68th Annual Conference of the International Communication Association (ICA), May 24-28 2018, Prague, Czech Republic.
- Roth, D., Lugrin, J.-L., Galakhov, D., Hofmann, A., Bente, G., Latoschik, M. E., & Fuhrmann,
 A. (2016). Avatar realism and social interaction quality in virtual reality. 2016 IEEE
 Virtual Reality (VR), 277–278. <u>https://doi.org/10.1109/vr.2016.7504761</u>

- Roth, D., Lugrin, J.-L., Latoschik, M. E., & Huber, S. (2017). Alpha IVBO Construction of a scale to measure the illusion of virtual body ownership. Proceedings of the 2017 CHI Conference Extended Abstracts on Human Factors in Computing Systems CHI EA '17, 2875–2883. <u>https://doi.org/10.1145/3027063.3053272</u>
- Ruis, A. R. (2018). Trois empreintes d'un même cachet: Toward a historical definition of nutrition. In E. T. Ewing & K. Randall (eds.), Viral networks: Connecting digital humanities and medical history (pp. 179–212). Blacksburg: VT Publishing.
- Ruis, A. R., Rosser, A. A., Quandt-Walle, C., Nathwani, J. N., Shaffer, D. W., & Pugh, C. M. (2018). The hands and head of a surgeon: Modeling operative competency with multimodal epistemic network analysis. American Journal of Surgery, 216(5), 835-840.
- Sang, G., Valcke, M., Braak, J., Tondeur, J., & Zhu, C. (2011). Predicting ICT integration into classroom teaching in Chinese primary schools: Exploring the complex interplay of teacher-related variables, Journal of Computer Assisted Learning, vol. 27, pp.160-172. doi: 10.1111/j.1365-2729.2010.00383.x
- Scavarelli, A., Arya, A., & Teather, R. J. (2020). Virtual reality and augmented reality in social learning spaces: A literature review. *Virtual Reality*. Advance online publication. https://doi.org/10.1007/s10055-020-00444-8
- Schmid, M., & Petko, D. (2020). Technological Pedagogical Content Knowledge als Leitmodell medienpädagogischer Kompetenz. MedienPädagogik: Zeitschrift für Theorie und Praxis der Medienbildung, 121–140. https://doi.org/10.21240/mpaed/jb17/2020.04.28.x
- Schmid, M., Brianza, E., & Petko, D. (2020). Developing a short assessment instrument for Technological Pedagogical Content Knowledge (TPACK.xs) and comparing the factor structure of an integrative and a transformative model. *Computers & Education*, 157, 103967. https://doi.org/10.1016/j.compedu.2020.103967
- Schmid, U., Goertz, L., Radomski, S., Thom, S., Behrens, J., & Bertelsmann Stiftung (2017). Monitor Digitale Bildung: Die Hochschulen im digitalen Zeitalter. <u>https://doi.org/10.11586/2017014</u>
- Schmidt, D. A., Baran, E., Thompson, A. D., Mishra, P., Koehler, M. J., & Shin, T. S. (2009). Technological pedagogical content knowledge (TPACK) the development and validation of an assessment instrument for preservice teachers. *Journal of research on Technology in Education*, 42(2), 123-149.
- Schön, D. A. (1987). Educating the reflective practitioner: Toward a new design for teaching and learning in the professions. Jossey-Bass.

- Schwan, S., & Buder, J. (2006). Virtuelle Realität und E-Learning. Tübingen. <u>https://www.e-</u> teaching.org/materialien/literatur/schwan-buder-2005
- Seufert, C. M., & Grafe, S. (2020). Förderung der Klassenführungskompetenzen von Lehramtsstudierenden unter Verwendung eines virtuellen Klassenzimmers. MedienPädagogik: Zeitschrift für Theorie und Praxis der Medienbildung, 39, 142–169. https://doi.org/10.21240/mpaed/39/2020.12.09.x
- Seufert, C. M., Oberdörfer, S., Roth, A., Grafe, S., Lugrin, J. L., & Latoschik, M. E. (2022). Classroom management competency enhancement for student teachers using a fully immersive virtual classroom. *Computers & Education*, 179, 104410. https://doi.org/10.1016/j.compedu.2021.104410
- Shaffer, D. W. (2012). Models of situated action: Computer games and the problem of transfer.In C. Steinkuehler, K. D. Squire, & S. A. Barab (Eds.), Games, learning, and society:Learning and meaning in the digital age (pp. 403–431). Cambridge, UK: Cambridge University Press.
- Shaffer, D. W. (2017). Quantitative ethnography. Madison, WI: Cathcart Press.
- Shaffer, D. W., & Ruis, A. R. (2017). Epistemic network analysis: A worked example of theorybased learning analytics. In C. Lang, G. Siemens, A. F. Wise, & D. Gasevic (Eds.), Handbook of learning analytics (pp. 175–187). Society for Learning Analytics Research.
- Shaffer, D. W., Collier, W., & Ruis, A. R. (2016). A tutorial on epistemic network analysis: Analyzing the structure of connections in cognitive, social, and interaction data. Journal of Learning Analytics, 3(3), 9–45.
- Shaffer, D. W., Hatfield, D. L., Svarovsky, G. N., Nash, P., Nulty, A., Bagley, E. A., Frank, K., Rupp, A. A., Mislevy, R. (2009). Epistemic network analysis: A prototype for 21st century assessment of learning. *International Journal of Learning and Media*, 1(1), 1– 2. doi: 10.1162/ijlm.2009.0013
- Sharma, S., Agada, R., & Ruffin, J. (2013). Virtual reality classroom as a constructivist approach. 2013 Proceedings of IEEE Southeastcon (pp.1-5). https://doi.org/10.1109/secon.2013.6567441
- Shih, Y.-C., & Yang, M.-T. (2008). A collaborative virtual environment for situated language learning using VEC3D. *Educational Technology & Society*, 11(1), 56–68.
- Shiram, K., & Schwarz, R. (2017). All are welcome: Using VR ethnography to explore harassment behavior in immersive social virtual reality. IEEE Virtual Reality (VR), Los Angeles, CA, 2017, pp. 225-226, doi:10.1109/VR.2017.7892258.

- Shulman, L. S (1987). Knowledge and teaching: Foundations of the new reform. HarvardEducationalReview,57(1),1–23.https://doi.org/10.17763/haer.57.1.j463w79r56455411
- Shulman, L. S. (1986). Those who understand: Knowledge growth in teaching. *Educational* researcher, 15(2), 4-14.
- Shih, C. L., & Chuang, H. H. (2012). The development and validation of an instrument for assessing college students' perceptions of faculty knowledge in technology-supported class environments. Computers & Education, 63, 109-118.
- Skarbez, R., Brooks, F., & Whitton, M. (2020). Immersion and Coherence: Research Agenda and Early Results. *IEEE Transactions on Visualization and Computer Graphics*, PP. <u>https://doi.org/10.1109/TVCG.2020.2983701</u>
- Skarbez, R., Brooks, Jr., F. P., & Whitton, M. C. (2017). A survey of presence and related concepts. ACM Comput. Surv., 50(6):96:1–96:39.
- Slater, M., & Sanchez-Vives, M. V. (2016). Enhancing Our Lives with Immersive Virtual Reality. *Frontiers in Robotics and AI*, 3. <u>https://doi.org/10.3389/frobt.2016.00074</u>
- Slater, M., & Steed, A. (2000). A virtual presence counter. Presence: Teleoperators and virtual environments, 9(5), 413–434. https://doi.org/10.1162/105474600566925
- Slater, M., & Wilbur, S. (1997). A framework for immersive virtual environments (FIVE): Speculations on the role of presence in virtual environments. *Presence: Teleoperators* and Virtual Environments, 6(6), 603–616. <u>https://doi.org/10.1162/pres.1997.6.6.603</u>
- Slater, M., Gonzalez-Liencres, C., Haggard, P., Vinkers, C., Gregory-Clarke, R., Jelley, S., Watson, Z., Breen, G., Schwarz, R., Steptoe, W., Szostak, D., Halan, S., Fox, D., & Silver, J. (2020). The Ethics of Realism in Virtual and Augmented Reality. *Frontiers in Virtual Reality*, *1*. <u>https://doi.org/10.3389/frvir.2020.00001</u>
- Southgate, E. (2020). Virtual Reality in Curriculum and Pedagogy. *Virtual Reality in Curriculum and Pedagogy*. <u>https://doi.org/10.4324/9780429291982</u>
- Southgate, E., Smith, S. P., Cividino, C., Saxby, S., Kilham, J., Eather, G., Scevak, J., Summerville, D., Buchanan, R., & Bergin, C. (2019). Embedding immersive virtual reality in classrooms: Ethical, organisational and educational lessons in bridging research and practice. *International Journal of Child-Computer Interaction*, 19, 19–29. https://doi.org/10.1016/j.ijcci.2018.10.002
- Ständige Konferenz der Kultusminister der Länder in der Bundesrepublik Deutschland [KMK] (2016). Bildung in der digitalen Welt: Strategie der Kultusministerkonferenz.

https://www.kmk.org/fileadmin/Dateien/pdf/PresseUndAktuelles/2016/Bildung_digita le_Welt_Webversion.pdf

- Stavroulia, K-E., Christofi, M., Zarraonandia, T., Michael-Grigoriou, D., & Lanitis, A. (2019). Virtual Reality Environments (VREs) for Training and Learning. In: Paloma Díaz, Andri Ioannou, Kaushal Kumar Bhagat & J. Michael Spector (Ed.): *Learning in a Digital World*, Bd. 9. Singapore: Springer Singapore (Smart Computing and Intelligence), S. 195–211.
- Su, M., & Foulger, T. (2019, March). We aren't there yet: A progression of literature on TPACK measures to assess technology integration. In *Society for Information Technology & Teacher Education International Conference* (pp. 2534- 2542). Association for the Advancement of Computing in Education (AACE).
- Sullivan, S. A., Warner-Hillard, C., Eagan, B. R., Thompson, R. J., Ruis, A. R., Haines, K., Pugh, C. M., Shaffer, D. W., & Jung, H. S. (2017). Using epistemic network analysis to identify targets for educational interventions in trauma team communication. Surgery, in press. doi: 10.1016/j.surg.2017.11.009.
- Taylor, S. J., Bogdan, R., & DeVault, M. (2015). Introduction to qualitative research methods: A guidebook and resource. Hoboken, NJ: John Wiley & Sons.
- Tiede, J., & Grafe, S. (2016). Media Pedagogy in German and U.S. Teacher Education. Comunicar, 24(49), 19–28. <u>https://doi.org/10.3916/c49-2016-02</u>
- Tiede, J., & Grafe, S. (2019). The Integration of media-related studies and competencies into US and German Initial Teacher Education. A cross-national analysis of contemporary practices and trends. In K. Graziano (Ed.), *Proceedings of Society for Information Technology & Teacher Education International Conference* (pp. 1709-1717). Las Vegas, NV, United States: Association for the Advancement of Computing in Education (AACE). <u>https://www.learntechlib.org/primary/p/207873/</u>.
- Tiede, J., (2020). Media-related educational competencies of German and US preservice teachers. A comparative analysis of competency models, measurements and practices of advancement». MedienPädagogik, (Media-related Educational Competencies), 1–37. https://doi.org/10.21240/mpaed/diss.jt/2020.07.05.X.
- Tietze, K.-O. (2010). Wirkprozesse und personenbezogene Wirkungen von kollegialer Beratung: Theoretische Entwürfe und empirische Forschung. VS Verlag für Sozialwissenschaften / GWV Fachverlage GmbH, Wiesbaden.

- Tietze, K.-O. (2013). Kollegiale Beratung: Problemlösungen gemeinsam entwickeln (6. Aufl., Orig.-Ausg). Miteinander reden: Vol. 61544. Rowohlt-Taschenbuch-Verl. http://www.socialnet.de/rezensionen/isbn.php? isbn=978-3-499-61544-3
- Tietze, K.-O. (2018). Kollegiale Beratung: Problemlösungen gemeinsam entwickeln (9. Auflage). Miteinander reden: Vol. 61544. Rowohlt Taschenbuch Verlag.
- Tietze, K.-O. (in press). Kollegiale Beratung für Studienberatende. In T. Grüneberg, I. Blaich,J. Egerer, B. Knickrehm, M. Liebchen, L. Lutz, U. Nachtigaller & R. Thiel (Hrsg.),Handbuch Studienberatung. Berufliche Orientierung und Beratung für akademischeBildungswege, Band 1. Bielefeld: wbv/utb
- Tondeur, J. (2018). Enhancing future teachers' competencies for technology integration in education: Turning theory into practice. *Seminar.net*, 14(2), 216–224. https://doi.org/10.7577/seminar.2981
- Tondeur, J., Scherer, R., Baran, E., Siddiq, F., Valtonen, T., & Sointu, E. (2019). Teacher educators as gatekeepers: Preparing the next generation of teachers for technology integration in education. *British Journal of Educational Technology*, 50(3), 1189–1209. <u>https://doi.org/10.1111/bjet.12748</u>
- Tseng, J. J., Cheng, Y. S., & Yeh, H. N. (2019). How pre-service English teachers enact TPACK in the context of web-conferencing teaching: A design thinking approach. *Computers & Education*, 128, 171–182. <u>https://doi.org/10.1016/j.compedu.2018.09.022</u>
- Tseng, J.-J., Chai, C. S., Tan, L., & Park, M. (2020). A critical review of research on technological pedagogical and content knowledge (TPACK) in language teaching. *Computer Assisted Language Learning*, 1–24. https://doi.org/10.1080/09588221.2020.1868531
- Tulodziecki, G. (1983). Theoriegeleitete Entwicklung und Evaluation von Lehrmaterialien als eine Aufgabe der Unterrichtswissenschaft. *Unterrichtswissenschaft*, 11(3), 27-45.
- Tulodziecki, G. (2012). Medienpädagogische Kompetenz und Standards in der Lehrerbildung. In *Jahrbuch Medienpädagogik 9* (pp. 271-297). VS Verlag für Sozialwissenschaften.
- Tulodziecki, G., Grafe, S., & Herzig, B. (2013). *Gestaltungsorientierte Bildungsforschung und Didaktik: Theorie Emprie Praxis*. Bad Heilbrunn, Deutschland: Klinkhardt.
- Tulodziecki, G., Herzig, B., & Grafe, S. (2021). Medienbildung in Schule und Unterricht: Grundlagen und Beispiele (3. revised and updated ed.). UTB GmbH.
- Tulodziecki, G., Herzig, B., & Blömeke, S. (2017). *Gestaltung von Unterricht: Eine Einführung in die Didaktik* (3., überarbeitete und erweiterte Auflage). *UTB Erziehungswissenschaft,*

Schulpädagogik, Allgemeine Didaktik: Vol. 3311. Verlag Julius Klinkhardt. http://www.utb-studi-e-book.de/9783838547947

- Tulodziecki, G., Herzig, B., & Grafe, S. (2018). Gestaltungs-und entwicklungsorientierte Forschung als Forschungsrichtung für die Medienpädagogik (pp. 423-448).
- Tulodziecki, G., Herzig, B., & Grafe, S. (2019). *Medienbildung in Schule und Unterricht: Grundlagen und Beispiele*. 2. Auflage. Bad Heilbrunn, Germany: Klinkhardt/ UTB.
- U.S. Department of Education, Office of Educational Technology. (2016). Future Ready Learning: Reimagining the Role of Technology in Education. 2016 National Education Technology Plan. Washington, DC: Author. Retrieved from https://tech.ed.gov/files/2015/12/NETP16.pdf
- Uerz, D., Volman, M., & Kral, M. (2018). Teacher educators' competences in fostering student teachers' proficiency in teaching and learning with technology: An overview of relevant research literature. *Teaching and Teacher Education*, 70, 12–23. https://doi.org/10.1016/j.tate.2017.11.005
- United Nations Educational, Scientific and Cultural Organization [UNESCO]. (2018). UNESCO ICT Competency Framework for Teachers. Paris, France: Author. Retrieved from https://en.unesco.org/
- Urban, E. R., Navarro, M., & Borron, A. (2018). TPACK to GPACK? The examination of the technological pedagogical content knowledge framework as a model for global integration into college of agriculture classrooms. *Teaching and Teacher Education*, 73, 81–89. https://doi.org/10.1016/j.tate.2018.03.013
- US Department of Education. (2017). *Reimagining the role of technology in education: 2017 national education technology plan update*. Office of Educational Technology Washington, DC.
- Van Den Akker, J. (1999). Principles and methods of development research. In J. Van Den Akker, R. Branch, K. Gustafson, N. Nieveen, & T. Plomp (Eds.), *Design approaches* and tools in education and training (pp. 1-14). Dordrecht: Kluwer Academic Publishers.
- Voogt, J., Fisser, P., Pareja Roblin, N., Tondeur, J., & van Braak, J. (2013). Technological pedagogical content knowledge–a review of the literature. *Journal of computer assisted learning*, *29*(2), 109–121.
- Wang, A., Thompson, M., Uz-Bilgin, C., & Klopfer, E. (2021). Authenticity, interactivity, and collaboration in virtual reality games: Best practices and lessons learned. *Frontiers in Virtual Reality*, 2. https://doi.org/10.3389/frvir.2021.734083

- Wang, M. (2020). Social VR: A new form of social communication in the future or a beautiful illusion? *Journal of Physics: Conference Series*, 1518(1), 12.032. hpps://doi.org/10.1088/1742-6596/1518/1/012032
- Wang, W., Schmidt-Crawford, D., & Jin, Y. (2018). Preservice teachers' TPACK development: A review of literature. *Journal of Digital Learning in Teacher Education*, 34(4), 234– 258. <u>https://doi.org/10.1080/21532974.2018.1498039</u>
- Wenger, E. (1999). Communities of practice: Learning, meaning, and identity. Cambridge, UK:

 Cambridge
 University
 Press.
 Retrieved
 from

 http://books.google.com/books?id=heBZpgYUKdAC&pgis=1
- Wiesner, C., & Schreiner, C. (2020). Digitale Kompetenzen: Computational Thinking als Basis
 eines Kompetenzmodells. *Bildung und Digitalisierung*, 29– 50.https://doi.org/10.5771/9783748906247-29.
- Willermark, S. (2017). Technological Pedagogical and Content Knowledge: A review of empirical studies published from 2011 to 2016. *Journal of Educational Computing Research*, 56(3), 315–343. <u>https://doi.org/10.1177/0735633117713114</u>
- Wooldridge, A. R., Carayon, P., Eagan, B. R., & Shaffer, D. W. (2018). Quantifying the qualitative with epistemic network analysis: A human factors case study of taskallocation communication in a primary care team. IIE Transactions on Healthcare Systems Engineering, in press.
- Yeh, S. -C., Tsai, C. -F., Fan, Y. -C., Liu, P. -C., & Rizzo, A. (2012). An innovative ADHD assessment system using virtual reality. 2012 IEEE-EMBS Conference on Biomedical Engineering and Sciences. https://doi.org/10.1109/IECBES.2012.6498026
- Zhang, L. -L., Wang, J. -Q., Qi, R. -R., Pan, L. -L., Li, M., & Cai, Y. -L. (2015). Motion sickness: Current knowledge and recent advance. CNS Neuroscience & Therapeutics, 22(1), 15–24. <u>https://doi.org/10.1111/cns.12468</u>
- Zohar, A., & Schwartzer, N. (2005). Assessing teachers' pedagogical knowledge in the context of teaching higher-order thinking. *International Journal of Science Education*, 27(13), 1595–1620. <u>https://doi.org/10.1080/09500690500186592</u>